FRANCHISE DISCLOSURE DOCUMENT



Rusty Taco Franchising, LLC a Delaware limited liability company 5412 W. Plano Pkwy., Suite 100 Plano, Texas 75093 (972)-467-8095 franchise@rustytaco.com www.rustytaco.com

The franchise is the right to develop a restaurant that offers freshly prepared food made with quality, fresh ingredients inspired by the flavors of Mexico and Texas, along with an assortment of alcoholic and non-alcoholic beverages under the Rusty Taco[®] trade name, trademark, and business system in an atmosphere designed to be a neighborhood gathering place.

The total investment necessary to develop a Rusty Taco restaurant ranges from \$528,400 to \$1,127,950. This includes \$30,000 to \$38,000 that must be paid to the franchisor or its affiliates.

The total investment necessary to begin operation of Rusty Taco area development business (with a minimum development commitment of 2 and a maximum development commitment of 5 Rusty Taco restaurants) is \$30,000 to \$75,000. This entire amount is payable to the franchisor or its affiliates.

This franchise disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this franchise disclosure document and all accompanying agreements carefully. You must receive this franchise disclosure document at least 14 calendar days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no governmental agency has verified the information contained in this document.**

You may wish to receive your franchise disclosure document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact our Franchise Development Department at 5412 W. Plano Pkwy., Suite 100, Plano, Texas 75093, (972)-467-8095.

The terms of your contract will govern your franchise relationship. Do not rely on the franchise disclosure document alone to understand your contract. Read your entire contract carefully. Show your contract and this franchise disclosure document to an advisor, like a lawyer or an accountant.

Buying a franchise is a complex investment. The information in this franchise disclosure document can help you make up your mind. More information on franchising, such as, "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this franchise disclosure document, is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issuance Date: April 24, 2025

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Exhibit E.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Exhibit A includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only Rusty Taco business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be a Rusty Taco franchisee?	Item 20 or Exhibit E list current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

What You Need to Know About Franchising Generally

<u>Continuing responsibility to pay fees.</u> You may have to pay royalties and other fees even if you are losing money.

Business model can change. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions.</u> You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions.</u> The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor.</u> Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

Renewal. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

When your franchise ends. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit G.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About *This* Franchise

Certain states require that the following risk(s) be highlighted:

- 1. <u>Out-of-State Dispute Resolution</u>. The franchise agreement and the area development agreement require you to resolve disputes with us by arbitration or litigation in California. Out-of-state arbitration or litigation may force you to accept a less favorable settlement for disputes. It also may cost more to arbitrate or litigate with us in California than in your home state.
- 2. **Short Operating History**. The franchisor is at an early stage of development and has a limited operating history. This franchise is likely to be a riskier investment than a franchise in a system with a longer operating history.
- 3. **Spousal Liability**. At the franchisor's request, your spouse must sign a document making your spouse liable for all financial obligations under the Franchise Agreement, even if your spouse has no ownership interest in the franchise. This Guarantee will place both your and your spouse's marital and personal assets (perhaps including your house) at risk if your franchise fails.
- 4. **<u>Financial Condition</u>**. The franchisor's financial condition, as reflected in its financial statements (see Item 21), calls into question the franchisor's financial ability to provide services and support to you.
- 5. <u>Unopened Franchises</u>. The franchisor has signed a significant number of franchise agreements with franchisees who have not yet opened their outlets. If other franchisees are experiencing delays in opening their outlets, you may also experience delays in opening your own outlet.
- 6. **Supplier Control.** You must purchase all or nearly all of the inventory or supplies that are necessary to operate your business from the franchisor, its affiliates, or suppliers that the franchisor designates, at prices the franchisor or they set. These prices may be higher than prices you could obtain elsewhere for the same or similar goods. This may reduce the anticipated profit of your franchise business.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

THE FOLLOWING APPLY TO TRANSACTIONS GOVERNED BY THE MICHIGAN FRANCHISE INVESTMENT LAW ONLY

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU:

- (a) A prohibition against you joining an association of franchisees.
- (b) A requirement that you assent to a release, assignment, novation, waiver, or estoppel which would deprive you of rights and protections provided under the Michigan Franchise Investment Law. This does not preclude you, after entering into a franchise agreement, from settling any and all claims.
- (c) A provision that permits us to terminate your franchise prior to the expiration of its term except for good cause. Good cause includes your failure to comply with any lawful provision of the Franchise Agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- (d) A provision that permits us to refuse to renew your franchise without fairly compensating you by repurchase or other means, for the fair market value at the time of expiration, of your inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to us and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This provision applies only if:
 - (i) The term of the franchise is less than five years; and
- (ii) You are prohibited by the Franchise Agreement or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise, or if you do not receive at least 6 months advance notice of our intent not to renew the franchise.
- (e) A provision that permits us to refuse to renew the franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This provision does not require a renewal provision in the Franchise Agreement or other agreement.
- (f) A provision requiring that arbitration or litigation be conducted outside of Michigan. This does not preclude you from entering into an agreement, at the time of the arbitration, to conduct arbitration at a location outside of Michigan.
- (g) A provision which permits us to refuse to permit a transfer of ownership of the franchise, except for good cause. This provision does not prevent us from exercising a right of first refusal to purchase the franchise. Good cause includes, but is not limited to:
- (i) The failure of the proposed transferee to meet our then-current reasonable qualifications or standards.
 - (ii) The fact that the proposed transferee is a competitor of ours.

- (iii) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.
- (iv) The failure of the franchisee or proposed transferee to pay any sums owing to us or to cure any default in the Franchise Agreement existing at the time of the proposed transfer.
- (h) A provision that requires you to resell to us items that are not uniquely identified with the franchisor. This does not prohibit a provision that grants us a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does it prohibit a provision that grants us the right to acquire the assets of a franchise for the market or appraised value of such assets if you have breached the lawful provisions of the Franchise Agreement and have failed to cure the breach in the manner provided in (c), above.
- (i) A provision which permits us to directly or indirectly convey, assign, or otherwise transfer our obligations to fulfill contractual obligations to you unless provision has been made for providing the required contractual services.

If the franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000, the franchisor shall, at the request of a franchisee, arrange for the escrow of initial investment and other funds paid by the franchisee until the obligations to provide real estate, improvements, equipment, inventory, training, or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENDORSEMENT BY THE ATTORNEY GENERAL.

Any questions concerning this notice should be directed to:

Michigan Department of Attorney General Consumer Protection Division Attention: Franchise Unit G. Mennen Williams Building, 1st Floor 525 W. Ottawa Street Lansing, Michigan 48933 Telephone Number: (517) 373-7117

Note: Despite subparagraph (f) above, we intend to fully enforce the arbitration provisions of the franchise agreement and the area development agreement. Paragraph (f) is preempted by federal law and cannot preclude us from enforcing these arbitration provisions. We will seek to enforce this section as written.

THE MICHIGAN NOTICE APPLIES ONLY TO FRANCHISEES WHO ARE RESIDENTS OF MICHIGAN OR LOCATE THEIR FRANCHISES IN MICHIGAN.

RUSTY TACO FRANCHISING, LLC FRANCHISE DISCLOSURE DOCUMENT TABLE OF CONTENTS

ITEM 1 TH	E FRANCHISOR AND ANY PARENTS, PREDECESSORS, AND AFFILIATES	2
ITEM 2 BU	JSINESS EXPERIENCE	4
ITEM 3 LI	ΓΙGATION	5
ITEM 4 BA	NKRUPTCY	5
ITEM 5 IN	ITIAL FEES	5
ITEM 6 OT	`HER FEES	6
	TIMATED INITIAL INVESTMENT	
ITEM 8 RE	STRICTIONS ON SOURCES OF PRODUCTS AND SERVICES	13
ITEM 9 FR	ANCHISEE'S OBLIGATIONS	15
	NANCING	
	RANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING	
	ERRITORY	
	RADEMARKS	
	ATENTS, COPYRIGHTS, AND PROPRIETARY INFORMATION	
	BLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISED BUSI	
	ESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL	
	ENEWAL, TERMINATION, TRANSFER, AND DISPUTE RESOLUTION	
	UBLIC FIGURES	
	NANCIAL PERFORMANCE REPRESENTATIONS	
	UTLETS AND FRANCHISEE INFORMATION	
	INANCIAL STATEMENTS	
	ONTRACTS	
ITEM 23 R	ECEIPTS	41
EXHIBITS		
Exhibit A	Financial Statements	
Exhibit B	Form of Area Development Agreement	
Exhibit C	Form of Franchise Agreement	
Exhibit D Exhibit E	Sample General Release List of Franchisees (Current Franchisees and Franchisees Who Have Left the System	m or Not
LAMOR E	Communicated with Franchisor)	III OI 1 1 01
Exhibit F	Table of Contents for Operations Manual	
Exhibit G	List of State Administrators/Agents for Service of Process	
Exhibit H	State Addendum and Riders	
Exhibit I	Form of Representation & Acknowledgement Statement	
Exhibit J	Receipts	

ITEM 1 THE FRANCHISOR AND ANY PARENTS, PREDECESSORS, AND AFFILIATES

To simplify the language in this franchise disclosure document (the "Disclosure Document"), we refer to Rusty Taco Franchising, LLC, the franchisor, as either "Franchisor" or "us," and we refer to the person or entity buying the franchise as either "franchisee" or "you." If you are not a natural person, we require all natural persons having at least a 10% direct or indirect ownership interest in you to guarantee and assume your obligations, so when describing those obligations, "you" includes your owners who are also guarantors of your obligations.

The Franchisor and Agents for Service

We are a Delaware limited liability company formed on October 6, 2022. Our principal business address is 5412 W. Plano Pkwy., Suite 100, Plano, Texas 75093, and our phone number is (949) 812–7817. Our agents for service of process are listed in Exhibit G. We do business only under our corporate name and under the name "Rusty Taco[®]." We began offering franchises for Rusty Taco-branded restaurants (each, a "Rusty Taco Restaurant") on May 5, 2023. Our only business is, and since our formation has been, offering and selling franchises for Rusty Taco Restaurants and providing services to our franchisees. We have not operated the kind of business you will operate or any other kind of business, and we have not offered franchises in any other line of business.

Our Parent, Predecessor, and Affiliates

Our parent is Rusty Taco Parent LLC ("Rusty Taco Parent"). Rusty Taco Parent's parent is GCP RT, LLC ("GCP RT"), and GCP RT's parent is Gala Family, LP. The principal address of Rusty Taco Parent, GCP RT, and Gala Family, LP is 3191 Red Hill, Suite 200, Costa Mesa, CA 92626. Rusty Taco Parent owns the trademarks and other intellectual property used in the operation of Rusty Taco Restaurants and has granted us a license to use them and sublicense their use by our franchisees. None of Rusty Taco Parent, GCP RT or Gala Family, LP has ever operated the kind of business you will operate or offered or sold franchises in this or any other line of business.

Our predecessor is Rusty Taco, Inc. ("Predecessor"), a Minnesota corporation formed August 11, 2014. Its last known principal address is Three Glenlake Pkwy NE, Atlanta, Georgia 30328. It sold franchises for Rusty Taco restaurants from May 2015 until December 2022 when its assets, including all of the then-existing franchise agreements for Rusty Taco Restaurants, were acquired by GCP RT and assigned to us. Note that Rusty Taco Restaurants operated as "R Taco" from September 2015 through August 2018. While Predecessor's affiliates owned and operated Rusty Taco Restaurants, Predecessor never did, nor did it own any other business or grant franchises in any other line of business.

Our affiliate, RT Gift Card LLC, an Arizona limited liability company, shares our principal business address and has been established to manage the gift card program that is used in Rusty Taco Restaurants. RT Gift Card LLC has never owned or operated a Rusty Taco Restaurant or offered franchises in this or any other line of business.

We have two affiliates who offer franchises in other lines of business:

Dunn Bros Franchising, LLC, a Delaware limited liability company whose address is 201 Third Avenue S, Minneapolis, Minnesota 55401, has offered franchises for Dunn Brothers Coffeebranded coffee shops since November 2022. As of the end of its most recent fiscal year (December 31, 2024), there are 48 Dunn Brothers Coffee franchises open (which includes franchises that were sold by our affiliate's predecessor from April 2001 to June 2022).

MOOYAH Franchising LLC, a Delaware limited liability company whose address is 6865 Windcrest Drive, Suite 400, Plano, Texas 75024, has offered franchises since April 2017 for MOOYAH-branded restaurants that sell primarily burgers, shakes and fries. As of the date of its most recent fiscal year (December 31, 2024), there are 71 MOOYAH franchises open (which includes franchises that were sold by our affiliate's predecessor from June 2006 to April 2017) and 13 additional franchises sold but not yet open.

Except as described above, we have no parents, predecessors, or affiliates required to be disclosed in this Item 1.

The Franchise Opportunity

We offer franchises for the right to develop, own and operate Rusty Taco Restaurants at specific locations that we approve. Each Rusty Taco Restaurant is identified by the Rusty Taco® trademark and other trademarks, service marks and commercial symbols and trade names that we may authorize you to use (collectively, the "Marks") and offers freshly prepared food made with quality, fresh ingredients inspired by the flavors of Mexico and Texas, along with an assortment of alcoholic and non-alcoholic beverages, in an atmosphere designed to be a neighborhood gathering place. Each Rusty Taco Restaurant is subject to and governed by a franchise agreement that is applicable only to that particular restaurant. The form of Franchise Agreement you would sign is attached as Exhibit C to this Disclosure Document. We refer to the Rusty Taco Restaurant that you will operate as "your Restaurant".

We also offer the opportunity to enter into a Development Agreement under which you would agree to enter into a specific number of franchise agreements and open Rusty Taco Restaurants in accordance with each franchise agreement at locations we approve within an agreed upon development area ("Development Area") and in satisfaction of an agreed upon development and opening schedule (the "Development Schedule"). We will only enter into a Development Agreement if you commit to opening at least 2 Rusty Taco Restaurants, but the number of Rusty Taco Restaurants you will be required to open will be subject to your and our agreement before you sign the Development Agreement. The form of Development Agreement you would sign is attached as Exhibit B to this Disclosure Document. If you don't sign the first Franchise Agreement when you sign the Development Agreement, you will sign the form of franchise agreement we are using to grant franchises when we approve the proposed location for your Restaurant, which could be materially different than the form attached to this Disclosure Document.

Market and Competition

The products and services sold by Rusty Taco Restaurants are sold to the general public. The restaurant business, and in particular the market for tacos and other Mexican food and Tex-Mex restaurants, is highly competitive and well developed. Your Restaurant will compete for customers, employees and real estate with other restaurants offering the same or similar products and services and with other restaurants offering different products but in a similar kind of neighborhood gathering-place environment, including those owned by us, our affiliates, and our or their franchisees.

Industry Specific Regulations

A wide variety of federal, state and local laws, rules and regulations have been enacted that may impact the operation of your Restaurant, and may include those which: (a) establish general standards, zoning, permitting restrictions and requirements and other specifications and requirements for the location, construction, design, maintenance and operation of restaurants; (b) set standards pertaining to employee health and safety; (c) regulate matters affecting the health, safety and welfare of your customers, such as general health and sanitation requirements for restaurants and laws and regulations relating to access by persons with disabilities; employee practices concerning the storage, handling, cooking and preparation of food; restrictions on smoking; availability of and requirements for public accommodations and

requirements for fire safety and general emergency preparedness; (d) establish requirements for food identification and labeling; (e) regulate advertisements and pricing; (f) require disclosures on menus and other collateral regarding nutritional information for the food products offered. State and local agencies inspect restaurants to ensure that they comply with these laws and regulations; and (g) regulate liquor licensing. Compensation of restaurant employees (including state and local minimum wage, hour, and overtime requirements) is governed by both federal and state laws. The U.S. Food and Drug Administration, the U.S. Department of Agriculture, and various state and local departments of health and other agencies have laws and regulations concerning the preparation of food, display of nutrition facts, and sanitary conditions of restaurant facilities. You should seek advice from your professional advisors regarding the requirements of these and other laws that might apply to your Restaurant.

ITEM 2 BUSINESS EXPERIENCE

Anand Gala – Chairman and Chief Executive Officer

Anand Gala has been our Chairman and Chief Executive Officer since December 2022. He also serves in the following roles: (i) President and Chief Executive Officer of Gala Holdings International, Inc. since April 2010; (ii) Managing Partner at Gala Capital Partners, LLC since May 2015; (iii) Managing Partner at Gala Development Partners, LLC since March 2007; (iv) Chairman of MOOYAH Franchising, LLC since April 2017; and (v) Chairman and Chief Executive Officer of Dunn Bros Franchising, LLC since June 2022. He formerly served as the Chief Executive Officer of On Smile LLC from March 2021 to October 2022 (but continues to serve as a member of its Board of Directors since March 2021), and a member of the Board of Directors at Famous Dave's America, Inc. from July 2015 to July 2021. Mr. Gala works primarily from Costa Mesa, California.

<u>Daniel Smith – President</u>

Mr. Smith has been our President since May 2024. Previously, Mr. Smith served as the Chief Operating Officer of Hopdoddy Burger Bar from October 2017 to January 2024 in Austin, Texas; and he was not employed from February 2024 to May 2024. Mr. Smith is based in Plano, Texas.

Denise Fenton – Brand Director

Ms. Fenton has been our Brand Director since December 2022. She is a founding partner of the Rusty Taco brand and has worked with the brand since April of 2010, including as our Predecessor's Brand Director from July 2013 to December 2022. Ms. Fenton is based in Dallas, Texas.

Shane Brewer – Senior Director of Operations

Mr. Brewer has been our Senior Director of Operations since December 2022. From February 2011 to December 2022, he was employed by our Predecessor and its predecessor in various roles such as the Senior Director of Operations from January 2019 to December 2022, as the National Training Manager from September 2015 to January 2019, as the Director of Operations from June 2013 to September 2015, and as the Restaurant General Manager from February 2011 to June 2013. Mr. Brewer is based in Dallas, Texas.

Natalie Johnson - Director of Marketing

Ms. Johnson has been our Director of Marketing since August 2024. Prior to that, Ms. Johnson served Raising Cane's Chicken Fingers in Plano, Texas from September 2019 to August 2024 in various roles such as the Senior Marketing Specialist from March 2023 to August 2024, as the Marketing Specialist from April 2021 to March 2023, and as the Training & Development Coordinator from September 2019 to April 2021. Ms. Johnson is based in Plano, Texas.

Patricia Perry – Director of Franchise Development and Sales (Gala Capital)

Patricia Perry has served as Director of Franchise Sales & Development for Gala Capital since April 2024, and in that role, she provides franchise development and sales assistance to us and our affiliates, Dunn Bros Franchising, LLC and MOOYAH Franchising LLC. Prior to April 2024, Ms. Perry served as Senior Vice President Franchise and License of Bagel Brands from May 2022 to December 2023 and as Vice President of Franchise Development, Business Gifting, and CPG of Edible Brands from February 2019 to May 2022. She was between positions from December 2023 until April 2024. She is currently based in Atlanta, Georgia.

ITEM 3 LITIGATION

No litigation is required to be disclosed in this Item.

ITEM 4 BANKRUPTCY

No bankruptcy is required to be disclosed in this Item.

ITEM 5 INITIAL FEES

Initial Franchise Fee

When you sign a Franchise Agreement, you will pay us an initial franchise fee of \$30,000. This fee is payable in a lump sum and is not refundable. This fee is uniformly imposed on all franchisees.

Training Fee

We do not charge any training fee for providing training to up to 3 people (one of which must be your Operating Principal (as defined in Item 15) or Restaurant Manager (as defined in Item 15), if applicable) of your or your affiliates' first Rusty Taco Restaurant. However, we may charge a training fee for providing training to the required trainees of your or your affiliates' second and each subsequent Rusty Taco Restaurant if we determine that you or your affiliates are unable to properly conduct the required training and, in that event, may charge you a non-refundable training fee of \$8,000, which is payable in lump sum and uniformly imposed.

Development Fee

If and when you sign a Development Agreement you will pay us a development fee equal to \$15,000 for each Rusty Taco Restaurant you agree to develop pursuant to the Development Agreement, and therefore, your total development fee will be \$30,000 to \$75,000 (with a minimum development commitment of 2 and a maximum development commitment of 5 Rusty Taco Restaurants). The development fee is payable in a lump sum, is not refundable, and is determined in the same manner for all franchisees who sign Development Agreements.

VetFran Program

Currently, we participate in the International Franchise Association's VetFran Program under which we provide qualified franchisees with a 50% discount on the standard initial franchise fee of \$30,000 and a development fee equal to \$15,000 each Rusty Taco Restaurant you agree to develop pursuant to the Development Agreement. We reserve the right to discontinue our participation in the VetFran Program at any time.

ITEM 6 OTHER FEES

TYPE OF FEE ¹	AMOUNT	DUE DATE	REMARKS
Royalty Fee	5% of Gross Sales	Weekly (currently on Wednesday)	Day of week is subject to change. See Note 2 for the definition of Gross Sales, Royalty Fee on revenue derived from sales of alcoholic beverages, and description of our current incentive program.
Brand Fund Contribution	2% of Gross Sales; may change subject to the Marketing Cap	Weekly (currently on Wednesday)	We may, upon 60 days notice to you, increase the amount you must contribute to the Fund (as defined in Item 11); however, we cannot require you to contribute to the Fund (as defined in Item 11), spend on local marketing, and contribute to the Local Advertising Co-Op (as defined in Item 11) (collectively, the "Required Marketing Spend") an amount that exceeds 6% of Gross Sales, in the aggregate (the "Marketing Cap"). See Notes 2 and 3.
Local Advertising Co-Op	Currently not assessed	As specified by the Co-Op	We may require you to pay this amount to us under certain circumstances. Your Local Advertising Co-Op contribution may change subject to the Marketing Cap. See Note 4.
Interest	Lesser of 18% per year or highest rate allowed by your Restaurant's jurisdiction	As incurred	We may charge interest on all overdue amounts.
Non-Sufficient Funds Fee	\$50 for each returned check or draft plus reimbursement of our costs incurred to collect	As incurred	Payable if any check or draft, electronic or otherwise, is returned for insufficient funds.
Additional or Remedial Training and Opening Assistance for Second and each Subsequent Rusty Taco Restaurant	\$400 per trainer per day, plus reimbursement of our expenses including the related costs of travel, lodging, and meals if such training is provided at your Restaurant	Before attending the training program	You must pay us our then-current fee for, and reimburse our related cost (including the related costs of travel, lodging, and meals) of, providing any training or opening assistance that we are not required to provide. During the term of your Franchise Agreement, we may revise this amount upon notice to you.
Private or Public Offering Fee	Our costs and expenses	As incurred	If you are not a natural person and you intend to issue equity interests pursuant to a public or private offering, you must first obtain our written consent, which consent will not be unreasonably withheld. We limit our review to the manner in which the offering materials treat our relationship with you.

TYPE OF FEE ¹	AMOUNT	DUE DATE	REMARKS
Transfer Fee	\$5,000, plus reimbursement of reasonable attorneys' fees and our expenses to inspect your Restaurant, which generally range from \$1,000 to \$2,000	As incurred	No transfer fee is payable for transfer by devise or inheritance.
Renewal Fee	The higher of \$7,500 or 25% of our then-current initial franchise fee	On execution of successor Franchise Agreement	Payable if we approve you to acquire a successor franchise for your Restaurant.
Indemnification (Franchise Agreement and Development Agreement)	Our actual costs	As incurred	You must defend and indemnify us and our related parties against certain claims relating to the development and operation of your Restaurant and claims arising from activities conducted at your Restaurant.
Audit Fee	Our actual costs	As invoiced	Payable if an audit shows you have understated Gross Sales by 2% or more.
Reimbursement of Expenses	Our actual costs plus an administrative fee which will not exceed 10% of the expense we incur on your behalf.	As incurred	You must reimburse our cost if we incur any expenses on your behalf (including amounts we pay on your behalf to technology vendors) or due to your failure to comply with the Standards, including, your failure to maintain required insurance coverages, pay taxes to governmental authorities, pass our inspections, etc.
Technology Fee	Currently not assessed but expected to be an amount in the range of \$250 to \$1,000 per month, plus to the actual technology and related out of pocket expense, which will be passed on to our franchisees	As incurred	In addition to paying us the technology fee, we you may required to directly purchase or lease certain technology products and related services directly from our designated or approved third-party vendors.
Email Address Fee	Our actual cost but currently not charged	As incurred	We may charge you a recurring fee for providing you a branded email address.
Learning Management System Fee	Currently not assessed but expected to be an amount up to \$1,000 per month	As agreed	You must participate in the Cornerstone Learning Management System to access certain training materials. We forward the payment to Cornerstone, our current designated vendor for the system. This system provides the delivery of electronic content, knowledge checks and certification

TYPE OF FEE 1	AMOUNT	DUE DATE	REMARKS
			exams. We may begin charging this fee upon written notice to you.
Relocation Expenses	Reimbursement of our costs and expenses (estimated to be up to \$5,000)	As incurred	This includes reimbursement of our reasonable costs and expenses (including attorneys' fees) associated with the relocation
De-Identification Fee	Reimbursement of actual costs	As incurred	Upon termination or expiration of the Franchise Agreement if you fail to deidentify your Restaurant's premises then we may enter the Restaurant premises and take all actions necessary to de-identify the premises as a Rusty Taco Restaurant. You must pay these costs we incur in our deidentification efforts.
Unauthorized Temporary Closure	110% of the average daily Royalty Fee owed for the week immediately prior to closure	As invoiced	Charged if your Restaurant does not generate Gross Sales because you have temporarily closed your Restaurant without our consent, for each whole or partial day your Restaurant is closed. This amount is payable in the same manner as Royalty Fee. Payment will not act as a cure of the unauthorized closure default and will not impair any other rights we have under the Franchise Agreement.
Non-Compliance Charge	Increase of one point to the rate of your Royalty Fee	As incurred	If you do not comply with your obligations under the Franchise Agreement, your Royalty Fee will be increased to 6% of Gross Sales until we have determined that you have cured your deficiencies. This increase to your Royalty Fee is intended to compensate us for additional expenses and certain losses that we will incur as a result of your non-compliance and is not a penalty or an expression of the total amount of such damages. Payment of this increase in Royalty Fee does not cure the non-compliance that triggered its payment.
Approval of Alternate Suppliers	\$2,500 to \$10,000	As incurred	Before approving a supplier at your request, we may charge you a reasonable fee and reimburse the costs and expenses incurred by us in evaluating your proposed supplier.

TYPE OF FEE ¹	AMOUNT	DUE DATE	REMARKS
Lost Revenue Damages	Varies	As incurred	Payable if we terminate the Franchise Agreement for your breach, or you terminate it without cause. See Note 5 for how lost revenue damages will be calculated.
Costs and Attorneys' Fees	Our actual costs	As incurred	Payable only if we are the prevailing party in any relevant litigation or arbitration.

Notes:

- 1. All amounts described above are collected by, and payable to, us via electronic funds transfer or such other means that we may designate from time to time. Fees paid by our existing franchisees may differ from the amounts described above. We may also offer various incentive programs from time to time pursuant to which eligible franchisees may temporarily be eligible to pay a reduced Royalty Fee.
- Definition of "Gross Sales": "Gross Sales" means the total revenue attributable to, and receipts from, all services and products and all income of every other kind and nature related to your Restaurant, whether or not in compliance with the Franchise Agreement, including income related to catering operations and special events and the full value of meals provided to your bona fide employees as a benefit of their employment (except you may deduct from Gross Sales the value of any employee discounts that are given during the week in which the meals are provided), whether for cash or credit, barter or exchange, and regardless of collection in the case of credit. Gross Sales does not include (a) sales (or other similar) taxes that you collect from your customers, if you transmit them to the appropriate taxing authority, promotions, voids, and discounts; (b) proceeds from isolated sales of trade fixtures that are not part of the products and services you offer and that do not have any material effect on the operation of your Restaurant; (c) tips or gratuities that Restaurant customers pay directly to your employees or to you which you then turn over to these employees in lieu of direct tips or gratuities; or (d) returns to shippers or manufacturers. Revenue from the purchase or redemption of gift certificates, gift cards, loyalty or similar programs is calculated as part of Gross Sales in accordance with our then-current guidelines for such programs. Gross Sales also includes all insurance proceeds you receive to replace revenue that you lose from the interruption of the operation of your Restaurant.

Alcohol Sales: If we are unable to collect the Royalty Fee or other amounts derived from the sale of alcohol products, then we reserve the right to modify your payment obligations under the Franchise Agreement and amend the applicable provisions hereunder in order to provide the same economic effect to both you and us contemplated by your Franchise Agreement as if such law had not taken effect (including, for example, by increasing the Royalty Fee and other payments owed on non-alcohol sales). In such event, you agree to execute the documents in the form we require to give effect to such changes.

3. It is your responsibility to provide us with written notice if the Required Marketing Spend exceeds the Marketing Cap, and until we receive your written notice (the "Marketing Notice"), you will fully comply with the Required Marketing Spend, and no excess amounts will be refunded to you. If the Required Marketing Spend exceeds the Marketing Cap, you may, after we receive your Marketing Notice, reduce the local advertising expenditure, but only to the extent and for the time necessary to not exceed the Marketing Cap. You must immediately return to full compliance with Required Marketing Spend once the Marketing Cap is no longer exceeded. We will not count towards your Marketing Cap the cost of free or discounted products or services, coupons, special offers or price reductions that you provide as a promotion, signs, personnel salaries, administrative costs, employee incentive programs, or other amounts that we, in our reasonable judgment, deem inappropriate for meeting the Marketing Cap.

- 4. We may periodically review your books and records and require you to submit reports to determine your Fund contribution, Local Advertising Co-Op contributions, and Local Advertising expenses. If you fail to spend (or prove that you spent) the required amount on Local Advertising, then we may, along with our other rights, require you to pay us the shortfall as an additional Fund contribution or to pay us the shortfall for us to spend on Local Advertising for your Restaurant. Each Rusty Taco Restaurant participating in a Local Advertising Co-Op, including those owned by us or our affiliates, will have one vote on all Co-Op decisions requiring a vote of participants.
- 5. Equal to the net present value of: (1) the lesser of 36 or the number of calendar months remaining in the franchise term absent the termination, multiplied by (2) the sum of the Royalty Fee and Fund contribution percentages in effect as of the termination date, multiplied by (3) the average monthly Gross Sales of your Restaurant during the 24 full calendar months immediately preceding the earlier of the termination date of the Franchise Agreement or the last date the date on which your Restaurant operated in compliance with the Franchise Agreement, minus (4) any cost savings we experienced as a result of the termination. If, as of the termination date of the Franchise Agreement or the last date on which your Restaurant operated in compliance with the Franchise Agreement, your Restaurant had not commenced operations or had operated for less than 24 months, the average monthly Gross Sales will equal the 24-month average Gross Sales of all Rusty Taco Restaurants that had operated for the full 24 calendar months immediately preceding termination of the Franchise Agreement.

ITEM 7 ESTIMATED INITIAL INVESTMENT

YOUR ESTIMATED INITIAL INVESTMENT – FRANCHISE AGREEMENT

TYPE OF EXPENDITURE	AMOUNT	METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE
Initial Franchise Fee	\$30,000	Lump Sum	On signing the Franchise Agreement	Us
Rent and Security Deposit ¹	\$3,000 to \$10,000	As Arranged	As Arranged	Landlord
Utility Deposits	\$6,000 to \$10,000	Lump Sum	Prior to Opening	Utility Companies
Construction, Remodeling and Leasehold Improvements ²	\$253,000 to \$621,000	As Incurred	As Arranged	Landlord or Contractor
Permits and Licenses ³	\$3,000 to \$6,950	As Incurred	Prior to Opening	Government Agencies
Architect Fee	\$20,000 to \$60,000	As Incurred	Prior to Opening	Designated Architect
Furniture, Fixtures, Equipment, and Signage	\$163,000 to \$250,000	As Incurred	Prior to Opening	Third-Party Suppliers
Initial & Opening Training Costs ⁴	\$0 to \$15,000	As Arranged	As Invoiced	Employees and Third- Party Suppliers
Computer System	\$2,000 to \$5,000	As Arranged	As Arranged	Third-Party Suppliers

TYPE OF EXPENDITURE	AMOUNT	METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE
Initial Inventory/Supplies ⁵	\$7,400 to \$10,000	As Arranged	As Arranged	Third-Party Suppliers
Professional Services	\$5,000 to \$15,000	As Arranged	As Arranged	Accountants, Lawyers, Architects, etc.
Initial Advertising and Marketing Expenses ⁶	\$10,000 to \$15,000	As Arranged	As Arranged	Third-Party Suppliers
Insurance ⁷	\$6,000 to \$30,000	As Arranged	As Arranged	Insurance Broker
Additional Funds (3 months) ⁸	\$20,000 to \$50,000	As Arranged	As Arranged	Third Parties
TOTAL	\$528,400 to \$1,127,950			

Notes:

- 1. Rusty Taco Restaurants are typically located in commercially zoned shopping or entertainment areas in premises that are typically leased rather than purchased. This range reflects the estimate of the security deposit required by the landlord and the first month's rent. See note to "Additional Funds" below with respect to rent for additional months. This Item is based on a facility containing approximately 1,500 to 2,500 square feet in a suitable free-standing building or an endcap in an in-line strip center.
- 2. The low end of this estimate assumes renovation of a 1,500 square foot existing restaurant space in "ready to occupy" condition. The range is impacted by various factors including size of space, scope of tenant improvement work, condition, location of the space and price differences among various suppliers and contractors. This estimate does not reflect any tenant improvement allowance that you negotiate with the landlord.
- 3. You will be required by certain government agencies to secure permits and licenses, including a liquor license, before opening your Restaurant. This estimate includes a typical cost for securing a liquor license, but those costs can vary significantly depending on the state or municipality that issues them and where the number of licenses is severely restricted or available only from an existing holder. You should retain legal counsel specialized in obtaining and maintaining liquor licenses.
- 4. We do not charge any training fee for providing training to up to 3 people (one of which must be your Operating Principal or Restaurant Manager, if applicable) of your or your affiliates' first Rusty Taco Restaurant. However, we may charge a training fee for providing training to the required trainees of your or your affiliates' second and each subsequent Rusty Taco Restaurant if we determine that you or your affiliates are unable to properly conduct the required training and, in that event, may charge you a non-refundable training fee of \$8,000 (See Item 5). This range also reflects the out-of-pocket costs estimated to be incurred in sending trainees to our training program (including, for example, travel, lodging and meals). The low end of the range assumes that the training is provided virtually or the initially required trainees are owner/operators who do not draw a salary and reside near a training facility. If you send salaried employees to our initial training program, you may be required to pay wages and other benefits to those employees while they are training depending on your state's law.
- 5. This estimate includes the cost of food, beverages, condiments, packaging, and other supplies for approximately the first 2 to 10 days of operations. Local costs will greatly affect these costs or expenses.

- 6. Unless we agree otherwise, you must conduct a grand opening marketing campaign for your Restaurant and spend at least \$10,000 on such campaign in accordance with our written specifications. Your grand opening marketing campaign will take place in the 45 days prior to, and 60 days following, the opening of your Restaurant. You must obtain our approval of all advertising items, methods, and media.
- 7. This amount represents an estimated cost of your annual insurance premiums. Your cost of insurance may vary depending on the insurer, the location of your Restaurant, your claims history, and other factors.
- 8. This range reflects other required expenses you will likely incur in the 3 months after opening (including an additional 2 months' rent), net of operating revenue estimated to be generated by the operation of your Restaurant during that period.

The estimates for the expenses shown in this Item 7 generally are based on a Rusty Taco Restaurant that is typical in size and type of location and is based on our management's experience in the Rusty Taco Restaurant System and on information we have gathered from recent Rusty Taco Restaurant openings. All amounts payable to us or our affiliates are not refundable. Refundability of any other amounts will be subject to your agreement with the party to whom the amounts are owed. We and our affiliates do not provide financing for any part of this investment. The estimates shown in this chart do not include any finance charges, interest, or debt service obligations.

YOUR ESTIMATED INITIAL INVESTMENT – DEVELOPMENT AGREEMENT

TYPE OF EXPENDITURES	AMOUNT	METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE
Development Fee ¹	\$30,000 to \$75,000	Lump Sum	Upon Signing the Development Agreement	Us
TOTAL	\$30,000 to \$75,000			

Note:

- 1. If and when you sign a Development Agreement you will pay us a development fee equal to \$15,000 for each Restaurant you agree to develop Pursuant to the Development Agreement. However, if you qualify for our Growth Incentive Program, sign a Development Agreement pursuant to the Growth Incentive Program, and you are in compliance with your development obligation under your Development Agreement, then (i) the development fee will be \$30,000 if the Development Schedule requires you to develop up to 3 Rusty Taco Restaurants, and (ii) for each additional Rusty Taco Restaurant that you commit to develop, the development fee will increase by an amount equal to \$10,000 multiplied by the number of additional Rusty Taco Restaurants that you agree to develop. For example, if you agree to develop a total 4 Rusty Taco Restaurants, your development fee will be \$40,000; and if you agree to develop a total 5 Rusty Taco Restaurants, your total development fee will be \$50,000. The development fee is payable in a lump sum, is not refundable, and is determined in the same manner for all franchisees who sign Development Agreements
- 2. This is the entire investment required to acquire development right, and it does not include the initial investment estimated to be incurred to open a Rusty Taco Restaurant. Please refer to the preceding table titled "Your Estimated Initial Investment Franchise Agreement" for a description of that investment.

ITEM 8 RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

Standards

In order to maintain the quality and uniformity of the services and products offered and sold at Rusty Taco Restaurants, you must purchase or lease all products and services for your Restaurant according to our mandatory standards and specifications for the development and operation of Rusty Taco Restaurants (the "Standards"), which we may change from time to time. You must follow all our then current Standards governing the development and operation of your Restaurant, and you also must use in your Restaurant only those food and beverage items, condiments, construction and decorative materials, equipment, software, signage, fixtures, furnishings, supplies and other products and services that we designate from time to time as Operating Assets (collectively, the "Operating Assets") that we have approved. We will not issue to you or to any approved suppliers (except as we deem necessary for purposes of production) the specifications for proprietary Operating Assets, if any. We will otherwise communicate our Standards and the list of approved Operating Assets to you in our operating and other technical manuals that we develop and modify, in writing, from time to time (collectively, the "Operations Manual").

Required Products & Services; Approved or Designated Suppliers

You must purchase or lease all Operating Assets from those suppliers or distributors that we designate or approve, which may include or be limited to us or our affiliates. As of the issuance date of this Disclosure Document, we either approve or have designated third-party suppliers for the following products and services: (a) fixtures, furniture, equipment (including computer hardware and software), interior and exterior signage, graphics, decor, trade dress, and restaurant design consulting services; (b) food products, ingredients, and condiments; (c) beverages; (d) uniforms, shirts, memorabilia, and merchandise and items intended for retail sale (whether or not bearing our Marks); (e) advertising, point-of-purchase materials, and other printed promotional and marketing materials; (f) gift certificates and restaurant value cards; (g) stationery, business cards, contracts, and forms; (h) bags, packaging, and supplies bearing our Marks; (i) third-party delivery services; and (j) catering related services. We may in the future require you to purchase other goods and/or services from our designated or approved suppliers. We will communicate information concerning approved and designated suppliers and distributors via the Operations Manual.

As of the issuance date of this Disclosure Document, neither we nor any of our affiliates are designated or approved suppliers for any products or services. None of our officers own any ownership in any privately held suppliers or a material interest in any publicly held suppliers of Operating Assets.

If you want to use any Operating Assets for or at your Restaurant that we have not yet evaluated or from a supplier that we have not yet approved, you first must submit sufficient information, specifications, samples, and other information as we require for us to determine whether the Operating Asset complies with our Standards, or the supplier meets our criteria. We generally review any request for an additional product or supplier within 30 days. In order to evaluate your proposed alternate supplier, you must pay us our then-current supplier evaluation fee and reimburse the costs incurred by us in evaluating a proposed new vendor/supplier (the total of which will be in the range of \$2,500 to \$10,000 (See Item 6)). We may also impose limits on the number of suppliers, products and services that we are willing to review. We maintain a list of criteria for reviewing and approving products, services, and suppliers; however, we do not issue these criteria to you. We may, with or without cause, revoke our approval of any supplier or product at any time.

Insurance

You must obtain and maintain insurance policies from a company with a minimum A.M. Best Rating of A-VII that we accept and that is underwritten by insurers licensed and admitted to write coverage in the state in which your Restaurant is located. All policies: (a) must name us and other persons or entities that we designate from time to time as additional insureds on a primary, non-contributory basis; and (b) must provide severability of interests and/or separation of insureds coverage.

As of the issuance date of this Disclosure Document, you must maintain at least the following insurance coverage: (a) property insurance equal to the full replacement value of, and covering, your Restaurant, Restaurant improvements, and all fixtures, equipment, supplies, and other property used in the operation of your Restaurant; (b) business interruption insurance that covers 12 months of your loss of income and 12 months of Royalty Fees owed to us; (c) comprehensive general liability insurance, including but not limited to premises, products, and contractual liability, in an amount of \$1,000,000 per occurrence and \$2,000,000 in the aggregate on an annual, per location basis; (d) liquor liability in an amount of \$1,000,000 per occurrence and \$1,000,000 in the aggregate on an annual, per location basis; (e) automobile liability, including coverage of all non-owned, rented, and hired vehicles in an amount of \$1,000,000 per occurrence, combined single limit coverage; (f) workers' compensation insurance within state, statutory requirements, covering all employees; (g) employment practices liability insurance in amounts of \$1,000,000 per accident, \$1,000,000 in the aggregate, and \$1,000,000 for disease; and (h) umbrella liability in an amount of \$1,000,000 per occurrence and \$1,000,000 in the aggregate on an annual basis. We may modify the minimum coverage requirements and require additional insurance as market or industry conditions warrant. We will provide you with written notice of these changes.

We must receive at least 30 days' prior written notice of material amendment, termination, expiration or cancellation of any insurance policy and you must submit to us a copy of the certificate or other evidence we require of the issuance, renewal or extension of each insurance policy before beginning operations for your Restaurant and each subsequent year. Premiums are payable to your insurers in amounts and at times as your insurers require.

Revenue Derived from Franchisee Purchases and Leases

During fiscal year 2024, neither we nor our affiliates received any revenue from direct sale of products or services to Rusty Taco Restaurant franchisees; however, we received \$27,685 (1.3% of our total revenue of \$2,131,071) from third-party vendors on account of purchases made by our franchisees. Currently, our food and beverage vendors pay us an amount ranging from \$0.03 to \$6.61 per unit purchased by our franchisees.

Estimated Proportion of Required Purchases and Leases to all Purchases and Leases

We estimate that approximately 90% to 95% of your initial investment and 90% to 95% of your ongoing expenditure to operate your Restaurant will be directed to purchase products and services that will be restricted by us in some manner.

Description of Purchasing Cooperatives; Purchasing Arrangements

Currently, no purchasing or distribution cooperatives exist in the Rusty Taco franchise system. We may negotiate purchase arrangements, including price terms, with certain primary suppliers. If we negotiate a cooperative purchase agreement for the region where your Restaurant is located, you must participate in the purchasing program. We do not provide material benefits to you (for example, renewal or granting additional franchises) based on your purchase of particular products or services or use of particular suppliers.

ITEM 9 FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the Franchise Agreement and other agreements. It will help you find more detailed information about your obligations in these agreements and in other items of this Disclosure Document.

Obligation	Section(s) in Franchise	Section(s) in Development	Item(s) in Disclosure
	Agreement	Agreement	Document
a. Site selection and acquisition/lease	Sections 3.1 and 3.3	Section 1.4	Items 8 and 11
b. Pre-opening purchases/leases	Sections 3.4, 6.5, 6.6, 10.1, and 10.2	Not applicable	Items 5, 6, 7, 8, and 11
c. Site development and other pre-opening requirements	Section 3.4	Article 4	Items 1, 7, 8, and 11
d. Initial and ongoing training	Sections 5.1 and 5.4	Not applicable	Items 6, 7, and 11
e. Opening	Sections 3.5 and 5.2	Section 4.1	Items 7 and 11
f. Fees	Article 4 and Sections 9.3, 9.4, 9.5, 12.2, 12.3, and 12.4	Article 2	Items 5, 6, 7, 8, and 11
g. Compliance with standards and policies/ Operations Manual	Article 8	Not applicable	Items 8, 11, 14, and 16
h. Trademarks and proprietary information	Article 7	Not applicable	Items 11, 13, and 14
i. Restrictions on products/services offered	Sections 6.4, 6.5, 6.6, and 6.7	Not applicable	Items 8 and 16
j. Warranty and customer service requirements	Not applicable	Not applicable	Item 16
k. Territorial development and sales quotas	Section 1.2	Not applicable	Item 12
Ongoing product/service purchases	Sections 6.5, 6.6, and 8.2	Not applicable	Items 8, 11, and 16
m. Maintenance, appearance and remodeling requirements	Sections 6.7 and 8.2	Not applicable	Item 8
n. Insurance	Section 11.2	Not applicable	Items 7 and 8
o. Advertising	Article 9	Not applicable	Items 6, 8, and 11
p. Indemnification	Section 11.3	Section 5.2	Item 6
q. Owner's participation/ management/staffing	Sections 6.2 and 6.3	Not applicable	Items 11 and 15

Obligation	Section(s) in Franchise Agreement	Section(s) in Development Agreement	Item(s) in Disclosure Document
r. Records and reports	Sections 3.5, 10.4, 10.5, and 10.6	Not applicable	Item 11
s. Inspections and audits	Sections 6.9 and 10.7	Not applicable	Items 6 and 11
t. Transfer	Article 12	Article 6	Items 6 and 17
u. Renewal or extension of rights	Sections 2.1 and 2.2	Not applicable	Items 6 and 17
v. Post-termination obligations	Section 15.2	Section 7.7	Item 17
w. Noncompetition covenants	Sections 15.1 and 15.2	Section 4.4	Item 17
x. Dispute resolution	Article 19	Article 11	Item 17
y. Guaranty	Attachment C	Not applicable	Item 15

ITEM 10 FINANCING

We, our affiliates, and our agents do not offer direct or indirect financing. We do not guarantee your notes, leases, or other obligations.

ITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING

Except as listed below, we are not required to provide you with any assistance.

Under the Franchise Agreement, before you open your Restaurant, we or our designee will:

- 1. Review your proposed site for compliance with our site selection guidelines and approve or deny the site and your proposed lease or contract of sale; however, you are solely responsible for conforming the premises of your Restaurant to local ordinances and building codes and obtaining the required permits. We do not typically lease premises to our franchisees (Franchise Agreement, Sections 3.2 and 3.4).
- 2. Provide you with access to approved specifications and layouts for your Restaurant, including requirements or recommendations (as applicable) for dimensions, design, interior layout, decor, decorative materials, fixtures, equipment, furniture, and signs, which you will implement and adapt for your Restaurant. We do not deliver or install any of these items (Franchise Agreement, Section 3.4).
- 3. Provide you with access to our Operations Manual (Franchise Agreement, Section 8.1).
- 4. Provide you with a list of food items, condiments, inventory, supplies, and other products and services that you must use and provide you with a list of any approved suppliers. We provide written specifications for some items in our Operations Manual. We currently do not provide these items directly or deliver or install any of these items (Franchise Agreement, Sections 6.4, 6.5, and 6.6).
- 5. Conduct an initial training program (Franchise Agreement, Section 5.1).

- 6. With regard to your first Restaurant, provide on-site opening assistance at no cost to you (Franchise Agreement, Section 5.2).
- 7. Review your written start-up marketing program that explains how you will spend the grand opening marketing expenses (Franchise Agreement, Section 9.2).

Under the Franchise Agreement, during the operation of your Restaurant, we or our designee will:

- 1. Give you any advice and written materials we may develop on the techniques of managing and operating Rusty Taco Restaurants (Franchise Agreement, Section 5.3).
- 2. Give you updated lists of approved suppliers (Franchise Agreement, Section 6.6).
- 3. Administer the Fund (Franchise Agreement, Section 9.3).
- 4. Provide you access to proprietary software programs (if any) that may be developed by us or on our behalf for use in the System. We reserve the right to charge a reasonable license fee (Franchise Agreement, Section 10.2).

Under the Franchise Agreement, during the operation of your Restaurant, we or our designee may:

- 1. Provide ongoing consultation and offer remedial or additional training (Franchise Agreement, Sections 5.3 and 5.4).
- 2. Subject to applicable laws, periodically set the maximum and minimum price that you may charge or advertise for products and services offered by your Restaurant (Franchise Agreement, Section 6.9).

Before you sign a Development Agreement, you and we will agree on the Development Area within which you will look for Rusty Taco sites (Development Agreement, Section 1.1 and Attachment A), the number of Rusty Taco Restaurants you will be required to open (Development Agreement, Section 1.1 and Attachment A), and the schedule by which you must sign Franchise Agreements and open Rusty Taco Restaurants in accordance with those agreements (Development Agreement, Section 1.1 and Attachment A). You are responsible for locating and presenting to us proposed sites for Rusty Taco Restaurants in the Development Area. We will use reasonable efforts to review and evaluate the proposed sites within 30 days after we receive all requested information and materials to evaluate the site. We will approve the proposed site for any Rusty Taco Restaurant if it meets our then-current standards. If we accept a proposed site, you (or your affiliate) must timely sign a separate Franchise Agreement for the site. (Development Agreement, Section 1.4). We are not required to provide assistance to you prior to the commencement of your activities under the Development Agreement.

During your activities under the Development Agreement, we will provide the following assistance:

- 1. Review and either approve or reject your proposed sites (Development Agreement, Section 1.4).
- 2. Issue Franchise Agreements for execution once sites are approved (Development Agreement, Sections 1.5 and 3.1).

Grand Opening Advertising

Unless we agree otherwise, you must, at least 60 days before your Restaurant's planned opening date, prepare and submit to us for our approval a proposed grand opening advertising program, which complies with our Standards. You must obtain our prior approval of your grand opening advertising program and spend at

least \$10,000 on approved marketing efforts during the period beginning 45 days before and ending 60 days after your Restaurant opens.

Local Advertising

We may require you to spend the amount that we periodically specify, subject to the Marketing Cap, on local advertising, including participation in all advertising, sales and branding promotion programs that we may authorize or develop for Rusty Taco Restaurants ("Local Advertising"). The current Local Advertising spending requirement is 2% of your Restaurant's Gross Sales. You must provide us with an accounting of your Local Advertising expenditures for the time periods we require.

Your advertising, promotion, and marketing must be completely clear, factual, and not misleading and conform to both the highest standards of ethical advertising and marketing and the advertising and marketing policies that we prescribe from time to time. You must submit for our approval samples of all Local Advertising and promotional materials not prepared or previously approved by us. You may not use any advertising or promotional materials that we have not approved or that we have disapproved. You must list and advertise your Restaurant in the business directory listings that we require (which may be traditional book listings or Internet listings) and in the format we require. Your cost for these listings is credited toward your spending requirement on Local Advertising.

Unless approved by us in writing, you may not engage in any promotional or similar activities, directly or indirectly, through or on the Internet or any other similar proprietary or common carrier electronic delivery system.

Brand Fund

Other than administering the Fund as described below, we have no obligation to advertise the Rusty Taco brand or the Rusty Taco Restaurants. However, if we choose to advertise the Rusty Taco brand or the Rusty Taco Restaurants, we may (i) use any form of media with local, regional, or national coverage, and (ii) create such marketing materials in-house or outsource this task to an outside advertising agency.

You must contribute to the Brand Fund ("Fund") currently in an amount equal to 2% of your Restaurant's Gross Sales; however, subject to the Marketing Cap we may change that amount by providing written notice to you. All franchisees must contribute to the Fund as directed, although some franchisees operating under different forms of franchise agreement might contribute at different rates. Although not contractually required to do so, the Rusty Taco Restaurant that we or our affiliates own currently contribute to the Fund on the same basis as our franchisees.

We or our designee will administer the Fund. The Fund is not our asset, and we do not owe any fiduciary obligation to you for administering the Fund or any other reason. We will not use Fund monies for creating or placing any advertisement that is principally a solicitation for new franchisees, but we may include in advertisements prepared using Fund monies (including Internet advertising) information concerning franchise opportunities and we may use a portion of Fund monies to create and maintain one or more pages on our Website devoted to advertising franchise opportunities and identifying and screening inquiries and applications that franchise candidates submit.

We reserve the right to reimburse ourselves or our designee for such reasonable administrative costs and overhead we or our designee incur for activities reasonably related to the administration and direction of the Fund, including the proportionate compensation of our or our designee's employees or agents, who devote time and provide service in the conduct, formulation, development and production of these advertising and promotion programs or the administration of the Fund.

We are not obligated to spend any amount on advertising in the area or territory where your Restaurant is located. We also have no fiduciary obligations regarding the Fund. If Fund contributions are not spent in any fiscal year, the excess amount of Fund contribution will be carried over for future use. We will prepare an annual unaudited statement of the operations of the Fund, and we will make it available to you upon your written request. We are not required to have the Fund statements audited.

We may terminate the Fund at any time. However, we will not terminate the Fund until we spend all monies contributed to the Fund for permitted uses or return them to the contributors in the manner we deem fair and equitable.

Below is the breakdown of the usage of the Fund contributions during the 2024 fiscal year:

Media Placement	65%
Printing and Production	16%
Others (creative development, branding, social media tools, email program,	19%
website, etc.)	
Total:	100%

Local Advertising Co-Ops

As of the issuance date of this Disclosure Document, there is no advertising cooperative in which you must participate. We may designate a geographic area in which 2 or more Rusty Taco Restaurants are located as an area for an advertising cooperative (each a "Local Advertising Co-Op"). The Local Advertising Co-Op members in any area are the owners of all of the Rusty Taco Restaurants operating in that area (including us and our affiliates, if applicable). We will determine how each Local Advertising Co-Op is organized and governed and when it begins operating, and this organization may be a formal, legal entity or may be an informal organization. We may form, change, dissolve, and merge Local Advertising Co-Ops. Each Local Advertising Co-Op's purpose is, with our approval, to administer advertising programs and develop advertising, marketing, and promotional materials for the area that the Local Advertising Co-Op covers. Whether or not the Local Advertising Co-Op is a formal, legal entity, or is an informal organization, you must sign the documents that we require to become a member of the Local Advertising Co-Op and participate in the Local Advertising Co-Op as those documents require. We anticipate that Local Advertising Co-Ops will operate from written governing documents and prepare periodic financial statements that members may review. We have the sole discretion to approve any marketing and advertising materials used by the Local Advertising Co-Ops.

The Local Advertising Co-Op's members will administer the Local Advertising Co-Op, but the amount you will be required to contribute will be subject to the Marketing Cap. All material decisions of the Local Advertising Co-Op, including contribution levels (which also require our approval), will require the affirmative vote of at least 51% of all Rusty Taco Restaurants operating within the Local Advertising Co-Op's area including, if applicable, those that we or our affiliates operate. Each participating Rusty Taco Restaurant receives one vote. All Local Advertising Co-Op members, including us or our affiliates for Restaurants located in the Local Advertising Co-Op's area that we or they own, will contribute at the same approved rate. The Local Advertising Co-Op will determine whether to prepare annual or periodic financial statements.

Franchisee Council

We do not have a Franchise Advisory Council ("FAC"); however, we may form one in the near future. If we form an FAC, we expect that its members will be selected by us and serve at our discretion.

System Websites and Electronic Advertising

We have established a website for the Rusty Taco System ("System Website"). We or one or more of our designees may use the System Website: (a) to advertise, market, and promote Rusty Taco Restaurants and other products they offer, and/or the Rusty Taco Restaurant franchise opportunity; (b) to function as the Intranet; and/or (c) for any other purposes that we determine are appropriate for Rusty Taco Restaurants or the System. If we include information about your Restaurant on the System Website, you must give us the information and materials that we periodically request concerning your Restaurant and otherwise participate in the System Website in the manner that we periodically specify. By posting or submitting information or materials for the System Website, you are representing to us that the information and materials are accurate, are not misleading, and do not infringe any third party's rights.

We own all intellectual property and other rights in the System Website and all information it contains, including the domain name or URL for the System Website, the log of "hits" by visitors, and any personal or business data that visitors (including you and your personnel) supply. We may implement and periodically modify Standards relating to the System Website and may discontinue the System Website, or any services offered through the System Website, at any time.

All advertising, marketing, and promotional materials that you develop for your Restaurant must contain notices of the System Website's URL in the manner that we periodically designate. You may not develop, maintain, or authorize any other website, other online presence, or other electronic medium that mentions or describes your Restaurant or displays any of the Marks without our prior approval. You may not conduct commerce or offer or sell any products or services using any website, another electronic means or medium, or otherwise over the Internet. We have the right to maintain websites other than the System Website and to offer and sell products and services under the Marks from the System Website, another website, or otherwise over the Internet without payment or obligation of any kind to you.

Computer Systems

You must lease, install, and maintain at your Restaurant only the point-of-sale cash registers and computer systems, printers, EMV readers, and other related hardware and software ("Computer System") that meet our Standards. These requirements may include hardware components, dedicated telephone lines, a high-speed, broadband Internet connection at our then-current, minimum bandwidth specification, modems, printers, and other computer-related accessories and peripheral equipment. The cost of the Computer System is in the range of \$2,000 to \$3,000.

The Computer System must at all times store all financial information and other information related to the operation of your Restaurant. We or our affiliates will have no limitations on our ability to independently access your Computer System at all times. We or our affiliates have no obligation to provide ongoing maintenance, repairs, upgrades or updates to your Computer System. However, you must, at your cost, update and upgrade the Computer System to comply with the Standards. There are no contractual limitations on our right to modify the Standards for the Computer System. We estimate that your annual cost of optional or required maintenance, updating, upgrading, or support contracts related to the Computer System will be in the range of \$3,000 to \$7,000 per year.

Training

Shane Brewer, Senior Director of Operations, directs the training program for Rusty Taco Restaurant openings and provides ongoing support. Mr. Brewer has over 2 years of experience with us, over 14 years of experience with the Rusty Taco brand, and over 33 years of experience in the subjects taught. Other employees who have experience with the operation of Rusty Taco Restaurants will assist Mr. Brewer with the development and administration of our training program.

We currently conduct our training on an as-needed basis. The Operations Manual contains the primary instruction material for training, in addition to other relevant materials which we will provide during training.

At least 60 days before opening your Restaurant, at least one person and up to three people (one of which must be your Operating Principal or Restaurant Manager, if applicable) must attend and complete to our satisfaction our initial training program on the general business aspects of operating a Rusty Taco Restaurant. We will provide initial training for no additional fee for up to three people associated with your first Restaurant, including you (or your Operating Principal) and your Restaurant Manager, if applicable, but you will be responsible for all travel and living expenses and related costs that your personnel incur. All persons in your group who attend training must do so at the same time. For your or your affiliates' second and subsequent Rusty Taco Restaurant, we may require you to conduct the initial training program for your trainees who have not previously completed the program, but we reserve the right to conduct it ourselves if we determine that you are unable to properly do so. In that instance, you must pay us a training fee of \$8,000 (See Item 5).

The initial training program consists of classes conducted at one of our company-owned Rusty Taco Restaurants in Dallas, Texas or other designated locations. This training will cover basic aspects of establishing and operating a Rusty Taco Restaurant, including all phases of restaurant opening and closing procedures, food preparation and presentation, pricing, customer service issues, employee training, cleaning procedures, cash management, budgeting and bill paying, forms, purchasing, inventory control and disposition, comprehensive marketing, selling skills, job functions, and maintenance of quality standards. The initial training program is anticipated to last approximately 28 to 35 days.

Our initial training program is subject to change, without notice, to reflect updates in the materials, methods, and manuals and changes in personnel. The subjects taught and the time periods allocated for each subject may vary based on the experience of the trainees. If you notify us within 6 months of completing the initial training program that you believe you did not receive adequate training, we will permit you and one additional person to attend our next regularly scheduled training program without charge, but you will be responsible for all travel and living expenses and related costs that your personnel incur. If we do not receive notification within this 6-month period, then you will be deemed to have waived any claim that you did not receive adequate training.

For your or your affiliates' first Rusty Taco Restaurant, we will provide on-site opening assistance (the number of persons, content and duration of which will be determined by us) at no charge to you, but you must pay all of your employee wages and the cost of food used during the assistance period. Your Operating Principal or your Restaurant Manager must be present in your Restaurant during the opening assistance period. If you or your affiliates ask us to provide opening assistance for your or your affiliates' second or subsequent Rusty Taco Restaurant, then you must pay us our then-current fee (\$400 per trainer per day), plus reimburse us for our related expenses, including costs of travel, lodging, and meals. These amounts are not refundable.

We will provide the on-site, opening assistance at your Restaurant once you have completed all improvements at your Restaurant, received a certificate of occupancy, applied for your liquor license, and have hired enough employees to staff your Restaurant.

If you do not complete the initial training to our satisfaction, in our sole discretion, we may terminate the Franchise Agreement. The following is an outline of the training you will receive:

TRAINING PROGRAM

Subject	Classroom Training Hours	On-the-Job Training Hours	Location
Introduction/Orientation	0	4	Dallas, Texas
Restaurant Operations	8-14	132-160	Dallas, Texas
Marketing/Customer Services	8-14	8	Dallas, Texas
TOTAL HOURS	16-28	144-172	

This training program is provided as an example of our recent training courses. We reserve the right to make adjustments to the training agenda above, without notice to you, including the method and manner in which the content is delivered.

We may also periodically offer additional mandatory, optional, refresher, advanced, or other training programs or seminars, addressing common problems that franchisees experience or new products, services, or techniques, and we may establish reasonable fees for attendance at these events. These seminars and training programs may discuss sales techniques, personnel training, bookkeeping, accounting, inventory acquisition and control, performance standards, advertising programs and merchandising procedures. You must pay for all travel, lodging, living expenses and compensation for you and your manager and/or employees incurred while attending any training program. Our ongoing training program may involve access to a learning management system for which you will pay an annual fee per Restaurant. This fee is not currently charged.

Operations Manual

After you sign the Franchise Agreement, we will provide you with online access to our Operations Manual via our Intranet or other means, such as via email. As of the issuance date of this Disclosure Document, our Operations Manual contains a total of 210 pages. A copy of the current table of contents of the Operations Manual is attached as Exhibit F.

Site Selection and Opening

When you sign the Franchise Agreement, we will agree on a "Site Selection Area" within which you may locate your Restaurant. You must identify a suitable site for your Restaurant, obtain our approval of the proposed site for your Restaurant, and acquire possession of the site (upon our approval) within 180 days after we and you enter into the Franchise Agreement. For each proposed site that you identify, you must deliver to us a franchise site application request packet in a form that we prescribe, including information about the site as we may reasonably require to perform our evaluation. We typically approve or refuse to approve your proposed site within 45 days of receiving all requested information about the site. Our failure to provide notification within this time period will not be considered either approval or disapproval. The criteria that we use to evaluate the site include general location and neighborhood, traffic patterns, parking, size, physical characteristics of existing buildings and lease terms. If you fail to obtain our approval of any proposed site for your Restaurant, and as a result you do not open your Restaurant by the 300th day following the date on which we execute the Franchise Agreement, we may terminate the franchise and forfeit the initial franchise fee. If you sign a Development Agreement with us, the Franchise Agreement you sign for each Rusty Taco Restaurant will contain the site selection and approval procedures.

The average time between signing a Franchise Agreement (or paying money to us) and opening of the Rusty Taco Restaurant is 6 months to a year. Factors which may affect the length of time between signing of the Franchise Agreement and opening for business include the time necessary to locate a location which we will approve; to obtain any financing you need; to obtain required permits and governmental agency approvals; to fulfill local ordinance requirements; to complete construction, remodeling, alteration, and improvement of the Franchised Location, including the installation of fixtures, equipment, and signs; to complete our initial training program; and to complete the hiring and training of personnel. Inclement weather, material or labor shortages, labor actions, slow deliveries, equipment shortages and similar factors may cause delays in construction.

ITEM 12 TERRITORY

Franchise Agreement

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control. However, if you are in full compliance with the Franchise Agreement, then during the Franchise Agreement's term, except as stated below, neither we nor our affiliates will operate, or authorize any other party to operate, a Rusty Taco Restaurant within the Protected Territory specified in the Franchise Agreement.

Generally, the "Protected Territory" will be a geographic area which is a circle having your Restaurant's front entrance as its center and a radius of two (2) miles; however, we reserve the right to change the definition of the Protected Territory in densely populated urban areas (New York City or Washington D.C., for example), and in such cases the Protected Territory may be as small as an office or retail building. Under the Franchise Agreement, you do not have any options, right of first refusal or similar rights to obtain additional Restaurants. You are not restricted from accepting orders from consumers outside of your Protected Territory; however, you may engage in catering and delivery services only with our approval and according to our Standards. That said, we may periodically add, eliminate, or modify authorized goods and services, including limiting the geographical area in which you can provide catering services.

We and our affiliates, ourselves or through authorized third parties, may (and you are not permitted to) without any obligation to compensate do the following: (a) develop, establish, and operate, and to grant to others the right to operate other businesses using the Marks, the Standards, or other names or marks; (b) advertise and promote the Rusty Taco franchise system in the Protected Territory and solicit customer in the Protected Territory; (c) acquire, be acquired by, merge, affiliate with, or engage in any transaction with other businesses (whether competitive or not) located anywhere and, even if such businesses are located in the Protected Territory, convert such other businesses to Rusty Taco Restaurants; (d) operate, and license others to operate, Rusty Taco Restaurants outside the Protected Territory Area and at non-traditional locations inside the Protected Territory; and (e) engage, directly or indirectly, at wholesale, retail or otherwise, in the production, distribution, license, and sale of any and all food, beverages, or other services and products, including those similar to those you offer, under the Marks, or under other names or marks, within and outside the Protected Territory, through any alternate method of distribution, including mail order catalogs, the Internet, telemarketing, direct marketing, e-commerce, product lines in other businesses, and other permanent, temporary, or seasonal food service facilities, carts, food trucks, or kiosks (which facilities may provide, in whole or in part, the products and services offered by a Rusty Taco Restaurant). We and our franchisees may solicit and accept orders from customers inside your Protected Territory. Currently, we and our affiliates do not operate or plan to operate or franchise businesses under different marks that sell goods and services that are the same or similar to those offered by your Restaurant.

You will operate your Restaurant from a location that you select and that we approve. You may not relocate your Restaurant without our prior written consent. As long as you are not in default, we will allow you to relocate your Restaurant to a location that meets our then-current criteria for Rusty Taco Restaurants. In

connection with the relocation, you must sign a new, then-current form of Franchise Agreement, and a general release by you and your related parties of all claims against us and our related parties. Site selection, construction, and opening of the relocated Rusty Taco Restaurant will be governed by the new franchise agreement. You are solely responsible for all relocation costs and expenses, which may include reimbursement of our reasonable costs and expenses associated with the relocation.

Unless you sign a Development Agreement, you have no options or similar rights to acquire additional franchises. Continuation of your territorial rights does not depend on your achieving a certain sales volume, market penetration or other contingency.

Development Agreement

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control. However, if you and your affiliates are in full compliance with the Development Agreement and all other agreements with us and our affiliates, and except as written below, neither we nor any of our affiliates will establish, or authorize any person or entity other than you or any of your affiliates to establish, a Rusty Taco Restaurant in the Development Agreement.

Under the Development Agreement, you agree to enter into a specific number of Franchise Agreements and open Rusty Taco Restaurants within the Development Area. We will approve the site for all Rusty Taco Restaurants that will be developed pursuant to a Development Agreement using our then-current site criteria. If you fail to meet any of your obligations under the Development Agreement, including the development obligations timeline, we may reduce the number of Restaurants you are allowed to establish under the Development Schedule, reduce the size of the Development Area where you plan to locate your Restaurant, or terminate the Development Agreement. Otherwise, continuation of your territorial rights does not depend on your achieving a certain sales volume, market penetration or other contingency, and we may not alter your Development Area or your territorial rights in the Development Area during the term of the Development Agreement. When the Development Agreement expires or is terminated, you cannot develop additional Restaurants in the Development Area, unless you and we mutually agree to enter into a new Development Agreement. Upon expiration or termination of the Development Agreement, the only territorial protection you will retain for each Restaurant developed under the Development Agreement will be the Protected Territory defined in the Franchise Agreement for that Restaurant.

We and our affiliates, ourselves or through authorized third parties, may (and you are not permitted to) without any obligation to compensate do the following: (a) develop, establish, and operate, and to grant to others the right to operate other businesses using the Marks, the Standards, or other names or marks; (b) advertise and promote the Rusty Taco franchise system inside and outside the Development Area and solicit customer in the Development Area; (c) acquire, be acquired by, merge, affiliate with, or engage in any transaction with other businesses (whether competitive or not) located anywhere and, even if such businesses are located in the Development Area, convert such other businesses to Rusty Taco Restaurants; (d) operate, and license others to operate, Rusty Taco Restaurants outside the Development Area and at nontraditional locations inside the Development Area; and (e) engage, directly or indirectly, at wholesale, retail or otherwise, in the production, distribution, license, and sale of any and all food, beverages, or other services and products, including those similar to those you offer, under the Marks, or under other names or marks, within and outside the Development Area, through any method of distribution, including mail order catalogs, the Internet, and other permanent, temporary, or seasonal food service facilities, carts, food trucks, or kiosks (which facilities may provide, in whole or in part, the products and services offered by a Rusty Taco Restaurant) regardless of the proximity to, or the competitive impact on, your operations in the Development Area..

Except for development rights granted under a Development Agreement, we do not grant any rights of first refusal to obtain additional Rusty Taco Restaurants. If you wish to obtain an additional location, you must

enter into a new Franchise Agreement for that location. The only provisions available to you to relocate a Rusty Taco Restaurant are contained in your Restaurant's Franchise Agreement, which requires you to obtain our prior written consent before relocating your Restaurant.

There is no mechanism for resolving any conflicts that may arise between franchised or company-owned Restaurants that operate under the Marks (including your Restaurant) or any restaurant that operates under the MOOYAH® or the Rusty Taco® trademarks. While we do not anticipate conflicts arising between franchisees of different brands owned by us and our affiliates, we and our affiliates will analyze any future conflict and take action (if any) that we deem appropriate. Any resolution of conflicts regarding location, customers, support, or services will be entirely within your and our business judgment.

ITEM 13 TRADEMARKS

The Franchise Agreement grants you the right to use the Marks. Currently, Rusty Taco Parent owns Marks. Under a License Agreement dated December 19, 2022, (the "License Agreement"), Rusty Taco Parent has licensed to us the right to use, and further sublicense the right to use, the Marks in connection with the franchising, development, and operation of Rusty Taco Restaurants. The License Agreement has a term of 99 years; however, the License Agreement will terminate (i) if we cease to be an affiliate of Rusty Taco Parent, or (ii) if we do not cure any material breach of our obligation within 30 days from our receipt of a notice of default. The termination of the License Agreement will result in the loss of our right to use and to sublicense the use of the Marks; however, your rights to use the Marks during the current term of your Franchise Agreement will not be affected by the termination or expiration of our license.

The following Marks are registered on the Principal Register of the United States Patent and Trademark Office ("USPTO"):

Mark	Registration Number	Registration Date
Rusty Taco (standard character)	3886304	December 7, 2010
Rusty	3889809	December 14, 2010

For each registration noted above, all required renewals and affidavits have been (or will be) filed. There is no presently effective determination of the USPTO, the Trademark Trial and Appeal Board, the trademark administrator of any state or any court, nor any pending infringement, opposition, or cancellation proceeding, nor any pending material litigation involving the Marks which is relevant to their ownership, use, or licensing.

Except the limitations described in the License Agreement, we know of no superior prior rights or infringing use that could materially affect your use of the Marks, and we know of no agreements currently in effect which significantly limit our rights to use or license the use of the Marks.

You must promptly notify us of any suspected unauthorized use of, or any challenge to the validity of the Marks or any challenge to our or our affiliates' ownership of, our license to use and to license others to use, or your right to use the Marks. We may, but are not required to, take any action we deem appropriate in connection with any infringement of the Marks. We have the right to control exclusively, or to delegate

control of, any legal, regulatory, or administrative proceeding, including the settlement thereof, regarding the Marks. You shall execute any and all documents and do such acts as may, in our opinion, be necessary to carry out such defense or prosecution, including, but not limited to, becoming a nominal party to any legal action. Except to the extent that such litigation is the result of your use of the Marks in a manner inconsistent with the terms of the Franchise Agreement, we will reimburse you for your associated costs.

We may, in our sole discretion, designate one or more new, modified, or replacement Marks for your use and require your use of any such new, modified, or replacement Marks in addition to or in lieu of any previously designated Marks. You must comply with any such directive relating to the use of new, modified, or replacement Marks within the time periods we specify depending on the reason for the change. We need not reimburse you for your expenses in complying with our direction in this regard or for any loss of revenue due to any modified or discontinued Mark. We may exercise these rights at any time and for any reason, and you waive any claims, demands or damages arising therefrom.

ITEM 14 PATENTS, COPYRIGHTS, AND PROPRIETARY INFORMATION

There are no patents or copyrights currently registered which are material to the development and operation of Rusty Taco Restaurants. We do claim copyright ownership and protection for the Franchise Agreement, Operations Manual, website and for various training, sales, promotional and other written materials relating to the operation of Rusty Taco Restaurants and the System ("Copyrighted Works"). We or our affiliates own all works of authorship relating to the franchise system and created in the future. We license you to use the Copyrighted Works to operate your Restaurant.

There is no presently effective determination of the USPTO, the United States Copyright Office (Library of Congress). Board of Patent Appeals and Interferences, or any court, or any pending infringement, opposition or cancellation proceeding or any pending material litigation involving any patents or copyrights. There are currently no agreements in effect that significantly limit our rights to use or license the use of any patents or copyrights in any manner material to the franchise. There are no infringing uses known to us which could materially affect your use of the patents or copyrights.

You must promptly notify us of any suspected unauthorized use of, or any challenge to the validity of the Copyrighted Works or any challenge to our or our affiliates' ownership of, our license to use and to license others to use, or your right to use the Copyrighted Works. We may, but are not required to, take any action we deem appropriate in connection with any infringement of the Copyrighted Works. We have the right to control exclusively, or to delegate control of, any legal, regulatory, or administrative proceeding, including the settlement thereof, regarding the Copyrighted Works. You shall execute any and all documents and do such acts as may, in our opinion, be necessary to carry out such defense or prosecution, including, but not limited to, becoming a nominal party to any legal action. Except to the extent that such litigation is the result of your use of the Copyrighted Works in a manner inconsistent with the terms of the Franchise Agreement, we will reimburse you for your associated costs.

You must modify or discontinue using the Copyrighted Works as we specify. We need not reimburse you for your expenses of complying with our direction in this regard or for any loss of revenue due to any modified or discontinued Copyrighted Works. We may exercise these rights at any time and for any reason, and you waive any claims, demands or damages arising therefrom.

You and your employees must maintain the confidentiality of all information contained in the Operations Manual and other information that we consider confidential, proprietary, or trade secret information. "Confidential Information" means all non-public information about the System and the operation of Rusty Taco Restaurants (including your Restaurant), some of which may constitute trade secrets under Applicable Law, regardless of whether it is marked "confidential," including the Standards (and other methods, formats, specifications, standards, systems, procedures, techniques, market research, customer data,

knowledge, and experience used in developing, promoting and operating Rusty Taco Restaurants and the products they offer and sell), and other elements of the System; all customer information; all information contained in the Operations Manual; site selection criteria; growth and development plans, strategies and forecasts related to the System; training and operations materials; knowledge of and specifications for, and vendors of, Items and other products and supplies; any software or other technology which is proprietary to us, our Affiliates, or the System, including, without limitation, digital passwords and identifications and any source code of, and data, reports, and other materials generated by, the software or other technology; knowledge of the operating results and financial performance of Rusty Taco Restaurants (including your Restaurant).

ITEM 15 OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISED BUSINESS

If you are not a natural person, you must designate a natural person to act as your "Operating Principal." Your Operating Principal must be a natural person, who is approved by us and must own at least a 10% direct or indirect ownership interest in you. You or your Operating Principal (if you are not a natural person) must engage in and have control over the on-premises day-to-day supervision of your Restaurant. Unless you appoint a Restaurant Manager as described below, the Operating Principal shall also devote full-time efforts to the operation of your Restaurant and shall not engage in any other business or activity, directly or indirectly, that requires substantial management responsibility or time commitment.

If, at any time, you or your affiliates operate more than one Rusty Taco Restaurant, then, in addition to the Operating Principal, you must appoint an individual (who need not have an ownership interest in you) to serve as "Restaurant Manager." The Restaurant Manager must be approved by us and, once approved by us to assume the role, shall have full control over day-to-day Restaurant management and operations. Your Restaurant Manager shall devote full-time efforts to the management and operation of your Restaurant and shall not engage in any other business or activity, directly or indirectly, that requires substantial management responsibility or time commitment.

If the Operating Principal or Restaurant Manager cease to serve in, or no longer qualify for, such position(s), you must designate another qualified person to serve as your Operating Principal or Restaurant Manager (as applicable) within 30 days after the date the prior Operating Principal or Restaurant Manager ceases to serve or no longer qualifies to serve. Any proposed replacement Operating Principal or Restaurant Manager is subject to our written approval and must successfully complete, to our satisfaction, all training we require before becoming the Operating Principal or Restaurant Manager (as applicable) and, in no event, later than 90 days after the previous Operating Principal or Restaurant Manager (as applicable) ceased to serve in such position(s). We may charge you our then-current per diem (not to exceed \$400 per trainer per day) for every replacement Restaurant Manager we train.

If you are not a natural person, then each individual or entity holding direct or indirect beneficial ownership in you must sign a Guaranty and Personal Undertaking substantially in the form attached as Attachment C to the Franchise Agreement. The spouse of every natural person that signs the Guaranty and Personal Undertaking must also sign the Guaranty and Personal Undertaking to bind their marital assets to the guaranty. Any person we designate, including individuals who attend our initial training program, must sign our then-current form of confidentiality agreement.

ITEM 16 RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

You must offer and sell all menu items that we require and only those menu items that we have approved as of the time you offer and sell them. You must prepare, package, and serve all menu items in accordance with our Standards, which may include use of specific ingredients, adherence to recipes, use of specific

containers and paper goods, packaging procedures, requirements or prohibitions relating to "combo meals," product holding times, and other standards for displaying for sale menu items and other merchandise. You must participate in all product and market research programs we require (which may include test-marketing new products) and provide us with timely reports and test results in accordance with our guidelines for each program.

We require you to participate in any customer loyalty and reward programs, contests, sweepstakes, meal deals, promotional programs, and similar campaigns we require and specify in writing. We may set certain minimum standards for uniform requirements and opening hours of your Restaurant, and we may also mandate or provide guidance on maximum and minimum prices you may charge and advertise in connection with the operation of your Restaurant. You must not offer for sale or sell any other category of products or use the premises for any purpose other than operating your Restaurant in compliance with the Franchise Agreement. You must discontinue offering and selling products that we do not or no longer approve for sale at or from your Restaurant. There are no limits on our right to make modifications to the approved products and services periodically as described in the Operations Manual, or otherwise in writing. Any failure to do so or to meet product quality standards may result in the termination of your Franchise Agreement.

ITEM 17 RENEWAL, TERMINATION, TRANSFER, AND DISPUTE RESOLUTION

These tables list certain important provisions of the Franchise Agreement, the Development Agreement, and related agreements. You should read these provisions in the Franchise Agreement and the Development Agreement attached to this Disclosure Document.

FRANCHISE AGREEMENT

Provisions	Section in Franchise Agreement	Summary
a. Length of the franchise term	Section 2.1	Unless sooner terminated as provided in this Agreement, the period beginning on the Effective Date and ending on the 10 th anniversary of the opening date of your Restaurant.
b. Renewal or extension of the term	Sections 2.1 and 2.2	If you are in good standing, you can renew the franchise for one additional 10-year term.
c. Requirements for franchisee to renew or extend	Section 2.2	You must meet the following conditions in order to renew the franchise rights for one additional 10-year term: (a) you have notified us of your desire to renew the Franchise no less than 12 months and no more than 24 months before the end of the Term; (b) you are not in default of any material provision of the Franchise Agreement and you have complied with the material terms and conditions of the Franchise Agreement throughout the Term; (c) all amounts owed to us and our affiliates and third-party suppliers to your Restaurant have been paid; (d) the Restaurant has been renovated and refurbished so that it reflects all then-current Standards; (e) you have the right to remain in possession of the site, or have secured an alternate site with our prior written approval; (f) you comply with our then-current qualifications and training requirements; (g) you sign our then-current form of Franchise

Provisions	Section in Franchise Agreement	Summary
		Agreement, the terms and conditions of which may be materially different than the terms and conditions of the Franchise Agreement, including but not limited to royalty and advertising fees and contribution requirements, and each Owner executes a personal guaranty and undertaking in the form we prescribe; (h) you and each Owner sign a general and full release in favor of us, and our affiliates and our and their respective, officers, directors, owners, managers, employees, and agents, of all claims you and your owners may then have; and (i) you pay a renewal fee, in an amount equal to 50% of the initial franchise fee we are then charging for the purchase of a Franchise. Additional renewals will be at our sole discretion.
d. Termination by franchisee	Not applicable	Not applicable (subject to state law)
e. Termination by franchisor without cause	Section 13.5	Under cross-default provision, we can terminate the Franchise Agreement if you or your affiliates fails to comply with any other agreement with us or our affiliates and do not cure such failure within the applicable cure period.
f. Termination by franchisor with cause	Section 13	We may terminate only if you or your owners (i) commit one of several violations described in Section 13 of the Franchise Agreement and fail to cure your default within the applicable cure period (if any), or (ii) fail to cure defaults under any other agreement executed between you (or your affiliates) and us (or our affiliates), including any Development Agreement.
g. "Cause" defined – curable defaults	Sections 13.3 to 13.6	You have 10 days to cure (a) failure to obtain or maintain required insurance; (b) failure to pay any amounts due to us or our affiliates when due; (c) failure to pay any amounts due to the landlord of your Restaurant's premises, your suppliers, or other trade creditors; (d) failure to pay any amounts for which we have advanced funds for or on your behalf, or upon which we are acting as guarantor of your obligations; (e) violation of any provision of the Franchise Agreement concerning the use and protection of the Marks or Copyrighted Works; or (f) violation of any provision of the Franchise Agreement concerning the preparation, service, appearance, or quality of products or services. We may also terminate the Franchise Agreement if you (or your affiliates) fail to cure any breach or default under any other agreement between you (or your affiliates) and us (or
		our affiliates), including a Development Agreement. You have 30 days to cure any other defaults not listed in Section 13.
h. "Cause" defined – non- curable defaults	Sections 13.1,13.2, and 13.5	(a) your Operating Principal or Restaurant Manager fails to successfully complete training; (b) you fail to open the Restaurant for business by the 300 th day following our

Provisions	Section in Franchise Agreement	Summary
		execution of the Franchise Agreement; (c) you (i) permanently close your Restaurant for business or inform us of your intention to cease operations, (ii) fail to actively operate your Restaurant for three or more consecutive days, or (iii) otherwise abandon or appear to have abandoned your rights under the Franchise Agreement; (d) you lose any license required to operate the Restaurant or your right to occupy the site; (e) you or any owner or Restaurant Manager are convicted of, or plead no contest to, a felony, a crime involving moral turpitude, or any other crime or offense that we believe is reasonably likely to have an adverse effect on your Restaurant or the System; (f) any Transfer or attempted Transfer in violation of the Franchise Agreement; (g) you or any owner fails to comply with the confidentiality or non-competition covenants; (h) you or any owner has made any material misrepresentations; (i) you fail to comply with notification requirements set forth concerning investigations and Crisis Management Events (as defined in the Franchise Agreement); (j) you understate any payment to us by 3% or more, or understate any such payment in any amount, twice in any two-year period; (k) an imminent threat or danger to public health or safety results from the operation of your Restaurant; (l) you knowingly maintain false books or records or submit any false reports or statements to us; (m) you offer unauthorized products or services from your Restaurant premises or in conjunction with the Marks or Copyrighted Works; (n) you purchase items for which we have identified a designated supplier from an unapproved source; (o) you fail to pass two or more quality assurance inspections within any rolling 12-month period; (r) you violate our policies for Rusty Taco Restaurant operations, without limitation failure to accurately account for all Gross Sales through the Computer System; (q) you fail to participate in any advertising program or fail to maintain required hours of operation on two or more occasions without pri

Provisions	Section in Franchise Agreement	Summary
i. Franchisee's obligations on termination or nonrenewal	Article 14	You must cease use of our Marks, de-identify, assign the lease for your Restaurant, pay all amounts due to us, and return the Operations Manual and other Confidential Information to us. We may, at our option, assume all telephone numbers for your Restaurant. You must, at our option, cancel or assign to us your rights to any Internet websites or web pages or email addresses which contain our Marks. You may also be responsible for paying lost revenue damages.
j. Assignment of contract by franchisor	Section 12.1	No restriction on our right to assign.
k. "Transfer" by franchisee – defined	Article 12	"Transfer" means the appointment of a third-party manager to assume day-to-day responsibility for the operation of your Restaurant or the sale, assignment, transfer, conveyance, give-away, pledge, mortgage, or other disposition or encumber of any direct or indirect interest in the Franchise Agreement (including, without limitation, any or all of your rights or obligations under it), your Restaurant or its assets (other than in the ordinary course of business), your right to possession of the premises on which your Restaurant is located, or any direct or indirect ownership interest in you (regardless of its size).
Franchisor approval of transfer by franchisee	Sections 12.2 and 12.3	You and, if applicable, your Owners, may not engage in any form of Transfer without our prior written consent.
m. Conditions for franchisor approval of transfer	Section 12.2	You must provide us written notice at least 60 days prior to the proposed transfer; we must determine that the Transfer will not materially and adversely affect your Restaurant, the Rusty Taco® brand or the System; the transferee meets our thencurrent standards; all monetary obligations are paid to us, our affiliates, and third parties; you are in full compliance with your Franchise Agreement and all other agreements with us and our affiliates; you or the transferee agree to refurbish your Restaurant premises in accordance with our Standards; you and each new owner signs a guarantee (subject to state law); you pay the transfer fee; you provide us all transfer documents and other information we request; the transferee signs our then-current form of franchise agreement; and the transferee completes our then-current training requirements.
n. Franchisor's right of first refusal to acquire franchisee's business	Section 12.6	We can match any bona fide offer for your Restaurant.
o. Franchisor's option to purchase your business	Section 14.5	We have the option to purchase some or all of your equipment, furnishings and fixtures on expiration or termination of your Franchise Agreement, at their then-current fair market value.

Provisions	Section in Franchise Agreement	Summary
p. Death or disability of franchisee	Section 12.7	Same requirements as for transfer in "m" above, except there is no transfer fee and we do not have right of first refusal. If your interest is not transferred within 6 months following your (or an owner or Operating Principal's) death or legal incapacity, your Franchise Agreement may be terminated.
q. Non-competition covenants during the term of the franchise	Section 15.1	You and your owners will not, either directly or indirectly, for yourselves, or through, on behalf of, or in conjunction with any person, or legal entity: (i) divert or attempt to divert any present or prospective customer of a Rusty Taco Restaurant to any competitor, or do or perform any other act injurious or prejudicial to the goodwill associated with the Marks and the System; or (ii) own, maintain, advise, operate, engage in, be employed by, make loans to, or have any interest in or relationship with a Competitive Business (A) anywhere in the world, or (B) at any location within the United States, its territories or commonwealths, or any other country, province, state, or geographic area in which we or our affiliates have used, sought registration of, or registered the Marks or similar marks.
		"Competitive Business" means any restaurant or other food service business deriving more than 10% of its revenue (excluding revenue derived from the sale of alcoholic beverages) from the sale of Mexican-inspired food, other than a Rusty Taco Restaurant operated pursuant to a then-currently effective Franchise Agreement with us or our Affiliate.
r. Non-competition covenants after the franchise is terminated or expires	Section 15.2	For an uninterrupted period of two (2) years after the termination or expiration of the Franchise Agreement, neither you nor your owners may own, maintain, advise, operate, engage in, be employed by, make loans to, or have any interest in a Competing Business that: (A) is, or is intended to be, located at the site; (B) is within a five-mile radius of your Restaurant; or (C) is within a five-mile radius of any other Rusty Taco Restaurant operating under the System and Marks, that is in existence or under development at any location within the United States, its territories, or commonwealths, or any other country, province, state, or geographic area in which we or our affiliates have used, sought registration of, or registered the Marks or similar marks, or operate or license others to operate a business under the Marks or similar marks, at the time of such expiration, termination, or Transfer.
s. Modification of the agreement	Section 18.1	Any change to the Franchise Agreement or any other agreement between us and you must be in writing and signed by all parties; however, we may freely change the system standards.
t. Integration/merger clause	Section 18.1	Only the terms of the Franchise Agreement, Standards described in our Operations Manual, and other related written agreements are binding (subject to state law).

Provisions	Section in Franchise Agreement	Summary			
		Notwithstanding the foregoing, nothing in this or any related agreement is intended to disclaim the express representations made in the Franchise Disclosure Document, its exhibits and amendments.			
u. Dispute resolution by arbitration or mediation	Section 19.2	We and you must arbitrate all disputes at a location in or within 50 miles of our or, as applicable, our successor's or assign's then current principal place of business (currently, Plano, Texas) (subject to state law).			
v. Choice of forum	Section 19.2	The venue of arbitration will be in the state where our or, as applicable, our successor's or assign's corporate headquarters are located (currently, Plano, Texas) (subject to applicable state law).			
w. Choice of law	Section 19.1	Delaware law (subject to applicable state law).			

DEVELOPMENT AGREEMENT

Provision	Section in Development Agreement	Summary
a. Length of the Agreement term	Data Sheet	The period beginning on the effective date and ending on the earlier of: (i) the date on which you have completed your development obligations, or (ii) 12:00 midnight EST on the last day specified in the development schedule described in the Development Agreement.
b. Renewal or extension of the term	Not applicable	You do not have the right to renew your Development Agreement.
c. Requirements for Developer to renew or extend		Not applicable
d. Termination by Developer	Not applicable	Not applicable (subject to state law)
e. Termination by the franchisor without cause	Not applicable	Not applicable
f. Termination by the franchisor with "cause"	Section 7	We may terminate only if (i) you or your owners commit one of several violations described in Section 7 of the Franchise Agreement and fail to cure your default within the applicable cure period (if any), or (ii) fail to cure defaults under any other agreement executed between you (or your affiliates) and us (or our affiliates), including any Franchise Agreement executed pursuant to the Development Agreement.

Provision	Section in Development Agreement	Summary
g. "Cause" defined - curable defaults	Sections 7.3, 7.4, and 7.5.	You have 10 days to cure: (a) failure to pay any amounts due to us; (b) failure to pay any amounts due to your trade creditors; or (c) failure to pay any amounts for which we have advanced funds for or on your behalf, or upon which we are acting as guarantor of your obligations. We also terminate the Development Agreement, if you (or your affiliates) fail to cure breach or default in the performance under any other agreement between you (or your affiliates) and us (or our affiliates), including any Franchise Agreement executed pursuant to the Development Agreement. You have 30 days to cure any other defaults not listed in Section 7.
h. "Cause" defined – non-curable defaults	Sections 7.1 and 7.2	(a) you fail to adhere to the Development Schedule; (b) you or any owner is convicted of, or pleads no contest to, a felony, a crime involving moral turpitude, or any other crime or offense that we believe is reasonably likely to have an adverse effect on the System; (c) there is any unauthorized transfer or attempted unauthorized transfer; (d) you or any owner fails to comply with the confidentiality or noncompetition covenants; (e) you or any owner has made any material misrepresentations; (f) we deliver to you three or more written notices of default pursuant to within any rolling 12-month period, whether or not the defaults described in such notices ultimately are cured; (g) if you are an individual, in the event of your death or permanent incapacity; (g) you terminate, without cause, any Franchise Agreement signed pursuant to this Agreement or we terminate any such Franchise Agreement based on your breach; (h) insolvency or the occurrence of a bankruptcy event; (i) your insolvency or bankruptcy; or (j) you (or your affiliates) fail to cure breach or default in the performance under any other agreement between you (or your affiliates) and us (or our affiliates), including any Franchise Agreement executed pursuant to the Development Agreement.
i. Developer's obligation on termination/non- renewal	Section 7.2	Upon termination or expiration of this Agreement, you shall have no further right to acquire any additional Franchises or to develop any Rusty Taco Restaurants for which a Franchise Agreement has not been signed. Termination or expiration of this Agreement will not affect any rights or obligations under any then-existing Franchise Agreement.
j. Assignment of contract by franchisor	Section 6.1	No restrictions on our right to assign.
k. "Transfer" by	Sections 8.2 and	"Transfer" includes the sale, assignment, transfer, conveyance,
Developer – defined	8.3	give-away, pledge, mortgage, or other disposition or encumber

Provision	Section in Development Agreement	Summary			
		of any direct or indirect interest in the Development Agreement, changes in ownership of the developer entity, changes in transfers of assets, and private and public offerings; and assignment of Franchise Agreement.			
Franchisor approval of transfer by Developer	Section 6.2	You may not engage in a transfer in respect of this Agreement. Your Owners may not engage in a Transfer of their interests in you without our prior written consent, which we may withhold or condition in our sole discretion.			
m. Conditions for franchisor approval of transfer	Not applicable	(i) you must provide all reasonable information and the transferee and the transfer deal terms (including, any financing) must meet our then-current criteria, (ii) your compliance with the Development Agreement and all payment obligations, (iii) execution of new Development Agreement by transferee and its owners and general lease by you, (iv) payment of transfer fee, (iv) compliance with post-termination non-compete agreement, and (v) transfer of all franchise agreement executed pursuant to the Development Agreement.			
n. Franchisor's right of first refusal to acquire Developer's business	Not applicable	Not applicable			
o. Franchisor's option to purchase Developer's business	Not applicable	Not applicable			
p. Death or disability of Developer	Sections 6 and 7.2	If you are an individual, we may terminate your Development Agreement following your death or permanent incapacity. If you are an entity, then any transfer of ownership interest in you that is held by the estate of your deceased owner must be transferred in accordance with the transfer provision of the Development Agreement.			
q. Non-competition covenants during the term of the Agreement	Section 4.4	Neither you nor any of your owners, your or your owners' Affiliates, or the officers, directors, managers or immediate family members of any of the foregoing will: (i) have any direct or indirect interest as an owner – whether of record, beneficially, or otherwise – in a Competitive Business (defined below), wherever located or operating (except that equity ownership of less than five percent (5%) of a Competitive Business whose stock or other forms of ownership interest are publicly traded on a recognized United States stock exchange will not be deemed to violate this subparagraph); (ii) perform services as a director, officer, manager, employee, consultant, representative, lessor, or agent for a Competitive Business, wherever located or operating; (iii) divert or attempt to divert any actual or potential business or customer of any Rusty Taco Restaurant to a Competitive Business; or (iv) directly or indirectly, appropriate, use or duplicate the Rusty Taco franchise system or Standards, or any portion thereof, for use in any other business or endeavor.			

Provision	Section in Development Agreement	Summary
		The term "Competitive Business" means any restaurant, food service or other business (other than a Rusty Taco Restaurant that is operating pursuant to a Franchise Agreement with us) (i) deriving more than 10% of its revenue (excluding revenue derived from the sale of alcoholic beverages) from the sale of Mexican-inspired food, or (ii) that grants franchises or licenses for the operation of any of the foregoing or provides services to the franchisor or licensor of any of the foregoing.
r. Non-competition covenants after the Agreement is terminated or expires	Section 4.4	For 2 years after the termination or expiration of the Development Agreement, you, your owners, and your and your owners' immediate family members shall not either directly or indirectly, for yourselves, or through, on behalf of, or in conjunction with any person, persons, or legal entity, own, maintain, advise, operate, engage in, be employed by, make loans to, lease property to, invest in, provide any assistance to, or have any interest in (as an owner or otherwise) or relationship or association with any Competitive Business that: (i) is, or is intended to be, located in the Development Area; (ii) is within a five-mile from the outer boundary of the Development Area; or (iii) is within a five-mile radius of any other Rusty Taco Restaurant. The above restrictions will apply to the departing owner for a two-year period beginning on the date such person ceases to be your owner.
s. Modification of the Development Agreement	Section 10.1	Any modification must be in writing and signed by all parties.
t. Integration/merger clause	Section 10.1	Only the terms of the Development Agreement and other related written agreements are binding (subject to state law). Notwithstanding the foregoing, nothing in this or any related agreement is intended to disclaim the express representations made in the Franchise Disclosure Document, its exhibits and amendments.
u. Dispute resolution by arbitration or mediation	Section 11.2	We and you must arbitrate all disputes at a location in or within 50 miles of our or, as applicable, our successor's or assign's then current principal place of business (currently, Plano, Texas) (subject to state law).
v. Choice of forum	Sections 11.2	The venue of arbitration will be in the state where our or, as applicable, our successor's or assign's corporate headquarters are located (currently, Plano, Texas) (subject to applicable state law).
w. Choice of law	Section 11.1	Delaware law (subject to applicable state law).

ITEM 18 PUBLIC FIGURES

We do not use any public figure to promote the franchise.

ITEM 19 FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the Disclosure Document. Financial performance information that differs from that included in Item 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

This information contained in this Item 19 is based on actual historical financial results for franchised Rusty Taco Restaurants during fiscal year 2024 which is the period from January 1, 2024 through December 29, 2024. As of the end of fiscal year 2024, 36 franchised Rusty Taco Restaurants were in operation. For purposes of the disclosures made below, we excluded from the data set (i) 9 Rusty Taco Restaurants that commenced operations in 2024 (including 1 that reopened in 2024 after a prolonged closure for 3 months due to change in ownership) and, as a result, experienced new opening sales volatility and did not have a full year's worth of data, and (ii) 3 Rusty Taco Restaurants that operated in locations that were atypical of our standard Rusty Taco Restaurant model and had limited operations and a small or no seating capacity. The remaining 24 Rusty Taco Restaurants, which offer our standard menu offering and have the required seating and other required operational standards, comprise the data set on which the information provided below is based.

The chart below reflects the Average Unit Volume during fiscal year 2024 of the 24 traditional Rusty Taco Restaurants in the data set. Below the chart, we have provided the Gross Sales of one Rusty Taco Restaurant that operates in an airport terminal (which we eliminated from the data set) because we expect to grant franchises for additional airport locations.

	Number of	Average Unit	#/%	Highest AUV	Lowest AUV	Median AUV
	Restaurants	Volume ("AUV")	Exceeding Average			
Total No. of	24	\$1,067,488	12 / 50%	\$1,791.183	\$408,807	\$600,638
Restaurants in						
the Data Set						
Top Third	8	\$1,521,669	4 / 50%	\$1,791,183	\$1,238,265	\$1,523,761
Middle Third	8	\$1,035,505	4 / 50%	\$1,220,211	\$821,136	\$1,048,567
Bottom Third	8	\$645, 291	4 / 50%	\$790,865	\$408,807	\$675,689

The Gross Sales during fiscal year 2024 of the Rusty Taco Restaurant that operates in the airport terminal were \$3,789,144.

Notes:

- (1) We calculated the figures in the tables above using information that franchisees provided.
- (2) "Average Unit Volume" is the average Gross Sales of Rusty Taco Restaurants in any given data set described above during fiscal year 2024.
- (3) "Gross Sales" means the total revenue attributable to and receipts from all services and products and all income of every other kind and nature related to Rusty Taco Restaurants, whether or not in compliance with Franchise Agreement including income related to catering operations and special events and the full value of meals provided to bona fide employees as a benefit of their employment (except franchisees may deduct from Gross Sales the value of any employee discounts that are given during the

week in which the meals are provided), whether for cash or credit, barter or exchange, and regardless of collection in the case of credit. Gross Sales does not include (a) sales (or similar) taxes that are collected from customers of the restaurant and paid to the appropriate taxing authority, promotions, voids, and discounts; (b) proceeds from isolated sales of trade fixtures that are not part of the products and services offered at Rusty Taco Restaurants and that do not have any material effect on the operation of Rusty Taco Restaurants; (c) tips or gratuities that restaurant customers pay directly to the restaurant employees or owner; or (d) returns to shippers or manufacturers. Revenue from the purchase or redemption of gift certificates, gift cards, loyalty or similar programs is calculated as part of Gross Sales in accordance with our Predecessor's then-current guidelines for such programs. Gross Sales also includes all insurance proceeds that the franchisees received to replace revenue they lost from the interruption of the operation of their Rusty Taco Restaurant. This is the same definition used to calculate your Royalty Fee and other Gross Sales-based fees under the current form of Franchise Agreement.

Some outlets have sold this amount. Your individual results may differ. There is no assurance you'll earn as much.

Written substantiation of this information is available to you upon reasonable request.

Other than the preceding financial performance representation, we do not make any financial performance representations about a franchisee's future financial performance or the past financial performance of franchised outlets. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting its General Counsel, Michael Raoufpour, at 5412 W. Plano Pkwy., Suite 100, Plano, Texas 75093, or by phone at (972)-467-8095, as well as the Federal Trade Commission, and the appropriate state regulatory agencies.

ITEM 20 OUTLETS AND FRANCHISEE INFORMATION

Table No. 1 Systemwide Outlet Summary For Years 2022 to 2024

Outlet Type	Year	Outlets at the Start of the Year	Outlets at the End of the Year	Net Change
	2022	32	33	+1
Franchised	2023	33	33	0
	2024	33	36	+3
	2022	4	1	-3
Company-Owned	2023	1	1	0
	2024	1	1	0
Total Outlets	2022	36	34	-2
	2023	34	34	0
	2024	34	37	+3

1. The figures in Tables No. 1 to 4 of this Item 20 are as of the following fiscal year-end dates: December 29, 2024; December 31 2023; and January 1, 2023.

2. Prior to the closing of the acquisition in December 2022, Company-Owned outlets were owned by Predecessor's affiliate(s), and after the closing of the acquisition, our affiliate owns Company-Owned outlet.

Table No. 2 Transfers of Outlets from Franchisee to New Owners (other than to Franchisor) For Years 2022 to 2024

State	Year	Number of Transfers
	2022	0
Iowa	2023	0
	2024	1
	2022	0
Minnesota	2023	3
	2024	0
	2022	0
Ohio	2023	0
	2024	1
	2022	0
Total	2023	3
	2024	2

Table No. 3
Status of Franchised Outlets For Years 2022 to 2024

State	Year	Outlets at Start of Year	Outlets Opened	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations- Other Reasons	Outlets at End of the Year
	2022	2	1	0	0	0	0	3
Arizona	2023	3	0	0	0	0	0	3
	2024	3	0	0	0	0	0	3
	2022	3	0	0	0	0	2	1
Colorado	2023	1	0	0	0	0	1	0
	2024	0	3	0	0	0	0	3
	2022	1	0	0	0	0	1	0
Georgia	2023	0	0	0	0	0	0	0
	2024	0	2	0	0	0	0	2
	2022	0	0	0	0	0	0	0
Idaho	2023	0	1	0	0	0	0	1
	2024	1	1	0	0	0	0	2
	2022	1	0	0	0	0	1	0
Indiana	2023	0	0	0	0	0	0	0
	2024	0	0	0	0	0	0	0
	2022	2	0	0	0	0	1	1
Iowa	2023	1	0	0	0	0	0	1
	2024	1	0	0	0	0	0	1
	2022	0	1	0	0	0	0	1
Kansas	2023	1	0	0	0	0	0	1
	2024	1	0	1	0	0	0	0

State	Year	Outlets at Start of Year	Outlets Opened	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations- Other Reasons	Outlets at End of the Year
	2022	7	0	0	0	0	0	7
Minnesota	2023	7	0	0	1	0	2	4
	2024	4	0	0	0	0	0	4
	2022	3	1	0	0	0	0	4
Nebraska	2023	4	0	0	0	0	0	4
	2024	4	0	0	0	0	0	4
	2022	0	1	0	0	0	0	1
Nevada	2023	1	0	0	0	0	0	1
	2024	1	0	1	0	0	0	0
Norre	2022	2	0	0	0	0	0	2
New Mexico	2023	2	0	0	0	0	0	2
Mexico	2024	2	0	1	0	0	0	1
	2022	4	0	0	0	0	1	3
Ohio	2023	3	0	0	0	0	0	3
	2024	3	0	0	0	0	0	3
	2022	6	2	0	0	0	0	8
Texas	2023	8	2	0	0	0	0	10
	2024	10	0	1	0	0	0	9
	2022	0	0	0	0	0	0	0
Utah	2023	0	0	0	0	0	0	0
	2024	0	2	0	0	0	0	2
	2022	1	1	0	0	0	0	2
Virginia	2023	2	1	0	0	0	0	3
	2024	3	0	1	0	0	0	2
	2022	32	7	0	0	0	6	33
Total	2023	33	4	0	1	0	3	33
	2024	33	8	5	0	0	0	36

Table No. 4 Status of Company-Owned Outlets For Years 2022 to 2024

State	Year	Outlets at Start of Year	Outlets Opened	Outlets Reacquired from Franchisees	Outlets Closed	Outlets Sold to Franchisees	Outlets at End of the Year
	2022	4	0	0	3	0	1
Texas	2023	1	0	0	0	0	1
	2024	1	0	0	0	0	1
	2022	4	0	0	3	0	1
Total	2023	1	0	0	0	0	1
	2024	1	0	0	0	0	1

Table No. 5
Projected Openings as of December 29, 2024

State	Franchise Agreements Signed but Outlet Not Opened	Projected New Franchised Outlets in the Next Fiscal Year	Projected New Company-Owned Outlets in the Next Fiscal Year
Colorado	4	4	0
Illinois	1	1	0
Total	5	5	0

Exhibit E reflects (i) the name of each of our franchisees and the address and telephone numbers of their Rusty Taco Restaurants as of December 29, 2024, and (ii) the name, city, state, and current business (or if unknown, home) telephone number of the franchisees who has had a franchise terminated, canceled, not renewed, or otherwise voluntarily or involuntarily ceased to do business under its Franchise Agreement during the most recently completed fiscal year-end or who has not communicated with us within 10 weeks of this Disclosure Document's issuance date.

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

During the last three fiscal years, in some instances, current or former franchisees have signed provisions restricting their ability to speak openly about their experience with Rusty Taco franchise system. You may wish to speak with current and former franchisees, but be aware that not all of those franchisees will be able to communicate with you.

There are no trademark specific franchisee associations that must be disclosed in this Disclosure Document.

ITEM 21 FINANCIAL STATEMENTS

Attached as Exhibit A to this Disclosure Document are our (i) unaudited internally prepared balance sheet as of March 30, 2025 and income statement for the period commencing from December 30, 2024 to March 30, 2025; and (ii) audited financial statements for fiscal years 2022, 2023, and 2024 comprised of the balance sheet and the related statements of operations, members' equity, and cash flows. Our fiscal year consists of a 52- or 53-week period ending on the last Sunday closest to December 31.

ITEM 22 CONTRACTS

This Disclosure Document includes a sample of the following contracts:

Form of Area Development Agreement – Exhibit B

Form of Franchise Agreement – Exhibit C

Form of General Release State – Exhibit D

State Riders – Exhibit H

Form of Representation & Acknowledgement Statement – Exhibit I

ITEM 23 RECEIPTS

<u>Exhibit J</u> of this Disclosure Document contains two copies of a detachable acknowledgment of receipt, one of which is for you and the other is to be signed by you and given to us.

EXHIBIT A

FINANCIAL STATEMENTS

UNAUDITED FINANCIAL STATEMENTS

Rusty Taco Franchising, LLC Balance Sheet March 30, 2025

FINANCIAL ROW	AMOUNT
ASSETS	
Current Assets	
Bank	\$25,111.37
Accounts Receivable	\$76,738.60
Other Current Asset	\$1,328.98
Total Current Assets	\$103,178.95
Fixed Assets	
17000 - Fixed Assets	\$10,949.27
Total Fixed Assets	\$10,949.27
Other Assets	
18000 - Intangible & Other Assets	\$421,889.62
Total Other Assets	\$421,889.62
Total ASSETS	\$536,017.84
Liabilities & Equity	
Current Liabilities	
Accounts Payable	\$607,284.72
Credit Card	\$361.29
Other Current Liability	\$681,766.62
Total Current Liabilities	\$1,289,412.63
Equity	(\$753,394.79)
Total Liabilities & Equity	\$536,017.84

Rusty Taco Parent : Rusty Taco Franchising, LLC Income Statement December 30, 2024 - March 30, 2025

FINANCIAL ROW	AMOUNT
Ordinary Income/Expense	
Income	
40000 - Sales & Incomes	\$418,480.91
Total - Income	\$418,480.91
Gross Profit	\$418,480.91
Expense	
60000 - Labor	\$351,227.85
70000 - Controllable Costs	\$274,318.94
80000 - Non Controllable Expenses	\$138,458.03
Total - Expense	\$764,004.82
Net Ordinary Income	(\$345,523.91)
Other Income and Expenses	
Other Expense	
90000 - Other Income & Expenses	\$6,501.00
Total - Other Expense	\$6,501.00
Net Other Income	(\$6,501.00)
Net Income	(\$352,024.91)

AUDITED FINANCIAL STATEMENTS



Financial Statements

December 29, 2024, December 31, 2023 and January 1, 2023

Table of Contents

December 29, 2024, December 31, 2023 and January 1, 2023

	_ Page
Independent Auditors' Report	1
Financial Statements	
Balance Sheets	3
Statements of Operations and Changes in Member's (Deficit) Equity	4
Statements of Cash Flows	5
Notes to Financial Statements	6



Independent Auditors' Report

To the Member and Board of Directors of Rusty Taco Franchising, LLC

Opinion

We have audited the financial statements of Rusty Taco Franchising, LLC (the Company), which comprise the balance sheets as of December 29, 2024, December 31, 2023 and January 1, 2023, and the related statements of operations and changes in member's (deficit) equity and cash flows for the years then ended, and the related notes to the financial statements.

In our opinion, the accompanying financial statements present fairly, in all material respects, the financial position of the Company as of December 29, 2024, December 31, 2023 and January 1, 2023 and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America (GAAP).

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under those standards are further described in the Auditors' Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with GAAP, and for the design, implementation and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the financial statements are available to be issued.

Auditors' Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditors' report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the financial statements, whether due to
 fraud or error, and design and perform audit procedures responsive to those risks. Such procedures
 include examining, on a test basis, evidence regarding the amounts and disclosures in the financial
 statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that
 raise substantial doubt about the Company's ability to continue as a going concern for a reasonable
 period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audits, significant audit findings, and certain internal control-related matters that we identified during the audits.

Los Angeles, California

Baker Tilly US, LLP

April 7, 2025

Balance Sheets

December 29, 2024, December 31, 2023 and January 1, 2023

	December 29 [2024		De	December 31, 2023		January 1, 2023	
Assets							
Current Assets Cash Accounts receivable Prepaid expenses	\$	61,119 33,807	\$	88,100 37,121 50,207	\$	372,469 35,488	
Total current assets		94,926		175,428		407,957	
Property and Equipment, Net		11,759		12,031		4,790	
Goodwill, Net		435,499		489,937		544,375	
Total assets	\$	542,184	\$	677,396	\$	957,122	
Liabilities and Member's (Deficit) Equity							
Current Liabilities Accounts payable and accrued expenses Bank overdraft liability Deferred franchise fee revenue, current Due to related parties	\$	268,816 35,338 125,000 486,275	\$	84,350 - 281,250 263,721	\$	37,428 62,058 10,003	
Total current liabilities		915,429	-	629,321		109,489	
Deferred Franchise Fee Revenue, Long-Term Total noncurrent liabilities		318,125 318,125		425,625 425,625		482,317	
Total liabilities		1,233,554		1,054,946		591,806	
Member's (Deficit) Equity		(691,370)		(377,550)		365,316	
Total liabilities and member's (deficit) equity	\$	542,184	\$	677,396	\$	957,122	

Rusty Taco Franchising, LLC
Statements of Operations and Changes in Member's (Deficit) Equity Years Ended December 29, 2024, December 31, 2023 and January 1, 2023

	December 29, 2024		-		January 1, 2023	
Revenues						
Royalty fee income	\$	1,237,275	\$	1,335,204	\$	35,488
Advertising fee income		309,632		200,418		-
Franchise fee income		515,500		142,500		-
Other income		68,664		23,287		-
Total revenues		2,131,071		1,701,409		35,488
Operating Expenses						
Salaries, wages and benefits		1,223,665		1,169,528		54,001
General and administrative		1,824,863		1,274,747		16,171
Total operating expenses		3,048,528		2,444,275		70,172
Loss from operations		(917,457)		(742,866)		(34,684)
Other Expense						
Other expense		121,363				
Total other expense		121,363				
Net loss	\$	(1,038,820)	\$	(742,866)	\$	(34,684)
Member's (Deficit) Equity, Beginning	\$	(377,550)	\$	365,316	\$	-
Member contribution		725,000		-		400,000
Net loss		(1,038,820)		(742,866)		(34,684)
Member's (Deficit) Equity, Ending	\$	(691,370)	\$	(377,550)	\$	365,316

Statements of Cash Flows

Years Ended December 29, 2024, December 31, 2023 and January 1, 2023

	December 29, 2024				January 1, 2023	
Cash Flows From Operating Activities						
Net loss	\$	(1,038,820)	\$	(742,866)	\$	(34,684)
Adjustments to reconcile net loss to net cash						
used in operating activities:						
Depreciation expense		2,977		1,375		81
Amortization expense		54,438		54,438		-
Bad debt expense		-		26,008		-
Non-cash franchise fee income		-		(5,000)		-
Change in operating assets and liabilities:						
Accounts receivable		(23,998)		(27,641)		(35,488)
Prepaid expenses		16,400		(50,207)		-
Accounts payable and accrued expenses		184,466		46,922		37,428
Bank overdraft liability		35,338		-		-
Due to related parties		222,554		253,718		10,003
Deferred revenue		(263,750)		162,500		
Net cash used in operating activities		(810,395)		(280,753)		(22,660)
Cash Flows From Investing Activities						
Purchase of property and equipment		(2,705)		(3,616)		(4,871)
Net cash used in investing activities		(2,705)		(3,616)		(4,871)
Cash Flows From Financing Activities						
Member contributions		725,000				400,000
Net cash provided by financing activities		725,000				400,000
Net change in cash and cash equivalents		(88,100)		(284,369)		372,469
Cash, Beginning		88,100		372,469		_
Cash, Ending	\$	-	\$	88,100	\$	372,469
Noncash Investing and Financing Information Goodwill recognized related to liabilities assumed from parent company	\$	<u>-</u> _	\$	<u>-</u> _	\$	384,259
Property and equipment received due to franchise termination	\$		\$	5,000	\$	-

Notes to Financial Statements December 29, 2024, December 31, 2023 and January 1, 2023

1. Organization and Description of Business

Rusty Taco Franchising, LLC (the Company), a Delaware limited liability company (LLC), was formed on October 6, 2022. The sole member of the Company is Rusty Taco Parent, LLC, a Delaware LLC (the Member).

The Company offers and sells franchises to operate restaurants under the Rusty Taco brand throughout the U.S. and internationally through a 99-year renewable trademark license agreement with its Member. The license agreement grants the Company a non-exclusive right to use the Rusty Taco trademarks and to license the trademarks to franchisees under franchise agreements. As of December 29, 2024, December 31, 2023 and January 1, 2023, the Company has 36, 43, and 33 franchise locations in operation, respectively, of which all are located in the United States.

2. Summary of Significant Accounting Policies

Basis of Presentation

The Company's financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America (U.S. GAAP). The Company believes this information includes all adjustments, consisting of normal recurring accruals, necessary to fairly present the financial condition as of December 29, 2024, December 31, 2023 and January 1, 2023. References to Accounting Standards Codification (ASC) and Accounting Standards Update (ASU) included hereinafter refers to the Accounting Standards Codification and Accounting Standards Update established by the Financial Accounting Standards Board (FASB) as the source of authoritative

U.S. GAAP.

Fiscal Year

The Company uses a fiscal year consisting of the 52 or 53-week period ending on the last Sunday closest to December 29, 2024. The fiscal period end date of the attached financial statements is December 29, 2024, December 31, 2023 and January 1, 2023.

Use of Estimates

The preparation of the financial statements, in accordance with U.S. GAAP, requires that management makes certain estimates and assumptions. These estimates and assumptions affect the reported amount of assets and liabilities and the disclosure of contingent assets and liabilities as of the balance sheet date. The actual results could differ significantly from those estimates.

Cash and Cash Equivalents

The Company considers all highly liquid debt instruments purchased with an original maturity of 90 days or less to be cash equivalents. As of December 29, 2024, December 31, 2023 and January 1, 2023, the Company carried no cash equivalents.

The Company maintains its cash and cash equivalents at major financial institutions in deposit accounts which, at times, may exceed federally insured limits. The Company has not experienced any losses in the accounts and management believes it is not exposed to significant risk of loss related to the deposits.

Notes to Financial Statements December 29, 2024, December 31, 2023 and January 1, 2023

Accounts Receivable

Accounts receivable are stated net of an allowance for doubtful accounts and consists of royalty fees collected from the Company's franchisees. On a periodic basis, the Company evaluates its accounts receivable balance and establishes an allowance for doubtful accounts, if required, based on a history of past write-offs and collections and current credit considerations. Management of the Company has determined there is no need for a reserve as of December 29, 2024, December 31, 2023 and January 1, 2023.

Concentrations of Credit Risk

As of December 29, 2024, one vendor accounted for approximately 16% of total accounts receivable and a related party franchisee accounted for approximately 20% of accounts receivable. For the year ended December 29, 2024, one franchisee comprised approximately 13% of total revenue.

As of and for the year ended December 31, 2023, one franchisee accounted for approximately 17% of accounts receivable and three franchisees accounted for approximately 38% of total revenue. The Company has a related party franchisee which accounted for 33% of accounts receivable and 1% of total revenue. There were no material concentrations for disclosure for the period ended January 1, 2023.

A loss of one of these franchisees could have a material adverse effect on the Company's financial position and results of operations.

Property and Equipment

Property and equipment are stated at cost, less accumulated depreciation. Depreciation is computed using the straight-line method over the estimated useful lives of the assets, which range from three to seven years for equipment and the lesser of seven years or the life of the lease for leasehold improvements. The Company capitalizes assets with useful lives greater than one year and a value of more than \$1,000. Expenditures for major additions or improvements which extend the useful lives of assets are capitalized. Minor replacements, maintenance and repairs, which do not improve or extend the lives of the assets, are charged to operations as incurred. Disposals are removed at cost less accumulated depreciation, and any resulting gain or loss is reflected in current operations. Property and equipment have estimated useful lives ranging from three to five years.

Goodwill

Goodwill represents the excess of costs over fair value of assets of the business acquired. In January 2014, the FASB issued guidance allowing private companies an accounting alternative for the subsequent measurement of goodwill. An entity within the scope of this guidance may elect to amortize goodwill on a straight-line basis over ten years or less than ten years if the entity demonstrates that another useful life is more appropriate. An entity that elects the accounting alternative is further required to make an accounting policy election to test goodwill for impairment at either the entity level or the reporting unit level. Goodwill should be tested for impairment when a triggering event occurs that indicates that the fair value of an entity (or a reporting unit) may be below its carrying amount. Under this accounting alternative, the goodwill impairment amount, if any, would be the excess of the entity's (or the reporting unit's) carrying amount over its fair value. The Company elected to adopt this guidance and amortizes goodwill on a straight-line basis over ten years and tests for impairment at the entity level when a triggering event occurs that indicates that the fair value of the entity may be below its carrying amount. For the years ended December 29, 2024, December 31, 2023 and January 1, 2023, amortization expense related to goodwill was \$54,438, \$54,438 and \$0, respectively.

Notes to Financial Statements December 29, 2024, December 31, 2023 and January 1, 2023

Fair Value Measurements

The Company's financial instruments, none of which are held for trading purposes, include cash, accounts receivable, accounts payable and accrued expenses. Management estimates that the fair value of all financial instruments at December 29, 2024, December 31, 2023 and January 1, 2023 does not differ materially from the aggregate carrying values of its financial instruments recorded in the accompanying financial statements.

Revenue Recognition

The Company records revenue under ASC No. Topic 606, *Revenue from Contracts With Customers (Topic 606)*, which requires revenue to be recorded as the transfer of promised goods or services to customers in an amount that reflects the consideration to which the reporting entity expects to be entitled in exchange for those goods or services.

Franchise and Development Revenues

The franchise arrangement between the Company and each franchise owner of a Rusty Taco store is documented in the form of a franchise agreement and, in select cases, a development agreement. The franchise arrangement requires the Company as franchisor to perform various activities to support the Rusty Taco brand and does not involve the direct transfer of goods and services to the franchise owner as a customer. The Company has selected the private opening alternative under ASU No. 2021-02 for pre-opening services as a single performance obligation. Based on this guidance, the Company recognizes initial franchise fees at the opening of the store location.

The transaction price in a standard franchise arrangement consists of (i) franchise/development fees; (ii) continuing franchise fees (royalties); and (iii) advertising fees. Based upon the Company's analysis of stand-alone selling prices, the Company has allocated the entirety of the initial franchise fees to pre-opening services which are considered distinct performance obligations from the use of franchise license under ASU No. 2021-02. Deferred revenue consists mainly of franchise and development fees paid in advance of the store opening.

Royalty and Advertising Fee Revenue

Royalty and advertising fee revenues are based on a percentage of sales and are recognized when the food items are delivered to or carried out by customers. Royalty fees amount to up to 6% of franchisee gross sales per year and advertising fees amount to up to 3% of franchisee gross sales per year. Payments for domestic royalties and advertising fees are generally due and collected within seven days of the prior weekend date. Payments for international royalties are due and collected within 30 days of month end.

Revenue related to the Company's franchise royalties and advertising fees is recorded when earned based upon the franchisee's sales. Additionally, the Company records revenues and related expenses of the advertising fund on a gross basis within the statements of operations. These funds exist solely for the purpose of promoting the Rusty Taco brand.

General and Administrative Expenses

The Company expense general and administrative expenses in the period in which the service was incurred.

Notes to Financial Statements December 29, 2024, December 31, 2023 and January 1, 2023

Income Taxes

As a single member LLC, the Company is considered a disregarded entity and the results of its operations will be filed with the Member's federal and state income tax returns. As such, the Company itself is typically not subject to income tax liability as the taxable income or loss of the Company is passed through to Member. Therefore, no liability for federal income taxes has been included in the financial statements.

The Company accounts for uncertain tax positions in accordance with ASC No. 740. ASC No. 740 prescribes a recognition threshold and measurement process for accounting for uncertain tax positions and also provides guidance on various related matters such as derecognition, interest, penalties and required disclosures. The Company does not have any entity-level uncertain tax positions.

Advertising and Marketing Costs

Advertising costs, including general brand marketing and contributions to local advertising cooperatives based on a percentage of sales, are expensed when incurred.

Franchised and Company-owned stores in the United States contribute to an Ad Fund that the Company manages on behalf of these stores. The Company is committed under its franchise and other agreements to spend revenues of the Ad Fund on marketing, creative efforts, media support, or related purposes specified in the agreements. Contributions to the Ad Fund are recognized as revenue, while expenditures are included in advertising expenses on the accompanying statements of operations. Expenditures of the Ad Fund are primarily amounts paid to third parties but may also include personnel expenses and allocated costs. At each reporting date, to the extent contributions to the Ad Fund exceed expenditures on a cumulative basis, the excess contributions are recorded in Franchisee payables on the accompanying balance sheets. While no profit is recognized on amounts received by the Ad Fund, when expenditures exceed contributions to the Ad Fund on a cumulative basis, income from operations and net income may be affected due to the timing of when revenues are received and expensed are incurred.

Total advertising and marketing costs for the years ended December 29, 2024, December 31, 2023 and January 1, 2023 was \$437,492, \$231,908 and \$0, respectively, and are included in general and administrative expense on the statements of operations and changes in member's (deficit) equity.

Recently Adopted Accounting Pronouncements

During January 2017, the FASB issued ASU No. 2017-04, *Simplifying the Test for Goodwill Impairment*. ASU No. 2017-04 simplifies how an entity is required to test goodwill for impairment by eliminating Step 2 from the goodwill impairment test. ASU No. 2017-04 (as amended) is effective for annual or any interim goodwill impairment tests in fiscal years beginning after December 15, 2022. There was no significant impact to the financial statements as a result of this adoption.

In June 2016, the FASB issued ASU 2016-13, *Financial Instruments - Credit Losses*. This guidance requires immediate recognition of management's estimates of current expected credit losses. Under the previous model, losses were recognized only as they were incurred, which FASB noted delayed recognition of expected losses that might not yet have met the threshold of being probable. The ASU also broadens the information that an entity must consider in developing its expected credit loss estimated for all assets. The new model applies to all financial instruments that are not accounted for at fair value through net income. This update is effective for financial statements issued for annual periods beginning after December 15, 2022 including interim periods within those fiscal years. The implementation of this ASU has no impact on the financial statements.

Notes to Financial Statements December 29, 2024, December 31, 2023 and January 1, 2023

In January 2021, the FASB issued ASU No. 2021-02, Franchisors - Revenue from Contracts With Customers (Subtopic 952-606): Practical Expedient. The amendments in ASU No. 2021-02 provide a practical expedient related to FASB ASC No. 606, Revenue from Contracts with Customers, that permits franchisors that are not public business entities (PBEs) to account for pre-opening services provided to a franchisee as distinct from the franchise license if the services are consistent with those included in a predefined list within the ASU. Additionally, amendments in ASU No. 2021-02 provide an accounting policy election to recognize the pre-opening services as a single performance obligation. If an entity already has adopted FASB ASC No. 606, the amendments in ASU No. 2021-02 are effective in interim and annual periods beginning after December 15, 2020. The Company has elected to adopt the ASU effective January 2, 2023 using the full retrospective method, resulting in a recasting of the years presented in these financial statements for all contracts in place as of the date of the adoption.

The deferred franchise fee revenues presented on the accompanying balance sheets are comprised of unamortized upfront fees received from franchisees. A summary of changes to the balance sheets for the year ended January 1, 2023 is as follows:

	January 1, 2023			
Deferred revenue, opening balance Subtopic 952-606 adjustment	\$	928,634 (384,259)		
Deferred revenue, adjusted opening balance	\$	544,375		

Subsequent Events

Subsequent events are events or transactions that occur after the balance sheet date but before the financial statements are available to be issued. The Company recognizes in the financial statements the effects of all subsequent events that provide additional evidence about conditions that existed at the date of the balance sheet, including the estimates inherent in the process of preparing the financial statements. The Company's financial statements do not recognize subsequent events that provide evidence about conditions that did not exist at the date of the balance sheet but arose after the balance sheet date and before the financial statements are available to be issued.

The Company has evaluated subsequent events through April 7, 2025, which is the date the financial statements were available to be issued and concluded that there were no additional events or transactions that need to be disclosed.

Notes to Financial Statements December 29, 2024, December 31, 2023 and January 1, 2023

3. Property and Equipment, Net

Property and equipment, net, consists of the following for the years ended December 29, 2024, December 31, 2023 and January 1, 2023:

		December 29, 2024		ember 31, 2023	January 1, 2023		
Computer equipment Vehicles Less accumulated depreciation and	\$	11,192 5,000	\$	8,487 5,000	\$	4,871 -	
amortization		(4,433)		(1,456)		(81)	
Property and equipment, net	\$	11,759	\$	12,031	\$	4,790	

Depreciation and amortization expense was \$2,977, \$1,375 and \$81, respectively, for the years ended December 29, 2024, December 31, 2023 and January 1, 2023, respectively.

4. Goodwill, Net

In December 2022, the Company originally recognized \$928,634 in goodwill related to a deferred revenue liability account which was pushed down to the Company by its Member as a result of the Member's acquisition of Rusty Taco. As a result of the Company's adoption of ASU 2021-02, and as the goodwill was still within one year of its measurement date, goodwill was adjusted by the deferred revenue remeasurement amount for a total of \$384,259. For the years ended December 29, 2024, December 31, 2023 and January 1, 2023, amortization expense related to goodwill was \$54,438, \$54,438 and \$0, respectively.

Estimated amortization expense for each of the next five years and thereafter is as follows:

Years ending December 31:		
2025	\$	54,438
2026		54,438
2027		54,438
2028		54,438
2029		54,438
Thereafter		163,309
Total	_ \$	435,499

Notes to Financial Statements December 29, 2024, December 31, 2023 and January 1, 2023

5. Related-Party Transactions

Parties are considered to be related if one party has the ability, directly or indirectly, to control the other party or exercise significant influence over the other party in making financial and operational decisions, or if they are subject to common control or common significant influence. The significant related-party transactions consist of borrowings from, and payments to, affiliated companies. Due to related party balances are as follows for the years ended December 29, 2024, December 31, 2023 and January 1, 2023:

	Dec	cember 29, 2024	Dec	cember 31, 2023	January 1, 2023		
Due from affiliate Due to affiliate	\$	23,079 (509,354)	\$	- (263,721)	\$	6,784 (16,787)	
Total	_ \$	(486,275)	\$	(263,721)	\$	(10,003)	

Additionally, the Company recognized goodwill in connection with a related party transaction as described in Note 4. The Company also has a franchisee which is a related party as disclosed in Note 2 which comprised \$17,091 and \$12,123 of total revenue and accounts receivable, respectively, for the year ended December 29, 2024. The same franchise comprised \$10,885 and \$12,156 of total revenue and accounts receivable, respectively, for the year ended December 31, 2023.

6. Contract Balances

The beginning and ending contract balances were as follows as of December 29, 2024, December 31, 2023 and January 1, 2023:

	December 29, December 31, 2024 2023		January 1, 2023		
Accounts receivable Deferred franchise fees	\$	61,119	\$ 37,121	\$	35,488
	\$	(443,125)	\$ (706,875)	\$	(544,375)

7. Commitments and Contingencies

Litigation

The Company from time to time may be involved in claims and legal proceedings in the ordinary course of its business. In the opinion of management, the Company is adequately insured against such claims and any ultimate liability arising from such proceedings will not have a material adverse effect on the financial condition, operations or cash flows of the Company.

8. Member's Equity

The Company's LLC operating agreement has a perpetual life. Excess cash flow, tax liability distributions and profits and losses are to be distributed to the Member in accordance with the operating agreement. The liability of the Company's Member is limited to the Member's specific capital balance. Upon liquidation of the Company, the net assets shall be distributed to the Member. During the years ended December 29, 2024, December 31, 2023 and January 1, 2023, the Company's Member made \$725,000, \$0 and \$400,000, respectively, in contributions.

EXHIBIT B DEVELOPMENT AGREEMENT

Rusty Taco Franchising, LLC 2025 FDD | Ex B – Development Agreement 1509.004.004/424787



RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT

RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT

TABLE OF CONTENTS

1.	GRA	NT; EXCLUSIVITY; EXERCISE OF DEVELOPMENT RIGHTS	1
	1.1.	Grant of Development Rights.	1
	1.2.	Exercise of Development Rights.	
	1.3.	Compliance with Development Schedule.	2
	1.4.	Locating Sites for Your Rusty Taco Restaurants.	2
	1.5.	Execution of Franchise Agreements.	
	1.6.	Agreement Regarding Limitation of Our Activities in the Development Area; Reservati Rights	
2.	FEES		3
	2.1.	Development Fee	3
3.		RIGHT NOT TO ISSUE OR SIGN FRANCHISE AGREEMENTS OR DELAY NINGS	3
	3.1.	Satisfaction of Conditions.	
	3.2.	Our Right to Delay Additional Openings.	
	3.3.	Closures and Relocations.	4
4.	DEV	ELOPER'S OBLIGATIONS	5
	4.1.	Satisfaction of Development Schedule.	
	4.2.	Compliance with Applicable Laws.	
	4.3.	Nondisclosure of Confidential Information.	
	4.4.	Noncompetition	
	4.5.	Non-Disparagement & Non-Interference.	7
5.	INDI	EPENDENT CONTRACTOR; INDEMNIFICATION; AND OTHER RESTRICTIVE	${\mathfrak T}$
		ENANTS	7
	5.1.	Independent Contractor.	7
	5.2.	Indemnification.	7
6.	TRA	NSFERS	8
	6.1.	Transfer by Us	8
	6.2.	Transfer by Developer	
	6.3.	Conditions for Approval of Transfer.	
7.	DEF	AULT AND TERMINATION	9
	7.1.	Termination In the Event of Bankruptcy or Insolvency.	-
	7.2.	Termination With Notice and Without Opportunity to Cure.	
	7.3.	Termination With 10-Day Cure Period.	
	7.4.	Termination With 30-Day Cure Period.	
	7.5.	Cross-Default.	
	7.6.	Additional Remedies.	10
	7.7.	Effect of Termination or Expiration.	10
8.	REP	RESENTATIONS	11
	8.1.	Our Representations.	11
	8.2.	Representations of Developer.	11
9.	NOT	ICES	11
	9.1.	Notices	11
		i	

10.	CONS	TRUCTION OF AGREEMENT	11
	10.1.	Entire Agreement.	11
	10.2.	No Waiver	12
	10.3.	Severability	12
	10.4.	Survival of Terms.	12
	10.5.	Captions.	12
	10.6.	Persons Bound	12
	10.7.	Rules of Construction.	12
	10.8.	Business Judgment.	13
	10.9.	Remedies Cumulative.	13
	10.10.	No Third-Party Beneficiary.	13
		No Liability for Related Parties.	
		Counterparts	
11.	APPLICABLE LAW; DISPUTES		
	11.1.	Choice of Law.	13
	11.2.	Requirement to Submit Disputes to Arbitration.	13
	11.3.	Nonexclusivity of Remedy.	
	11.4.	Waiver of Jury Trial.	
	11.5.	Waiver of Punitive Damages.	
	11.6.	Right to Injunctive Relief.	
	11.7.	Attorneys' Fees.	
	11.8.	Limitation of Claims.	
	11.9.	Consent to Jurisdiction.	16

ATTACHMENTS

Attachment A - Data Sheet

Attachment B - Guaranty & Assumption of Obligations

RUSTY TACO FRANCHISING, LLC

DEVELOPMENT AGREEMENT

THIS DEVELOPMENT AGREEMENT ("**Agreement**") is made and entered into as of the Effective Date, by and between Rusty Taco Franchising, LLC, a Delaware limited liability company with its principal office at 5412 W. Plano Pkwy., Suite 100, Plano, Texas 75093 ("**us**"), and the person or entity who signed this Agreement as Developer, as shown on the signature page below ("**you**"). The "**Effective Date**" is the date we sign this Agreement, as shown on the signature page below.

RECITALS

- A. We grant franchises for the right to own and operate restaurants that are currently identified by and use the federally registered trademark *Rusty Taco*® and other certain trade names, service marks, trademarks, logos, emblems, and indicia of origin that we designate from time to time (collectively, the "Marks") and that are required to develop and operate in accordance with our standards and specifications that we may update freely from time to time (collectively, "Standards"), and that offer their customers authentic freshly prepared food made with quality, fresh ingredients inspired by the flavors of Mexico and Texas, along with an assortment of alcoholic and non-alcoholic beverages (each a "Rusty Taco Restaurant").
- B. We also grant the right to acquire multiple franchises to persons or entities who we believe have the financial capability and experience, and are willing to commit those resources, to develop and operate multiple Rusty Taco Restaurants within a specific area and to open them in satisfaction of a development schedule (the "**Development Rights**").
- C. Based on your own investigation and diligence, you have requested that we grant you the Development Rights for the Development Area (as defined below and described herein) and, to support your request, you and, as applicable, your owners have provided us with certain application materials ("Application Materials"). In reliance on, among other things, the Application Materials, we are willing to grant you the Development Rights on the terms contained in this Agreement.

AGREEMENT

In consideration of the foregoing Recitals (which are incorporated herein), the promises contained in this Agreement, and other valuable consideration, the receipt and sufficiency of which are acknowledged, the parties agree as follows:

1. GRANT; EXCLUSIVITY; EXERCISE OF DEVELOPMENT RIGHTS

1.1. Grant of Development Rights. We hereby grant you the Development Rights, which must be exercised in strict compliance with this Agreement. The Development Rights may be exercised from the Effective Date and, unless sooner terminated as provided herein, will expire on the earlier of (1) the opening of the last Rusty Taco Restaurant required to be open under the Development Schedule shown on the Data Sheet attached hereto as Attachment A (the "Development Schedule") opens for regular business, or (2) the date by which the last such restaurant is required to be open in satisfaction of the Development Schedule (the "Term"). You accept the grant of the Development Rights and agree to, at all times, faithfully, honestly, and diligently perform your obligations under this Agreement and fully exploit the Development Rights during the Term and throughout the entire Development Area identified on the attached Data Sheet (the "Development Area"). You agree that non-traditional locations (i.e., any location that (1) is part of a larger venue or facility, (2) is not generally and easily accessible to the general public, or (3) whose operating hours are limited to those imposed by the owner or operator of the venue in which it sits (for example, military bases, shopping malls, hotels, school and college campuses,

1

airports, train stations, travel plazas, toll roads, beaches, parks and other seasonal facilities, government buildings and establishments, prisons, hospitals, convenience stores, trucks, casinos, sports or entertainment venues or stadiums, and retail shop locations being sublet under a lease to a master concessionaire) that become available during the Term and are located within the Development Area are deemed to be excluded from the Development Area unless we indicate otherwise on the Data Sheet attached as Attachment A hereto. You further acknowledge and agree that this Agreement does not grant you or your Affiliates the right or license to use any of the Marks. Your or your Affiliates' right to operate a Rusty Taco Restaurant and use the Marks is derived solely from the Franchise Agreements that you or your Affiliates may enter into with us pursuant to this Agreement.

- 1.2. Exercise of Development Rights. You must perform all of your obligations under this Agreement, and you may not subcontract or delegate any of those obligations to any third parties; except that, we may also your approved Affiliates to execute Franchise Agreements to fulfill your development obligations. The Development Rights may only be exercised for Rusty Taco Restaurants to be located in the Development Area. To maintain your rights under this Agreement you must, among other things, satisfy the Development Schedule, as described in Section 1.3. Failure to do so will be grounds for either removal of the limitations on our activities as described in Section 1.6 or, at our option, a termination of this Agreement. Upon expiration or termination of this Agreement, the only territorial protection you will retain will be the Protected Territory granted under each then-existing Franchise Agreement between us and you.
- 1.3. **Compliance with Development Schedule.** The Development Schedule is comprised of two (2) components: (i) a restaurant opening requirement (the "Opening Requirement") with multiple interim periods (each a "Development Period") and (ii) a Franchise Agreement execution requirement (the "Contract Execution Requirement") based on agreed upon dates by which restaurants are projected to open (each a "Projected Opening Date"). Each Rusty Taco Restaurant you are required to open in satisfaction of the Development Schedule will be subject to a separate Franchise Agreement as described in Section 1.5. To satisfy the Opening Requirement, the minimum number of Rusty Taco Restaurants required to be open during and as of the end of each Development Period must be open and operating, each in accordance with its Franchise Agreement. To satisfy the Contract Execution Requirement, you or your Affiliate that we approve must execute a Franchise Agreement for each Restaurant at or prior to the applicable execution date set forth in the Development Schedule, which must be a date no later than 180 days prior to the Projected Opening Date for the applicable Restaurant. The Development Schedule is not our representation, express or implied, that the Development Area can support, or that there are or will be sufficient sites for, the number of Rusty Taco Restaurants specified in the Development Schedule or during any particular Development Period. We are relying on your knowledge and expertise of the Development Area and your representation that you have conducted your own independent investigation and have determined that you can satisfy the development obligations under each Development Period of the Development Schedule.
- 1.4. Locating Sites for Your Rusty Taco Restaurants. Despite any assistance we may provide, you are entirely responsible for locating and presenting to us proposed sites for Rusty Taco Restaurants in the Development Area as necessary to comply with the Development Schedule (each a "Site"). You agree to give us all information and materials we request to assess each proposed Site as well as your or your proposed Affiliate's financial and operational ability to develop and operate a Rusty Taco Restaurant at the proposed Site. We have the absolute right to reject any proposed Site or any Affiliate (a) that does not meet our criteria or (b) if you or your Affiliates are not then in compliance with any existing Franchise Agreements executed pursuant to this Agreement or operating your or their Rusty Taco Restaurants in compliance with our mandatory Standards. We agree to use our reasonable efforts to review and evaluate your proposed Sites within 30 days after we receive all requested information and materials. If we accept a proposed Site, you or your approved Affiliate must sign a separate Franchise Agreement for the Site within 15 days after we provide you with an execution copy of the Franchise Agreement, failing which, we may withdraw our acceptance. Our approval of any proposed Site is entirely for our own purposes and, by approving your Site, we are not representing or guaranteeing that it will perform as you or we expect it to do. We are not responsible if the Site we recommend or approve fails to meet your expectations.

- 1.5. Execution of Franchise Agreements. Within 15 days after we have accepted a Site, you or your approved Affiliates must sign our then-current form of Franchise Agreement and related documents for your such Site. Each Franchise Agreement will govern the development and operation of the Rusty Taco Restaurant at the accepted Site identified therein. You are not entitled, and we are not required, to execute more than the number of Franchisee Agreements necessary to satisfy the Development Schedule.
- 1.6. Agreement Regarding Limitation of Our Activities in the Development Area; Reservation of Rights. Subject to your and your Affiliates' full compliance with your and their obligations under any agreement with us or our affiliates, and subject to the retained rights described in this Section, neither we nor any of our Affiliates shall establish, or authorize any person or entity other than you or any of your Affiliates to establish, a Rusty Taco Restaurant in the Development Area during the Term. Notwithstanding the foregoing, you expressly agree that we and our Affiliates retain, without compensation to you, all such rights and all other rights that are not expressly granted to you in this Agreement and that we do not expressly agree to refrain from doing, including:
- 1.6.1. the right to develop, establish, and operate anywhere (including in the Development Area), and the right to grant to others the right to operate anywhere (including in the Development Area), other businesses using the Marks or other names or marks;
- 1.6.2. the right to, and to grant to others the right to, solicit customers and advertise and promote the Rusty Taco franchise system inside and outside the Development Area;
- 1.6.3. the right to acquire, be acquired by, merge, affiliate with, or engage in any transaction with other businesses (whether competitive or not) located anywhere and, even if such businesses are located in the Development Area, convert the other businesses to Rusty Taco Restaurants;
- 1.6.4. the right to operate, and grant license to others the right to operate, Rusty Taco Restaurants outside the Development Area and at any Non-Traditional Location within the Development Area; and
- 1.6.5. the right to engage, directly or indirectly, at wholesale, retail or otherwise, in the production, distribution, license, and sale of any and all food, beverages, or other services and products, under the Marks, or under other names or marks, within and outside the Development Area, through any method of distribution, including, but not limited to, mail order catalogs, the Internet, and other permanent, temporary or seasonal food service facilities, carts, food trucks, or kiosks (which facilities may provide, in whole or in part, the products and services offered by a Rusty Taco Restaurant) regardless of the proximity to, or the competitive impact on, your operations in the Development Area.

2. **FEES**

2.1. **Development Fee.** Upon execution of this Agreement, you shall pay us a Development Fee in the amount set forth in Attachment A ("**Development Fee**"). The Development Fee is fully earned by us when paid and is not refundable, in whole or in part, under any circumstances.

3. OUR RIGHT NOT TO ISSUE OR SIGN FRANCHISE AGREEMENTS OR DELAY OPENINGS

3.1. <u>Satisfaction of Conditions.</u> For each Franchise Agreement that you are required to execute under this Agreement, you shall submit to us a request for Franchise Agreement on our standard form and provide us with all information we request in order to evaluate your request. If you intend to have an Affiliate sign the requested Franchise Agreement, you must also provide us with information we request regarding the proposed Affiliate and its owners. We need not issue a Franchise Agreement if you or your owners do not meet our then-

current operational, financial, and legal standards, qualifications, and conditions for new Rusty Taco Restaurant owners (the "Conditions"), including, without limitation, the following Conditions:

- 3.1.1. Operational Conditions: You must be in compliance with the Development Schedule and this Agreement, and you or your Affiliates must be in compliance with any other agreement with us or our Affiliates. You and your Affiliates must be operating your and their existing Rusty Taco Restaurants, if any, and be capable, in our determination, of developing and operating the proposed Restaurant in accordance with its Franchise Agreement.
- 3.1.2. <u>Financial Conditions</u>: You or your Affiliate and your or their owners, as applicable, must demonstrate, to our reasonable satisfaction, that you and they satisfy our then-current financial criteria for developers and owners of Rusty Taco Restaurants. You and your Affiliates and your or their owners, as applicable, must have been and are faithfully performing all terms and conditions under each of their existing Franchise Agreements with us. You and your Affiliates must not then be in default and must not have been in default during the rolling 12 months preceding your request for the Franchise Agreement, of any monetary obligations owed to us or our Affiliates under any agreement with us or our Affiliates. You acknowledge and agree that it is vital to our interests that each of our franchisees be financially sound to avoid failure of a Rusty Taco Restaurant and that such failure would adversely affect the reputation and goodwill associated with the Marks and Rusty Taco franchise system.
- 3.1.3. <u>Legal Conditions</u>: If you intend the Franchise Agreement to be executed by your Affiliate, you must have provided us with a list of, and applications from, each of its owners, a list of the percentage ownership held by each owner, and copies of its as-filed articles of formation, its governing documents, and resolutions establishing the authority of the individual who will be signing the Franchise Agreement on behalf of the Affiliate. In addition, you must have submitted to us, in a timely manner, all other information and documents we request as a basis for the issuance of the requested Franchise Agreement.
- 3.2. Our Right to Delay Additional Openings. We may delay the development and/or opening of additional Rusty Taco Restaurants within the Development Area for the time period we deem best if we believe, in our sole judgment, when you submit your application for another franchise for Rusty Taco Restaurants or after you have developed and constructed but not yet opened a particular Rusty Taco Restaurant, that you or your Affiliate, as applicable, is not yet operationally, managerially, or otherwise prepared (no matter the reason) to develop, open, and/or operate the additional Rusty Taco Restaurant in full compliance with our Standards. The period of delay under this Section will not, in our reasonable opinion, cause you to breach your development obligations under the Development Schedule (unless we are willing to extend the Development Schedule proportionately to account for the delay).
- Franchise Agreement signed pursuant to this Agreement is closed with our approval in accordance with the applicable Franchise Agreement, you must develop a replacement Rusty Taco Restaurant in order to comply with your obligations under this Agreement. To count toward the Opening Requirements for the Development Period (described on Attachment A) in which the Rusty Taco Restaurant closed, you must re-open the closed restaurant, either at its original location or at another location we approve, within a reasonable time (not to exceed 120 days) after you ceased operation of the closed restaurant. If, during the Term of this Agreement, you sell or transfer a Rusty Taco Restaurant in accordance with the terms of the applicable Franchise Agreement, the transferred restaurant will continue to be counted in determining whether you have complied with the Development Schedule so long as it continues to be operated as a Rusty Taco Restaurant. If the transferred restaurant ceases to be operated as a Rusty Taco Restaurant during the Term of this Agreement, you shall develop a replacement Restaurant within a reasonable time (not to exceed 120 days) thereafter.

4. **DEVELOPER'S OBLIGATIONS**

- 4.1. <u>Satisfaction of Development Schedule.</u> You agree to comply with the Development Schedule, including the Opening Requirements for each Development Period described on Attachment A and the Contract Execution Requirements for each Projected Opening Date.
- 4.2. <u>Compliance with Applicable Laws</u>. You shall fully comply with all Applicable Laws when exercising the Development Rights and fulfilling your obligations under this Agreement. It is your sole responsibility to apprise yourself of the existence and requirements of all such requirements and to adhere to them at all times during the Term of this Agreement.
- 4.3. Nondisclosure of Confidential Information. In connection with your rights under this Agreement, you and your owners and personnel may from time to time be provided and/or have access to non-public information about the Rusty Taco franchise system and the operation and development of Rusty Taco Restaurants (including information arising from the Rusty Taco Restaurants you develop) (collectively, the "Confidential Information"), including: (1) criteria for Site selection; (2) training programs and operations materials; (3) the Standards; (4) market research and marketing strategies (including expansion strategies and targeted demographics); (5) specifications for, suppliers of, and methods of ordering, products and services; (6) any software or technology which is proprietary to us or the Rusty Taco franchise system, including digital passwords and identifications and any source code of, and data, reports, and other printed materials generated by, the software or similar technology; (7) knowledge of the operating results and financial performance of Rusty Taco Restaurants; (8) information generated by you or used or developed by you, including information relating to market research; and (9) any other information designated as confidential or proprietary by us.

All Confidential Information will be owned by us (other than personally identifiable information relating to the employees, officers, contractors, owners or other personnel of you, your affiliates, or your Rusty Taco Restaurants, and/or such other personally identifiable information designated by us from time to time). You acknowledge and agree that: (i) you will not acquire any interest in any of our Confidential Information, other than the right to use it as we specify under this Agreement or the Franchise Agreements you sign, in each case in accordance with the terms of such agreement; and (ii) our Confidential Information is proprietary, includes our trade secrets, and is disclosed to you only on the condition that you will protect it. You acknowledge that any unauthorized use or disclosure of our Confidential Information would be an unfair method of competition and a breach of trust and confidence and will result in irreparable harm to us and our affiliates. You (and if you are conducting business as an entity, each of your owners) therefore agree that during and after the Term you will, and will cause each of your respective current and former spouses, immediate family members, owners, officers, directors, employees, representatives, affiliates, successors and assigns to:

- (a) process, retain, use, collect, and disclose our Confidential Information strictly to the limited extent, and in such a manner, as necessary for exercise of your Development Rights in accordance with this Agreement, and/or the operation of Rusty Taco Restaurants under the respective Franchise Agreements, and not for any other purpose of any kind;
- (b) process, retain, use, collect, and disclose our Confidential Information strictly in accordance with the privacy policies and system standards we establish from time to time, and our and our representative's instructions;
- (c) keep confidential and not disclose, sell, distribute, or trade our Confidential Information to any person other than those of your personnel and representatives who need to know such Confidential Information for the purpose of assisting you in exercising your Development Rights in accordance with this Agreement, and/or operating Rusty Taco Restaurants in accordance with Franchise Agreements with us; and you agree that you will be responsible for any violation of this requirement by any of your personnel and representatives;

- (d) not make unauthorized copies of any of our Confidential Information;
- (e) adopt and maintain administrative, physical and technical safeguards to prevent unauthorized use or disclosure of any of our Confidential Information, including by establishing reasonable security and access measures, restricting its disclosure to key personnel, and/or by requiring persons who have access to such Confidential Information to be bound by contractual obligations to protect such Confidential Information and preserve our rights and controls in such Confidential Information, in each case that are no less protective or beneficial to us than the terms of this Agreement; and
- (f) at our request, destroy or return any of the Confidential Information.

Confidential Information does not include information, knowledge, or know how, which is lawfully known to the public without violation of applicable law or an obligation to us or our affiliates.

We are not making any representations or warranties, express or implied, with respect to the Confidential Information. We and our affiliates have no liability to you for any errors or omissions from the Confidential Information.

All ideas, concepts, techniques, or materials relating to this Agreement, your Development Rights, and/or the system created by you, your owners or your employees (or for you, your owners or your employees), whether or not protectable intellectual property, must be promptly disclosed to us and will be our sole and exclusive property, part of the System, and works made-for-hire for us. To the extent that any item does not qualify as a "work made-for-hire" for us, you hereby waive all moral rights in that item, assign ownership of that item, and all related rights to that item, to us and agree to take whatever action (including signing assignment or other documents) we request to evidence our ownership or to help us obtain intellectual property rights in the item.

4.4. <u>Noncompetition.</u> You acknowledge that we have granted you the Development Rights in consideration of and reliance upon your agreement to deal exclusively with us. You therefore agree that, during the Term, neither you nor any of your owners, your or your owners' Affiliates, or the officers, directors, managers or immediate family members of any of the foregoing will: (i) have any direct or indirect interest as an owner – whether of record, beneficially, or otherwise – in a Competitive Business (defined below), wherever located or operating (except that equity ownership of less than five percent (5%) of a Competitive Business whose stock or other forms of ownership interest are publicly traded on a recognized United States stock exchange will not be deemed to violate this subparagraph); (ii) perform services as a director, officer, manager, employee, consultant, representative, lessor, or agent for a Competitive Business, wherever located or operating; (iii) divert or attempt to divert any actual or potential business or customer of any Rusty Taco Restaurant to a Competitive Business; or (iv) directly or indirectly, appropriate, use or duplicate the Rusty Taco franchise system or Standards, or any portion thereof, for use in any other business or endeavor.

Commencing on the later of: (a) a Transfer permitted under Article 6 of this Agreement, expiration of this Agreement, or termination of this Agreement (regardless of the cause for termination); or (b) a final court order (after all appeals have been taken) with respect to any of the foregoing events or with respect to enforcement of this Section, and continuing for an uninterrupted period of two (2) years thereafter, you, your owners, and your and your owners' immediate family members shall not either directly or indirectly, for yourselves, or through, on behalf of, or in conjunction with any person, persons, or legal entity, own, maintain, advise, operate, engage in, be employed by, make loans to, lease property to, invest in, provide any assistance to, or have any interest in (as an owner or otherwise) or relationship or association with any Competitive Business that: (i) is, or is intended to be, located in the Development Area; (ii) is within a five-mile from the outer boundary of the Development Area; or (iii) is within a five-mile radius of any other Rusty Taco Restaurant. If any of your owner ceases to be an owner for any reason during the Term of this Agreement, the foregoing covenant will apply to the departing owner for a two-year period beginning on the date such person ceases to meet the definition of an owner. The time period

during which the post-term non-competition obligations in this Section apply will be tolled for any period of noncompliance.

The term "**Competitive Business**" means any restaurant, food service or other business (other than a Rusty Taco Restaurant that is operating pursuant to a Franchise Agreement with us) (i) deriving more than 10% of its revenue (excluding revenue derived from the sale of alcoholic beverages) from the sale of Mexican-inspired food, or (ii) that grants franchises or licenses for the operation of any of the foregoing or provides services to the franchisor or licensor of any of the foregoing.

4.5. Non-Disparagement & Non-Interference. During and after the Term, you agree not to (and to use your best efforts to cause your current and former owners, members, officers, directors, principals, agents, partners, employees, representatives, attorneys, spouses, heirs, Affiliates, successors and assigns not to) disparage or otherwise speak or write negatively, directly or indirectly, of us, our affiliates, any of our or our affiliates' owners, directors, officers, employees, representatives or Affiliates, current and former franchisees or developers of us or our affiliates, the Rusty Taco brand, the Rusty Taco franchise system, any Rusty Taco Restaurant, any business using the Marks, any other brand or service-marked or trademarked concept of us or our affiliates, or which would subject the Rusty Taco brand or such other brands to ridicule, scandal, reproach, scorn, or indignity, or which would negatively impact the goodwill of us, the Rusty Taco brand or such other brands. You further agree that, either during or after the Term, neither you nor any of your owners, your or your owners' Affiliates, or the officers, directors, managers or immediate family members of any of the foregoing, will: (i) interfere or attempt to interfere with our or our affiliates' relationships with any vendors or consultants; or (ii) engage in any other activity which might injure the goodwill of the Marks or the Rusty Taco franchise system.

5. INDEPENDENT CONTRACTOR; INDEMNIFICATION; AND OTHER RESTRICTIVE COVENANTS

- 5.1. **Independent Contractor**. The activities you conduct under this Agreement will be conducted as an independent contractor. Nothing contained in this Agreement will create or be construed to create a partnership, joint venture, agency or employment relationship between the parties. Neither party has any fiduciary obligations to the other or is liable for the debts or obligations of the other. Neither party shall have the right to bind the other, transact business in the other party's name, or in any manner make any promises or representations on behalf of the other party, unless otherwise agreed in writing by the parties. We do not participate in the hiring, promoting, disciplining, or discharging of your employees or in setting or paying wages or benefits to your employees, and you acknowledge that we have no power, responsibility, or liability in respect to the hiring, promoting, disciplining, or discharging of employees or in setting or paying their wages. You shall conspicuously identify yourself and the business contemplated under this Agreement in all dealings with your employees, customers, contractors, suppliers, public officials, and others, as an entity separate and apart from us and do nothing to imply that you are acting on our behalf as you conduct your activities. Additionally, you must communicate to all employees that you, not us, are their employer; and you must ensure that no payroll checks or other employment-related documents (such as job applications and W-2s) contain or reference the Marks or our name. Each of the parties will file its own tax, regulatory, and payroll reports with respect to its respective employees and operations, saving and indemnifying the other party hereto of and from any liability of any nature whatsoever by virtue thereof.
- 5.2. <u>Indemnification.</u> You shall indemnify and hold harmless to the fullest extent permitted by law, us, our Affiliates and our and their respective directors, officers, employees, owners, agents, successors and assigns (collectively, "Indemnitees") from any and all "losses and expenses" incurred in connection with any litigation or other form of adjudicatory procedure, claim, demand, investigation, or formal or informal inquiry (regardless of whether same is reduced to judgment) or any settlement thereof which arises directly or indirectly from, as a result of, or in connection with your exercise of the Development Rights under this Agreement, your employment or other relationship with your employees, contractors, or other third parties (collectively, "event"), and regardless of whether same resulted from any strict or vicarious liability imposed by law on the Indemnitees;

provided, however, that this indemnity will not apply as to an Indemnitee in respect of any losses and expenses arising from the gross negligence of such Indemnitee (except to the extent that joint liability is involved, in which event the indemnification provided in this Agreement will extend to any finding of comparative negligence or contributory negligence attributable to you). For the purpose of this Section, the term "losses and expenses" will be deemed to include compensatory, exemplary, or punitive damages; fines and penalties; attorneys' fees; experts' fees; court costs; costs associated with investigating and defending against claims; settlement amounts; judgments; compensation for damages to our reputation and goodwill; and all other costs associated with any of the foregoing losses and expenses. Under no circumstances will an Indemnitee be required or obligated to seek recovery from third parties or otherwise mitigate its losses in order to maintain a claim under this provision, and its failure to seek such recovery or mitigate its losses and expenses will in no way reduce the amounts recoverable by such Indemnitee under this provision. You must give us prompt notice of any event of which you are aware, for which indemnification is required, and, at your expense and risk, we may elect to assume (but under no circumstance is obligated to undertake) the defense and/or settlement thereof, provided that we will seek your advice and counsel. Any assumption by us will not modify your indemnification obligation. We may, in our sole and absolute discretion, take such actions as we deem necessary and appropriate to investigate, defend, or settle any event or take other remedial or corrective actions with respect thereof as may be, in our sole and absolute discretion, necessary for the protection of the Indemnitees or the Rusty Taco franchise system. This provision survives termination or expiration of this Agreement, regardless of the reason for expiration or termination.

6. TRANSFERS

- 6.1. Transfer by Us. We have the right to delegate the performance of any portion or all of our rights and obligations under this Agreement to third-party designees. You represent that you have not signed this Agreement in reliance on any particular person or entity remaining with us in any capacity. We may change our ownership or form or assign this Agreement and any other agreement to a third party without restriction.
- 6.2. Transfer by Developer. Your rights and duties under this Agreement are personal to you (or your owners), and we have granted you the Development Rights in reliance upon our perceptions of your (or your owners') individual or collective character, skill, aptitude, attitude, business ability, and financial capacity. Accordingly, neither you nor any owners, nor any of your or their permitted successors or assigns, shall sell, assign, transfer, convey, give away, pledge, mortgage, or otherwise dispose of or encumber this Agreement (or any direct or indirect interest in this Agreement), the Development Rights, or any direct or indirect ownership interest in you (regardless of its size) (each, a "Transfer"), without our prior written. Any Transfer without our prior written approval is a material breach of this Agreement and has no effect.

If you intend to list your Development Rights for sale with any broker or agent, you shall do so only after obtaining our written approval of the broker or agent and of the listing agreement and any advertising materials. You may not use any Mark in advertising the transfer or sale of your Development Rights or of any ownership in you without our prior written consent.

- 6.3. <u>Conditions for Approval of Transfer.</u> We may consider, and you will provide or assist us in compiling, any information we deem necessary or appropriate in connection with our assessment of a proposed Transfer. If we elect to approve a proposed Transfer, we may, in our sole discretion, condition our approval in any manner we deem necessary and appropriate to protect the Rusty Taco brand and our interests in the Rusty Taco franchise system and this Agreement, including any of the following (each of which you agree is reasonable):
- (1) The transferee must demonstrate to our satisfaction that the transferee meets our then-current educational, managerial, and business standards; possesses a good moral character, business reputation, financial capabilities, and credit rating; has the aptitude and ability to exercise the Development Rights; and that terms of the sale will not materially and adversely affect the post-transfer viability of any Restaurant owned by you or your Affiliates;

- (2) you and any person obligated under this Agreement must be in compliance with your or its obligations;
- (3) you and the proposed transferee and its owners (if the transferee is an Entity) must provide all information and documents we request regarding the Transfer and the proposed transferee and its owners;
- (4) you must provide us with executed versions of any relevant documents to effect the Transfer, and all other information we request about the proposed Transfer;
- (5) if you, the transferor or any third party offers the transferee financing for any part of the purchase price, all of the transferee's obligations under promissory notes, agreements, or security interests reserved in your Development Rights must be subordinate to the transferee's obligation to pay all amounts due to us, our Affiliates, and third party vendors and otherwise agree to comply with this Agreement (or any applicable Franchise Agreement with us);
- (6) you (and your owner(s)) must sign a general release, in a form satisfactory to us, of any and all claims against us and our shareholders, officers, directors, employees and agents;
- (7) each transferee and all its direct and indirect owners and their respective spouses must execute and deliver to us a copy of our then current form of Guaranty undertaking personally to be bound, jointly and severally, by all provisions of this Agreement (or our then-current form of Area Development Agreement if the Transfer involves transfer of a controlling interest) and any other ancillary agreements;
- (8) you must pay all amounts owed to us, our Affiliates, and third-party vendors and have submitted all required reports and statements under this Agreement and any Franchise Agreement with us;
- (9) you and your owners must not have violated any provision of this Agreement or any other agreement with us or our Affiliates during both the sixty (60) day period before you requested our consent to the Transfer and the period between your request and the effective date of the Transfer;
- (10) if the Transfer involves a transfer of the Development Rights or a transfer of a controlling interest in you, the transferee, at our request, must sign our then-current form of area development agreement and related documents, any and all of the provisions of which may differ materially from any and all of those contained in this Agreement;
- (11) you must pay or cause to be paid to us a transfer fee in the amount of Five Thousand Dollars (\$5,000) plus reasonable attorneys' fees, and reimburse us our reasonable expenses (in addition to the transfer fee payable under each Franchise Agreement transferred in accordance with Section 6.3 (13) hereto);
- (12) You and your transferring owners (and your and their immediate family members) will not, for two (2) years beginning on the transfer's effective date, engage in any of the activities proscribed in Sections 4.4 and 4.5 above; and
- (13) the Transfer of this Agreement must not be made separate and apart from the Transfer to the transferee of all Franchise Agreements that were signed pursuant to this Agreement.

7. **DEFAULT AND TERMINATION**

7.1. **Termination In the Event of Bankruptcy or Insolvency.** You shall be deemed to be in default under this Agreement, and we may terminate this Agreement immediately upon notice, if you become insolvent or make a general assignment for the benefit of creditors; if a petition in bankruptcy is filed by you or such a petition is filed and against you and you do not oppose it; if you are adjudicated as bankrupt or insolvent; if a bill in equity

or other proceeding for the appointment of a receiver for you or other custodian for your business or assets is filed and consented to by you; if a receiver or other custodian (permanent or temporary) of your assets or property, or any part thereof, is appointed by any court of competent jurisdiction; if proceedings for a composition with creditors under any state or federal law is instituted by or against you; if a final judgment remains unsatisfied or of record for 30 days or longer (unless a *supersedeas* bond is filed); if you are dissolved; if execution is levied against your business or property; if judicial, non-judicial or administrative proceedings to foreclose any lien or mortgage against premises or assets or equipment of a Rusty Taco Restaurant you or your Affiliates own is instituted against you and not dismissed within 30 days; or if the real or personal property of the Rusty Taco Restaurant you or your Affiliates own is sold after levy thereupon by any sheriff, marshal, or constable.

- Termination With Notice and Without Opportunity to Cure. We have the right to terminate this Agreement, which termination will become effective upon delivery of notice without opportunity to cure if: (a) you fail to adhere to the Development Schedule; (b) you or any owner is convicted of, or pleads no contest to, a felony, a crime involving moral turpitude, or any other crime or offense that we believe is reasonably likely to have an adverse effect on the System; (c) there is any Transfer or attempted Transfer in violation of Section 6.2 of this Agreement; (d) you or any owner fails to comply with the confidentiality or noncompetition covenants in Sections 4.3 and 4.4 of this Agreement; (e) you or any owner has made any material misrepresentations in the Application Materials or otherwise in connection with your application for Development Rights; (f) we deliver to you three or more written notices of default pursuant to this Article 7 within any rolling 12-month period, whether or not the defaults described in such notices ultimately are cured; (g) if you are an individual, in the event of your death or permanent Incapacity; or (h) you terminate, without cause, any Franchise Agreement signed pursuant to this Agreement or we terminate any such Franchise Agreement based on your breach.
- 7.3. Termination With 10-Day Cure Period. We have the right to terminate this Agreement, which termination will become effective upon delivery of written notice of termination, if you fail to cure the following defaults within 10 days after delivery of written notice: (a) failure to pay any amounts due to us; (b) failure to pay any amounts due to your trade creditors (unless such amount is subject to a bona fide dispute); or (c) failure to pay any amounts for which we have advanced funds for or on your behalf, or upon which we are acting as guarantor of your obligations.
- 7.4. <u>Termination With 30-Day Cure Period.</u> Except as otherwise provided in this Article 7, we have the right to terminate this Agreement, which termination will become effective upon delivery of written notice of termination, if you fail to cure any curable default within 30 days after delivery of written notice.
- 7.5. <u>Cross-Default.</u> Any default under any agreement between you or your Affiliates and us or our Affiliates, which you or your Affiliates, as applicable, fail to cure within any applicable cure period, will be considered a default under this Agreement and will provide an independent basis for termination of this Agreement.
- 7.6. Additional Remedies. If you are in default of this Agreement, we may, in our sole discretion, elect to reduce the number of Rusty Taco Restaurants which you may establish pursuant to the Development Schedule or reduce the size of your Development Area. If we elect to exercise this remedy as set forth above, you agree to continue to develop Rusty Taco Restaurants in accordance with your rights and obligations under this Agreement, as modified. Our exercise of the remedy under this Section 7.6 will not constitute our waiver of our right to terminate this Agreement at any time with respect to a subsequent event of default of a similar or different nature.
- 7.7. <u>Effect of Termination or Expiration.</u> Upon termination or expiration of this Agreement, you shall have no further right to acquire any additional franchises to develop any Rusty Taco Restaurants for which a Franchise Agreement has not been signed.

8. **REPRESENTATIONS**

8.1. <u>Our Representations</u>. We represent and warrant that: (a) we are duly organized and validly existing under the law of the state of our formation; (b) we are duly qualified and authorized to do business in each jurisdiction in which our business activities or the nature of the properties we own requires such qualification; and (c) the execution of this Agreement and the performance of the transactions contemplated by this Agreement are within our corporate power and have been duly authorized.

8.2. **Representations of Developer**. You represent and warrant to us that:

- 8.2.1. the ownership information set forth in Attachment A is accurate and complete in all material respects, and you will notify us in writing prior to any change in such information;
- 8.2.2. (a) you are duly organized and validly existing under the law of the state of your formation; (b) you are duly qualified and authorized to do business in each jurisdiction in which your business activities or the nature of the properties you own require such qualification; (c) your organizational charter and governing documents will at all times provide that your activities are confined exclusively to the activities described here and, if applicable, the operation of Rusty Taco Restaurants under Franchise Agreements executed pursuant to this Agreement; (d) neither you nor any of your Affiliates or owners own, operate, or have any financial or beneficial interest in any business in Competitive Business; and (e) the execution of this Agreement and the performance of the transactions contemplated by this Agreement are permitted under your charter and governing documents;
 - 8.2.3. you have conducted an independent investigation of the franchise opportunity; and
- 8.2.4. neither you nor your owners, officers, directors, managers, partners, agents, or employees, or their respective interests therein is now and will not be, during the Term, (a) identified, either by name or an alias, pseudonym, or nickname, on any of the lists of restricted or denied parties maintained by the United States government, or (b) in violation of any law prohibiting money laundering or the aid or support of persons who conspire to commit acts of terror against any person or government, including acts prohibited by the U.S. Patriot Act (http://epic.org/privacy/terrorism/hr3162.html), U.S. Executive Order 13224 (http://www.treasury.gov/resource-center/sanctions/Programs/Documents/terror.pdf), or any similar law.

9. **NOTICES**

9.1. <u>Notices.</u> All written notices, reports, and payments permitted or required to be delivered by this Agreement will be deemed to be delivered on the earlier of the date of actual delivery or one of the following: (i) at the time delivered by hand, (ii) at the time delivered to the server in case of delivery via computer transmission and, in the case of the Royalty Fees, Fund contributions, and other amounts due, at the time we actually receive electronic payment, or (iii) one (1) business day after being placed in the hands of a nationally recognized commercial courier service for next business day delivery. Notices must be sent to you as shown on the attached Data Sheet or the most current principal business address of which the notifying party has notice; except that, it will always be deemed acceptable during the Term to send notices to you at the address of any of your or your Affiliates' Rusty Taco Restaurants. Notices to us must be sent to the address shown on the first page of this Agreement, Attention: President, with a copy (which shall not constitute notice) to Legal Department.

10. CONSTRUCTION OF AGREEMENT

10.1. **Entire Agreement**. This Agreement, and any other agreements executed by the parties concurrently with the parties' execution of this Agreement, represents the entire fully integrated agreement between the parties and supersedes all other negotiations, agreements, representations, and covenants, oral or written. Notwithstanding the foregoing, nothing in this Agreement will disclaim or require you to waive reliance

on any representation that we made in the franchise disclosure document (including its exhibits and amendments) that we delivered to you in connection with this franchise offering. Except for those changes permitted to be made unilaterally by us hereunder, no amendment, change, or variance from this Agreement will be binding on either party unless mutually agreed to in writing by the parties and executed by their authorized officers or agents.

The following provision applies if you or the franchise granted hereby are subject to the franchise registration or disclosure laws in California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, or Wisconsin: No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

- 10.2. <u>No Waiver.</u> No delay, waiver, omission, or forbearance on our part to exercise any right, options, duty, or power arising out of any breach or default by you or any of you owners under this Agreement will constitute our waiver to enforce any such right, option, duty, or power against you or your owners, or as to a subsequent breach or default by you or any of your owners.
- 10.3. <u>Severability</u>. The invalidity or unenforceability of any provision of this Agreement will not affect the validity or enforceability of any other provision of this Agreement.
- 10.4. **Survival of Terms.** Any provision or covenant of this Agreement which expressly or by its nature imposes obligations beyond the expiration or termination of this Agreement will survive such expiration or termination.
- 10.5. <u>Captions.</u> All captions in this Agreement are intended for the convenience of the parties, and none will be deemed to affect the meaning or construction of any provision of this Agreement.
- 10.6. **Persons Bound.** This Agreement will be binding on the parties and their respective successors and assigns.
- 10.7. **Rules of Construction**. Neither this Agreement nor any uncertainty or ambiguity in this Agreement will be construed or resolved against the drafter of this Agreement, whether under any rule of construction or otherwise. Terms used in this Agreement will be construed and interpreted according to their ordinary meaning. If any provision of this Agreement is susceptible to two or more meanings, one of which would render the provision enforceable and the other(s) which would render the provision unenforceable, the provision will be given the meaning that renders it enforceable.

References in the Agreement to "we," "us," and "our" include any of our Affiliates with whom you deal. The term "Affiliate" means any person or entity directly or indirectly owned or controlled by, under common control with, or owning or controlling you or us. "Control" means the power to direct or cause the direction of management and policies. "Including" means "including, without limitation."

If two or more persons are at any time the owners of the Franchise and your Restaurant, whether as partners or joint venturers, their obligations and liabilities to us will be joint and several. References to "owner" mean any person holding a direct or indirect ownership interest (whether of record, beneficially, or otherwise) or voting rights in you (or a transferee of the Agreement and your Restaurant or an ownership interest in you), including, without limitation, any person who has a direct or indirect interest in you (or a transferee), the Agreement, the Franchise, or your Restaurant and any person who has any other legal or equitable interest, or the power to vest in himself or herself any legal or equitable interest, in their revenue, profits, rights, or assets. References to an "ownership interest" in you or one of your owners (if you are not a natural person) mean the

percent of the voting shares or other voting rights that results from dividing 100% of the ownership interests by the number of owners. "Person" means any natural person, corporation, limited liability company, general or limited partnership, unincorporated association, cooperative, or other legal or functional entity. Unless otherwise specified, all references to a number of days shall mean calendar days and not business days.

- 10.8. **Business Judgment.** Whenever we reserve discretion in a particular area or where we agree to exercise our rights reasonably or in good faith, we will satisfy such obligations whenever we exercise reasonable business judgment in making our decision or exercising our rights. Our decisions or actions will be deemed to be the result of reasonable business judgment, even if other reasonable or even arguably preferable alternatives are available.
- 10.9. Remedies Cumulative. All rights and remedies of the parties to this Agreement are cumulative and not alternative or in addition to and are not exclusive of any other rights or remedies which are provided for herein or which may be available at law or in equity in case of any breach, failure, or default or threatened breach, failure, or default of any term, provision, or condition of this Agreement or any other agreement between you and any of your Affiliates and us and any of our Affiliates. The rights and remedies of the parties to this Agreement will be continuing and will not be exhausted by any one or more uses thereof and may be exercised at any time or from time to time as often as may be expedient; and any option or election to enforce any such right or remedy may be exercised or taken at any time and from time to time. The expiration, earlier termination, or exercise of our rights pursuant to Article 7 of this Agreement will not discharge or release you or any of your owners from any liability or obligation then accrued, or any liability or obligation continuing beyond, or arising out of, the expiration, the earlier termination, or the exercise of such rights under this Agreement. Additionally, you and your owners will pay all courts costs and reasonable attorneys' fees and costs we incur in obtaining any remedy available to us for any violation of this Agreement.
- 10.10. **No Third-Party Beneficiary**. Except as expressly provided to the contrary herein, nothing in this Agreement is intended, nor will be deemed, to confer rights to or upon any person or legal entity who is not a party to this Agreement.
- 10.11. No Liability for Related Parties. You agree that none of our past, present, or future directors, officers, employees, incorporators, members, partners, stockholders, subsidiaries, Affiliates, controlling parties, entities under common control, ownership, or management, vendors, service providers, agents, attorneys, or representatives will have or will have any liability for: (a) any of our obligations or liabilities relating to or arising from this Agreement; (b) any claim against us based on, in respect of, or by reason of, the relationship between you and us; or (c) any claim against us based on any of our alleged, unlawful acts or omissions.
- 10.12. <u>Counterparts.</u> This Agreement may be executed in one or more counterparts, each of which will be deemed to be an original, but all of which together will constitute one agreement.

11. APPLICABLE LAW; DISPUTES

- 11.1. <u>Choice of Law.</u> Except to the extent governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. Sections 1051 *et seq.*) or other federal law, this Agreement will be governed by and construed in accordance with the laws of the State of Delaware without regard to its conflict of law rules, except that any law regulating the offer or sale of franchises, business opportunities, or similar interests or governing the relationship between you and us will not apply unless its jurisdictional requirements are met independently without reference to this Section.
- 11.2. **Requirement to Submit Disputes to Arbitration.** We and you agree that all controversies, disputes, or claims between us or any of our Affiliates, and our and their respective owners, officers, directors, agents, and employees, on the one hand, and you (and your owners, guarantors, Affiliates, and employees), on the other hand, arising out of or related to:

- (g) this Agreement or any other agreement between you (or any of your owners) and us (or any of our Affiliates);
- (h) our relationship with you;
- (i) the scope or validity of this Agreement or any other agreement between you (or any of your owners) and us (or any of our Affiliates) or any provision of any of such agreements (including the validity and scope of the arbitration provision under this Section 11.3, which we and you acknowledge is to be determined by an arbitrator, not a court); or
- (j) any System Standard, must be submitted for binding arbitration, on demand of either party, to the American Arbitration Association (the "AAA"). The arbitration proceedings will be conducted by one arbitrator and, except as this Section otherwise provides, according to the AAA's then-current Commercial Arbitration Rules. All proceedings will be conducted at a suitable location chosen by the arbitrator that is within 50 miles of our or, as applicable, our successor's or assign's then current principal place of business (currently, Plano, Texas). All matters relating to arbitration will be governed by the Federal Arbitration Act (9 U.S.C. §§ 1 et seq.). The interim and final awards of the arbitrator shall be final and binding upon each party, and judgment upon the arbitrator's awards may be entered in any court of competent jurisdiction.

The arbitrator has the right to award or include in an award any relief which he or she deems proper, including money damages, pre- and post-award interest, interim costs and attorneys' fees, specific performance, and injunctive relief, provided that the arbitrator may not declare any of the trademarks owned by us or our affiliates generic or otherwise invalid, or award any punitive or exemplary damages against any party to the arbitration proceeding (we and you hereby waiving to the fullest extent permitted by law any such right to or claim for any punitive or exemplary damages against any party to the arbitration proceeding). In any arbitration brought pursuant to this Section, and in any action in which a party seeks to enforce compliance with this provision, the prevailing party shall be awarded its costs and expenses, including attorneys' fees, incurred in connection therewith.

We and you agree to be bound by the provisions of any applicable contractual or statutory limitations provision, whichever expires earlier. We and you further agree that, in any arbitration proceeding, each party must submit or file any claim which would constitute a compulsory counterclaim (as defined by Rule 13 of the Federal Rules of Civil Procedure) within the same proceeding. Any claim which is not submitted or filed as required will be forever barred. The arbitrator may not consider any settlement discussions or offers that might have been made by either you or us.

WE AND YOU AGREE THAT ARBITRATION WILL BE CONDUCTED ON AN INDIVIDUAL BASIS AND THAT AN ARBITRATION PROCEEDING BETWEEN US AND ANY OF OUR AFFILIATES, OR OUR AND THEIR RESPECTIVE OWNERS, OFFICERS, DIRECTORS, AGENTS, AND EMPLOYEES, ON THE ONE HAND, AND YOU (OR YOUR OWNERS, GUARANTORS, AFFILIATES, AND EMPLOYEES), ON THE OTHER HAND, MAY NOT BE: (I) CONDUCTED ON A CLASS-WIDE BASIS, (II) COMMENCED, CONDUCTED OR CONSOLIDATED WITH ANY OTHER ARBITRATION PROCEEDING, (III) JOINED WITH ANY SEPARATE CLAIM OF AN UNAFFILIATED THIRD PARTY, OR (IV) BROUGHT ON YOUR BEHALF BY ANY ASSOCIATION OR AGENCY. Notwithstanding the foregoing, if any court or arbitrator determines that all or any part of the preceding sentence is unenforceable with respect to a dispute, controversy or claim that otherwise would be subject to arbitration under this Section, then all parties agree that this arbitration clause shall not apply to that dispute, controversy or claim and that such dispute, controversy or claim shall be resolved in a judicial proceeding in accordance with the dispute resolution provisions of the Agreements.

We and you agree that, in any arbitration arising as described in this Section, the arbitrator shall have full authority to manage any necessary exchange of information among the parties with a view to achieving an efficient and economical resolution of the dispute. The parties may only serve reasonable requests for documents, which must be limited to documents upon which a party intends to rely or documents that are directly relevant and material to a significant disputed issue in the case or to the case's outcome. The document requests shall be restricted in terms of time frame, subject matter and persons or entities to which the requests pertain, and shall not include broad phraseology such as "all documents directly or indirectly related to." You and we further agree that no interrogatories or requests to admit shall be propounded, unless the parties later mutually agree to their use.

The provisions of this Section are intended to benefit and bind certain third-party non-signatories.

The provisions of this Section will continue in full force and effect subsequent to and notwithstanding the expiration or termination of the Agreement.

Any provisions of this Article 11 that pertain to judicial proceedings shall be subject to the agreement to arbitrate contained in this Section.

- 11.3. <u>Nonexclusivity of Remedy.</u> No right or remedy conferred upon or reserved to us or you by this Agreement is intended to be, nor will be deemed, exclusive of any other right or remedy in this Agreement or by law or equity provided or permitted, but each will be cumulative of every other right or remedy.
- 11.4. <u>Waiver of Jury Trial.</u> We and you irrevocably waive trial by jury in any action, proceeding, or counterclaim, whether at law or in equity, brought by either of us against the other, whether or not there are other parties in such action or proceeding.
- 11.5. <u>Waiver of Punitive Damages</u>. The parties hereby waive to the fullest extent permitted by law any right to or claim of any punitive or exemplary damages against the other and agree that in the event of a dispute between them each shall be limited to the recovery of any actual damages sustained by it.
- 11.6. <u>Right to Injunctive Relief.</u> Nothing contained in this Agreement will bar our right to seek injunctive relief from any court of competent jurisdiction; and you agree to pay all costs and reasonable attorneys' fees we incur in obtaining such relief.
- 11.7. <u>Attorneys' Fees.</u> If either party commences a legal action against the other party arising out of or in connection with this Agreement, the prevailing party shall be entitled to have and recover from the other party its reasonable attorneys' fees and costs of suit.
- 11.8. <u>Limitation of Claims.</u> You and your owners agree not to bring any claim asserting that any of the Marks are generic or otherwise invalid. EXCEPT FOR CLAIMS ARISING FROM YOUR NON-PAYMENT OR UNDERPAYMENT OF AMOUNTS YOU OWE US, ANY AND ALL CLAIMS ARISING OUT OF OR RELATING TO THE AGREEMENT OR OUR RELATIONSHIP WITH YOU WILL BE BARRED UNLESS A JUDICIAL OR ARBITRATION PROCEEDING IS COMMENCED IN ACCORDANCE WITH THE AGREEMENT WITHIN ONE (1) YEAR FROM THE DATE ON WHICH THE PARTY ASSERTING THE CLAIM KNEW OR SHOULD HAVE KNOWN OF THE FACTS GIVING RISE TO THE CLAIMS. The parties understand that such time limit might be shorter than otherwise allowed by law. You and your owners agree that your and their sole recourse for claims arising between the parties shall be against us or our successors and assigns. You and your owners agree that our and our Affiliates' members, managers, shareholders, directors, officers, employees, and agents shall not be personally liable nor named as a party in any action between us or our Affiliates and you or your owners. No previous course of dealing shall be admissible to explain, modify, or contradict the terms of the Agreement. No implied covenant of good faith and fair dealing shall be used to alter the express terms of the Agreement.

WE AND YOU AGREE THAT ANY PROCEEDING WILL BE CONDUCTED ON AN INDIVIDUAL BASIS AND THAT ANY PROCEEDING BETWEEN US AND ANY OF OUR AFFILIATES, OR OUR AND THEIR RESPECTIVE SHAREHOLDERS, OFFICERS, DIRECTORS, AGENTS, AND EMPLOYEES, ON THE ONE HAND, AND YOU (OR YOUR OWNERS, GUARANTORS, AFFILIATES, AND EMPLOYEES), ON THE OTHER HAND, MAY NOT BE: (I) CONDUCTED ON A CLASS-WIDE BASIS, (II) COMMENCED, CONDUCTED OR CONSOLIDATED WITH ANY OTHER PROCEEDING, (III) JOINED WITH ANY CLAIM OF AN UNAFFILIATED THIRD-PARTY, OR (IV) BROUGHT ON YOUR BEHALF BY ANY ASSOCIATION OR AGENT.

11.9. <u>Consent to Jurisdiction</u>. Subject to Section 11.2 above and the provisions below, we and you agree that all controversies, disputes, or claims between us or any of our affiliates (and our and their respective shareholders, officers, directors, agents, and employees), on the one hand, and you (and your owners, guarantors, affiliates, and employees), on the other hand, arising out of or related to this Agreement or any other agreement between you (or any of your owners) and us (or any of our affiliates) or our relationship with you must be commenced exclusively in state or federal court closest to our (or our successor's or assign's, as applicable) then-current principal place of business (currently, Plano, Texas), and the parties irrevocably consent to the jurisdiction of those courts and waive any objection to either the jurisdiction of or venue in those courts. Nonetheless, the parties agree that any of us may enforce any arbitration orders and awards in the courts of the state or states in which you are or the Development Area is located.

[Signature Page to Follow]

IN WITNESS WHEREOF, the parties have executed and delivered this Agreement effective on the Effective Date.

RUSTY TA	CO FRANCHISING, LLC	[DEVELOPER]	
By:		By:	
Name:		Name:	
Title:	·	Title:	
Date:		Date:	
	*This is the Effective Date		

RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT

ATTACHMENT A DATA SHEET

1. Effective Date of Agreement:				
2. Developer:				
Name:				
Address:				
Attention:				
Email Address:				
Phone:				
Type of Entity:				
Date of Formation:				
State of Formation:				
3. Developer's Owners [Section 8.2.1]:				
	[Section 6.2.1].			
Name	Residential and Email Address	Type of Interest	Percentage Held	
_		Type of Interest	Percentage Held	
_		Type of Interest	Percentage Held	
_		Type of Interest	Percentage Held	
Name	Residential and Email Address Section 1.1]: \$	Type of Interest	Percentage Held	

6. Development Schedule	[Section 1.1]:	
Opening F	Requirements:	
Development Period	Expiration Date of Development Period	Cumulative Total Number of Restaurants Locate in the Development Area Which Developer Shall Have Open and in Operation
1		
2		
Projected Opening Dates:		
Restaurant	Projected Opening Date	Franchise Agreement Execution Date
amend the provisions of		control in the event of any conflicts: t as applicable]
RUSTY TACO FRANCE	HISING, LLC	[DEVELOPER]
By:		By:
Name:		Name:
Title:		Title:
Date:		Date:

RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT

ATTACHMENT B

GUARANTY AND ASSUMPTION OF OBLIGATIONS

THIS GUARANTY AND ASSUMPTION OF OBLIGATIONS is given by each of the undersigned persons indicated below who have executed this Guaranty (each a "Guarantor") to be effective as of the Effective Date of the Agreement (defined below).

Each Guarantor consents and agrees that: (1) Guarantor's direct and immediate liability under this Guaranty will be joint and several, both with Developer and among other guarantors; (2) Guarantor will render any payment or performance required under the Agreement upon demand if Developer fails or refuses punctually to do so; (3) this liability will not be contingent or conditioned upon our pursuit of any remedies against Developer or any other person; (4) this liability will not be diminished, relieved, or otherwise affected by any extension of time, credit, or other indulgence which we may from time-to-time grant to Developer or to any other person, including, without limitation, the acceptance of any partial payment or performance or the compromise or release of any claims, none of which will in any way modify or amend this Guaranty, which will be continuing and irrevocable during the term of the Agreement; and (5) at our request, each Guarantor shall present updated financial information to us as reasonably necessary to demonstrate such Guarantor's ability to satisfy the financial obligations of Developer under the Agreement.

Each Guarantor waives: (i) all rights to payments and claims for reimbursement or subrogation which any Guarantor may have against Developer arising as a result of the Guarantor's execution of and performance under this Guaranty; and (ii) acceptance and notice of acceptance by us of Guarantor's undertakings under this Guaranty, notice of demand for payment of any indebtedness or non-performance of any obligations hereby guaranteed, protest and notice of default to any party with respect to the indebtedness or nonperformance of any obligations hereby guaranteed, and any other notices to which he or she may be entitled.

Each Guarantor represents and warrants that, if no signature appears below for such Guarantor's spouse, such Guarantor is either not married or, if married, is a resident of a state which does not require the consent of both spouses to encumber the assets of a marital estate.

The provisions contained in Section 11 (Applicable Law; Disputes) of the Agreement, are incorporated into this Guaranty by reference and shall govern this Guaranty and any disputes between the Guarantors and us. The Guarantors shall reimburse us for all costs and expenses we incur in connection with enforcing the terms of this Guaranty.

By signing below, the undersigned spouse of each Guarantor indicated below, acknowledges and consents to the guaranty given herein by his/her spouse. Such consent also serves to bind the assets of the marital estate to Guarantor's performance of this Guaranty. We confirm that a spouse who signs this Guaranty solely in his or her capacity as a spouse (and not as an owner) is signing merely to acknowledge and consent to the execution of the Guaranty by his or her spouse and to bind the assets of the marital estate as described therein and for no other purpose (including, without limitation, to bind the spouse's own separate property).

Each Guarantor that is an entity, retirement or investment account, or trust acknowledges and agrees that if Franchisee is delinquent in payment of any amounts guaranteed hereunder, that no dividends or distributions may be made by such Guarantor (or on such Guarantor's account) to its owners, accountholders or beneficiaries or otherwise, for so long as such delinquency exists, subject to applicable law

This Guaranty is binding upon each Guarantor and its respective executors, administrators, heirs, beneficiaries, and successors in interest.

[Signature page follows]

IN WITNESS WHEREOF, each of the undersigned has affixed his or her signature on the same day and year as this Guaranty and Assumption of Obligations was executed.

GUARANTOR(S)	SPOUSE(S)
#1:	#1:
Signature:	Signature:
Name:	Name:
Address:	Address:
Email:	Email:
#2:	#2:
Signature:	Signature:
Name:	Name:
Address:	Address:
Email:	Email:
#3:	#3:
Signature:	Signature:
Name:	Name:
Address:	Address:
Email:	Email:

EXHIBIT C

FRANCHISE AGREEMENT



RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT

RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT

TABLE OF CONTENTS

1.	GRA]	NT	1		
	1.1.	Grant of Franchise			
	1.2.	Protected Territory.			
	1.3.	Reservation of Rights.			
2.		EWAL			
	2.1.	Renewal	2		
2	CITE	GEL ECTION CONGEDITOTION DEGENERAL DANIELO CATION	_		
3.		SELECTION, CONSTRUCTION; RESTAURANT LOCATION	2		
	3.1.	Site Selection and On-Site Evaluation.			
	3.2.	Approval of Location.			
	3.3.	Lease of Building.			
	3.4.	Restaurant Design and Build-Out.			
	3.5.	Opening.			
	3.6.	Liquidity and Financing.			
	3.7.	Relocation	4		
4	PPPC				
4.	FEES 4.1.	Initial Pagashina Cas			
	4.1. 4.2.	Initial Franchise Fee.			
		Royalty Fees.			
	4.3.	Unauthorized Temporary Closure.			
	4.4.	Non-Compliance Charge.			
	4.5.	Other Payments.			
	4.6.	Set-Off Rights.			
	4.7.	Accounting Period			
	4.8.	Payment Terms and Procedures.			
	4.9.	Electronic Funds Transfer			
	4.10.	Interest; Nonsufficient Funds Charge			
	4.11. 4.12.	Partial Payments; Application of Payments			
	4.12.	Laws Proffibiting Payments on Alcohol Sales.			
5.	TRAI	NING AND ASSISTANCE	<i>6</i>		
	5.1.	Initial Training Program.	6		
	5.2.	Restaurant Opening Assistance			
	5.3.	Ongoing Consultation.	7		
	5.4.	Additional Training Programs.	7		
6.	ODE	RATION OF THE RESTAURANT	C		
υ.	6.1.	General Operating Requirements.			
	6.2.	Operating Principal; Restaurant Manager			
	6.3.	Employee Policy; Uniforms and Employee Appearance			
	6.4.				
	6.5.	Authorized Menu Offerings Purchases from Designated Sources			
	6.5. 6.6.	The Restaurant Facilities; Vehicles.			
	6.6. 6.7.				
	6.7. 6.8.	Days and Hours of Operation. Inspections; Testing.			
	6.9.	Pricing			
	6.10.	Technology Fees.			
	6.11.	Contact Identifiers and Listings.	1 2		

	6.12.	Data Privacy and Information Security.	12
7.	MAD	KS, STANDARDS AND COPYRIGHTS	13
1.	7.1.	Acknowledgments	
	7.1.	Limitations on Your Use of the Marks.	
	7.2.	Infringement.	
	7.3. 7.4.		
	7.4. 7.5.	Changes to and Discontinuance of Marks.	
	1.3.	Non-Disparagement.	14
8.	SYST	EM, OPERATIONS MANUAL, AND CONFIDENTIAL INFORMATION	14
	8.1.	Operations Manual	14
	8.2.	System Modification.	14
	8.3.	Confidentiality	14
	8.4.	Innovations.	15
9.	ADVI	ERTISING AND MARKETING	15
<i>)</i> .	9.1.	General; Marketing Cap.	
	9.2.	Grand Opening Marketing Campaign.	16
	9.2. 9.3.	Brand Development Fund.	
	9.3. 9.4.	Local Advertising	
	9. 4 . 9.5.	Franchisee Advertising Cooperatives.	
	9.5. 9.6.	Loyalty Programs, Prize Promotions, Meal Deals, and Promotional Literature	
	9.0. 9.7.	Participation in Marketing Programs	
	9.7. 9.8.	Business Directory Listings.	
	9.8. 9.9.	Websites and Online Presences.	
	9.9.	weosites and Online Presences.	10
10.	COM	PUTER SYSTEM; ACCOUNTING AND RECORDS; TAXES	
	10.1.	Computer System.	
	10.2.	Software.	
	10.3.	Independent Access	20
	10.4.	Maintenance of Records.	
	10.5.	Submission of Financial Statements and Tax Returns.	20
	10.6.	Submission of Performance Reports.	
	10.7.	Our Right to Inspect Your Restaurant.	21
	10.8.	Audit of Franchisee Records	21
	10.9.	Disclosure of Information.	21
	10.10.	Taxes.	21
11.	INDE	PENDENT CONTRACTOR, INSURANCE, AND INDEMNIFICATION	21
11.	11.1.	Independent Contractor.	
	11.2.	Insurance Obligations.	
	11.3.	Indemnification.	
12.	TDAN	ICEED OF INTERECT	22
12.		ISFER OF INTEREST	
	12.1.	Transfer by Us.	
	12.2.	Transfer By You or Your Owners	
	12.3.	Transfers Void.	
	12.4.	Security Interest.	
	12.5.	Private or Public Offerings.	
	12.6.	Right of First Refusal.	
	12.7.	Transfer Upon Death or Incapacity.	
	12.8.	Non-Waiver of Claims.	25
13.	DEFA	ULT AND TERMINATION	26

	13.1.	Termination In the Event of Bankruptcy, Insolvency or Transfers Required on Death of			
	13.2.	Termination With Notice and Without Opportunity To Cure.			
	13.3.	Termination With 10-Day Cure Period.			
	13.4.	Termination With 30-Day Cure Period.			
	13.5.	Cross-Default			
	13.6.	Additional Remedies.	27		
14.	OBLIG	GATIONS UPON TERMINATION OR EXPIRATION	27		
	14.1.	Cease Use of Intellectual Property.	27		
	14.2.	Assignment of Lease; De-Identification.	27		
	14.3.	Return of Operations Manual and Other Confidential Information	27		
	14.4.	Pay Accrued Amounts	28		
	14.5.	Our Right to Purchase Tangible Assets.	28		
	14.6.	Trademark Infringement	28		
	14.7.	Lost Revenue Damages.			
15.	COVE	ENANTS	29		
	15.1.	Noncompetition During Term of Agreement.			
	15.2.	Noncompetition After Expiration or Termination of Agreement			
	15.3.	Additional Provisions.			
	15.4.	Covenants from Individuals.			
	15.5.	Breach of Covenants Causes Irreparable Injury.			
	15.6.	Exception for Publicly Held Companies.			
16.	DEDD	REPRESENTATIONS			
10.	16.1.	Our Representations.			
	16.2.	Representations of Franchisee.			
17.	NOTIO	NOTICES			
1 / .	17.1.	Notices.			
	17.1.	Nouces.			
18.		TRUCTION OF AGREEMENT			
	18.1.	Entire Agreement.			
	18.2.	No Waiver			
	18.3.	Severability			
	18.4.	Survival of Terms.			
	18.5.	Definitions and Captions.			
	18.6.	Persons Bound.			
	18.7.	Rules of Construction.			
	18.8.	Business Judgment.	32		
	18.9.	Counterparts.	32		
	18.10.	Remedies Cumulative	32		
	18.11.	No Third-Party Beneficiary.	32		
	18.12.	No Liability for Related Parties.	32		
19.	APPL	APPLICABLE LAWS; DISPUTES; ARBITRATION			
	19.1.	Choice of Law	33		
	19.2.	Requirement to Submit Disputes to Arbitration.			
	19.3.	Non-Exclusivity of Remedy.			
	19.4.	Waiver of Jury Trial.			
	19.5.	Waiver of Punitive Damages.			
	19.6.	Right to Injunctive Relief.			
	19.7.	Attorneys' Fees.			
	19.8.	Limitation of Claims & Class Action Bar.			

19.9. Consent to Jurisdiction......35

Attachments

Attachment A Data Sheet

Additional Definitions Attachment B

Guaranty and Personal Undertaking Lease Addendum Attachment C

Attachment D

RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT

THIS FRANCHISE AGREEMENT ("**Agreement**") is made effective as of the Effective Date by and between Rusty Taco Franchising, LLC, a Delaware limited liability company with its principal office at 5412 W. Plano Pkwy., Suite 100, Plano, Texas 75093 ("**us**"), and the person or entity who signed this Agreement as Franchisee, as shown on the signature page below ("**you**"). The "Effective Date" is the date we sign this Agreement, as shown on the signature page below.

RECITALS

- A. We grant Franchises for the right to own and operate restaurants that are currently identified by and use the federally registered trademark Rusty Taco[®] and other Marks that we designate from time to time, that are required to operate in accordance with our Standards, and that offer their customers authentic freshly prepared food made with quality, fresh ingredients inspired by the flavors of Mexico and Texas, along with an assortment of alcoholic and non-alcoholic beverages (each a "**Rusty Taco Restaurant**").
- B. Based on your own investigation and diligence, you have requested that we grant you a Franchise and, to support your request, you and, as applicable, your owners have provided us with certain Application Materials. In reliance on, among other things, your Application Materials, we are willing to grant you the Franchise on the terms and conditions contained in this Agreement.
- C. Certain terms are defined where used in this Agreement and others are defined on <u>Attachment B</u> (Additional Terms) hereto.

AGREEMENT

In consideration of the foregoing Recitals (which are incorporated herein), the promises contained in this Agreement, and other valuable consideration, the receipt and sufficiency of which are acknowledged, the parties agree as follows:

1. **GRANT**

- 1.1. Grant of Franchise. We grant you the Franchise, during the Term, to develop, own, and operate a Rusty Taco Restaurant solely at the location and premises identified (or to be identified) in Attachment A (the "Location") and in strict compliance with our Standards and this Agreement. The Rusty Taco Restaurant that you will develop and operate pursuant to this Agreement is hereafter referred to as "your Restaurant" or "the Restaurant". You agree to conduct the business of your Restaurant only at the Location and within the catering and delivery areas that, in our discretion, we may designate from time to time, and to use the Location only for your Restaurant. Once your Restaurant opens for business, you agree to continuously operate it in accordance with this Agreement for the duration of the Term.
- 1.2. **Protected Territory.** Subject to the reservation of rights described in Section 1.3 below and your compliance with this Agreement, we will not operate or authorize any other party to operate, during the Term, a Rusty Taco Restaurant physically located within the Protected Territory.
- 1.3. Reservation of Rights. You agree that this Agreement does <u>not</u> grant you the right, and you agree not, to co-brand with any other concept, to sublicense any part of the rights granted to you under this Agreement, to delegate performance of any obligations you undertake in this Agreement, to ship Rusty Taco[®] branded products, regardless of the destination, or to distribute Rusty Taco[®] branded products through wholesale channels, or retailers such as supermarkets, convenience stores, or other retailers, or through food service providers. You expressly agree that we and our Affiliates retain, without

compensation to you, all such rights and all other rights that are not expressly granted to you in this Agreement and that we do not expressly agree to refrain from doing, including:

- 1.3.1. the right to develop, establish, and operate anywhere (including in your Protected Territory), and the right to grant to others the right to operate anywhere (including in your Protected Territory), other businesses using the Marks or other names or marks
- 1.3.2. the right to, and to grant to others the right to, solicit customers and advertise and promote the Rusty Taco franchise system inside and outside your Protected Territory;
- 1.3.3. the right to acquire, be acquired by, merge, affiliate with, or engage in any transaction with other businesses (whether competitive or not) located anywhere and, even if such businesses are located in your Protected Territory, convert the other businesses to Rusty Taco Restaurants
- 1.3.4. the right to operate, and grant license others to operate, Rusty Taco® Restaurants in any Non-Traditional Location (defined below), including within the Protected Territory; and
- 1.3.5. the right to engage, directly or indirectly, at wholesale, retail or otherwise, in the production, distribution, license, and sale of any and all food, beverages, or other services and products, under the Marks, or under other names or marks, within and outside the Protected Territory, through any method of distribution, including, but not limited to, mail order catalogs, the Internet, and other permanent, temporary or seasonal food service facilities, carts, food trucks, or kiosks (which facilities may provide, in whole or in part, the products and services offered by a Rusty Taco Restaurant) regardless of the proximity to, or the competitive impact on, your operations in the Protected Territory.

2. **RENEWAL**

2.1. **Renewal.** You may renew the Franchise for one (1) continuous 10-year term if, at the end of the Term, each of the following conditions has been satisfied: (a) you have notified us of your desire to renew the Franchise no less than 12 months and no more than 24 months before the end of the Term; (b) you are not in default of any material provision of this Agreement and you have complied with the material terms and conditions of this Agreement throughout the Term; (c) all amounts owed to us and our Affiliates and third-party suppliers to your Restaurant have been paid; (d) the Restaurant has been renovated and refurbished so that it reflects all then-current Standards; (e) you have the right to remain in possession of the Location, or have secured an alternate site with our prior written approval; (f) you comply with our then-current qualifications and training requirements; (g) you sign our then-current form of Franchise Agreement, the terms of which may be materially different than the terms of this Agreement, including but not limited to royalty and advertising fees and contribution requirements, and each Owner executes a personal guaranty and undertaking in the form we prescribe; (h) you and each Owner sign a general and full release in favor of us, and our Affiliates and our and their respective, officers, directors, owners, managers, employees, and agents, of all claims you and your Owners may then have; and (i) you pay a renewal fee, in an amount equal to the higher of \$7,500 or 25% of the initial franchise fee we are then charging for the purchase of a Franchise. Additional renewals will be at our sole discretion.

3. SITE SELECTION, CONSTRUCTION; RESTAURANT LOCATION

3.1. <u>Site Selection and On-Site Evaluation.</u> Despite any assistance or guidance we provide, you are solely responsible for locating a proposed site and securing possession of the site once we have approved it. Within 180 days after the Effective Date, you must have (a) identified a proposed Location for your Restaurant, (b) submitted to us a site approval request packet, as set forth in Section 3.2 below, and (c) upon our approval of the Location, acquired possession of the Location for your Restaurant. The Location

must be situated within the Site Selection Area identified in <u>Attachment A</u> and must otherwise be mutually acceptable to you and us. We will conduct one, on-site, site selection evaluation visit at no cost to you.

- 3.2. Approval of Location. For each proposed site that you identify, you must deliver to us a completed site approval request packet, on our approved forms, containing all information we require regarding the proposed site. You may not proceed with any site until we approve it, in writing, to be the Location on which your Restaurant may be developed and operated. Our approval of a proposed site is merely an indication that the site meets our then-current criteria that we have established for our own purposes. You may not rely on it for any purposes, including as an assurance that you will be able to secure possession of the site, that local authorities will allow you to develop the Restaurant in compliance with our Standards, that the Restaurant can be developed within any particular time period, that there will not be extraordinary costs associated with the development of the Restaurant at the Location, or that the Restaurant will achieve a certain sales volume or level of profitability.
- 3.3. Lease of Building. If you intend to occupy the Location under a lease with a third-party landlord, you may not sign the lease until we have approved its terms. At a minimum, the lease must contain the terms we periodically specify from time to time to, in our judgment, protect the Rusty Taco® brand and our interests as franchisor of the brand, including our option to assume the lease in the event of expiration or termination of this Agreement or your default of the lease. Our current lease requirements are reflected in Attachment D and are subject to change. You may not rely on our approval of your lease for any purposes, including as to the favorability of its terms or your rights under the lease. You must deliver to us a fully executed copy of the lease within 10 days after its execution.
- 3.4. Restaurant Design and Build-Out. You must follow our procedures and Standards for the construction and build-out of your Restaurant, which includes engaging the design consultant and/or general contractor that we designate or otherwise approve. You must engage and pay any fees and costs associated with the architect we designate or otherwise approve. We will provide you access to specifications and layouts for the Restaurants that we approve. You must, working with your architect and contractor, do all things necessary to construct and build out the Restaurant according to our Standards for design, decor, and layout, including (a) equipping your Restaurant according to our requirements for fixtures, furnishings, equipment, interior and exterior signage, artwork and graphics, trade dress, and awnings, (b) sourcing, hiring, and paying for the professional services needed to complete your Restaurant design and build-out obligations, (c) obtaining all government approvals, zoning classifications, permits, and clearances related to your Restaurant and for complying with all Applicable Laws including those related to accessibility, and (d) securing and maintaining all insurance coverages necessary to comply with Applicable Laws and our requirements. You must notify us in writing when construction begins and thereafter provide a weekly progress report. We and our designees have the right to inspect the site at all reasonable times and conduct such number of on-site visits of the Restaurant during the construction phase that we deem appropriate, which may be only one visit.
- 3.5. Opening. You must, at your expense, do all things necessary to commence the operations of the Restaurant by the Opening Deadline in accordance with our Standards. You must not open the Restaurant until you receive our written permission to do so, which will be granted only if: (a) all amounts due to us and our Affiliates, (b) the Restaurant has been constructed and equipped according to our Standards, (c) all of your pre-opening and training obligations have been satisfied, (d) we have received from you a signed ACH Authorization on our then-current form; (e) we have received from you a fully executed copy of your approved Restaurant lease; (f) we have received from you certificates of insurance as required by this Agreement and Applicable Laws; and (g) you are otherwise in good standing under this Agreement. You must open the Restaurant within five (5) days following our written approval. No later than 120 days after the Opening Date, you must provide us a report, in the format and containing the information that we reasonably specify, identifying the amounts that you spent in various categories relating to the development and opening of the Restaurant.

- 3.6. Liquidity and Financing. We have granted the Franchise to you based, in part, on your representations to us regarding, and our assessment of, your liquidity as of the Effective Date. You will ensure that, throughout the Term, you will maintain, to our reasonable satisfaction, sufficient liquidity to meet your obligations under this Agreement. You acknowledge and agree that, if at any time you or your Affiliates propose to obtain any financing with respect to the Location, the development or operation of your Restaurant, or furniture, fixtures, equipment or other assets used in the operation of the Restaurant in which such items are pledged as collateral to secure your performance in connection with such financing (each a "Financing"), the Financing is a Transfer subject to our approval and all other conditions set forth in Sections 12.4 and 12.6 of this Agreement. Our consent to any Financing may be conditioned upon the inclusion in the loan and security agreements of various terms and conditions that are acceptable to us. Subject to this Section, and unless we approve otherwise, in writing, you must apply for and diligently pursue any government-issued, government-sponsored, or governmental-guaranteed non-recourse grants, loans, and/or bailouts for which you qualify and that are made available to small businesses for economic stimulus.
- 3.7. **Relocation.** You may not relocate the Restaurant without our prior written consent, which we may grant, withhold or condition in our sole discretion. Subject to our approval of your proposed relocation site and your proposed relocation plan, we will not withhold our approval of your request to relocate if your lease expires or terminates through no fault of yours, or if the Restaurant is destroyed or materially damaged by fire, flood, or other natural catastrophe, and you are not in default of this Agreement or any other agreement between you and us. In connection with the relocation, you must sign a new, then-current form of Franchise Agreement, and a general release by you and your related parties of all claims against us and our related parties. Site selection, construction, and opening of the relocated Restaurant will be governed by the new franchise agreement. You are solely responsible for all relocation costs and expenses, which may include reimbursement of our reasonable costs and expenses (including attorneys' fees) associated with the relocation.

4. **FEES**

- 4.1. <u>Initial Franchise Fee.</u> Upon execution of this Agreement, you shall pay us an Initial Franchise Fee in the amount specified in <u>Attachment A</u> (less, if applicable, any credit required under a Development Agreement). The Initial Franchise Fee is fully earned by us when paid and is not refundable.
- 4.2. **Royalty Fees.** You also agree to pay us a nonrefundable, continuing and recurring royalty fee in an amount equal to five percent (5%) of the Gross Sales of the Restaurant for each Accounting Period ("**Royalty Fee**"). We will determine the amount of the Royalty Fee and other amounts due under this Agreement by accessing and retrieving Gross Sales data from your computer system, as permitted by Article 10, or any other means available to us to determine your Restaurant's Gross Sales for the applicable Accounting Period. If we are unable to access your computer system to determine the amount of the Royalty Fee, we may debit your account for 110% of the average of the last three (3) Royalty Fee and Fund contributions that we debited. If the amounts that we debit from your account under this paragraph are less than the amounts you actually owe us (once we have determined the true and correct Gross Sales), we will debit your account for the balance on the day we specify. If such amounts are greater than what you actually owe, we will credit the excess against the amounts we otherwise would debit from your account during the following week.
- 4.3. <u>Unauthorized Temporary Closure.</u> If your Restaurant does not generate Gross Sales because you have temporarily closed your Restaurant without our consent, your Royalty Fee for each whole or partial day your Restaurant is closed ("**Temporary Royalty**") will be 110% of the average daily amount of the Royalty Fee that was owed for the week immediately prior to the closure. Temporary Royalty will be due and payable as, when and in the same manner as the Royalty Fee. Payment of a Temporary Royalty will not act as a cure of the default caused by the unauthorized closure and will not alter or impair any other

rights we have under this Agreement or any other agreements between us and you or your Affiliates, all of which are reserved. If we do not exercise our right to terminate based on your unauthorized closure, you may only re-open the Restaurant with our prior written consent, and your Royalty Fee will resume as described above in Section 4.2 upon reopening.

- 4.4. Non-Compliance Charge. The Royalty Fee rate we charge under this Agreement was determined based on the assumption that you will comply with your obligations hereunder. If you do not comply with your obligations, we will incur additional costs and expenses and, depending on the nature of the noncompliance, may lose the revenue stream we bargained for in entering into this Agreement. Therefore, if we determine that you are not in compliance with your obligations under this Agreement, your Royalty Fee rate will be increased to six percent (6%) until we determine that you have cured all deficiencies and are compliant with all terms of the Agreement, at which time it will revert to the rate referenced in Section 4.2 above. Payment of the non-compliance charge is intended to compensate us for certain revenue we may have lost and expenses or losses we incur as a result of the non-compliance. It is not a penalty or an expression of the total amount of such damages nor is it a cure of the non-compliance that triggered its payment. Nothing in this Section limits any of our other rights and remedies available under the terms of this Agreement.
- 4.5. Other Payments. In addition to all other payments provided in this Agreement, you shall pay us and our Affiliates promptly when due:
 - 4.5.1. all amounts we or our Affiliates advance for your benefit or for which we or our Affiliates have become obligated to pay on your behalf for any reason whatsoever; and
 - 4.5.2. the amount of all sales taxes, use taxes, personal property taxes, and similar taxes, which are or may be imposed upon you and required to be collected or paid by us (except our own income taxes) on account of your Gross Sales or fees you pay or owe to us under this Agreement. We may, in our discretion, collect the taxes in the same manner as Royalty Fees are collected and promptly pay the tax collections to the appropriate governmental authority; provided, however, that unless we so elect, it shall be your responsibility to pay all such taxes now or hereinafter imposed by any governmental authorities;
 - 4.5.3. amounts due relating to your participation in marketing programs pursuant to Sections 9.5, 9.6, 9.7, and 9.8 of this Agreement; and
 - 4.5.4. all amounts due to us and our Affiliates, all Designated Suppliers, Designated Distributors, and your other trade creditors and your landlord (unless such amount is subject to a bona fide dispute).
- 4.6. <u>Set-Off Rights.</u> You may not set off, deduct, or otherwise withhold any fees or other amounts due us under this Agreement on grounds of our alleged nonperformance of any of our obligations or for any other reason. We may set off against any amounts we owe you any amounts you owe us and our Affiliates.
- 4.7. Accounting Period. We have the right to define applicable accounting periods for purposes of calculating and paying amounts due under this Agreement. Each period, which may be defined in terms of days, weeks, or months, in our sole discretion, will be considered an "Accounting Period" for all purposes under this Agreement. We have the right to change or modify the definition of an Accounting Period, in our discretion, for the entire System generally, or for you, individually, if you do not comply with any of your obligations under this Agreement. We will provide you with at least 30 days' advance written notice of any change in Accounting Period.
- 4.8. **Payment Terms and Procedures.** All payments required by this Agreement must be paid on the day or date we specify ("**Due Date**"). If the Due Date is not a Business Day, then payment will be due on the next Business Day.

- 4.9. Electronic Funds Transfer. You shall participate in our electronic funds transfer program, as it may be changed from time to time, authorizing us to use a pre-authorized bank draft system. You shall: (a) comply with our procedures, as specified in the Operations Manual or otherwise in writing; (b) perform those acts and sign and deliver those documents as may be necessary to accomplish payment by electronic funds transfer as described in this Section 4.9; (c) give us an authorization in the form we designate from time to time to initiate debit entries and/or credit correction entries to an appropriately funded bank account for payments of the Royalty Fee and all other amounts payable under this Agreement, including any interest charges; and (d) make sufficient funds available in your designated bank account for withdrawal by electronic funds transfer no later than the Due Date for payment thereof. Notwithstanding the provisions of this Article 4., we reserve the right to modify, at our option, the method by which you pay the Royalty Fee and other amounts owed under this Agreement upon our written notice. Your failure to have sufficient funds in the account for which you have given us authorization to make electronic funds transfers is a material breach of this Agreement.
- 4.10. <u>Interest; Nonsufficient Funds Charge.</u> Any payments we do not receive by the Due Date will accrue interest at the rate of 18% per annum or the highest lawful interest rate permitted by the jurisdiction in which the Restaurant operates, whichever is less. If any check or draft, electronic or otherwise, is returned for insufficient funds, you shall pay us a nonsufficient funds charge for each check or draft in the amount of \$50 and reimburse us for all expenses that we incur on account of such nonsufficient funds. Payment of interest will not constitute a cure of the non-payment.
- 4.11. **Partial Payments; Application of Payments.** If you pay less than the amount due, your payment will be considered a partial payment on account. We may accept such payment as a partial payment, irrespective of any endorsement or other statement that the payment constitutes full payment. Our acceptance of such partial payment will not be considered a waiver of our right to demand or receive full payment, and you hereby waive any estoppel defense in this regard. We may apply your payments to any indebtedness, in our sole and reasonable discretion, regardless of any designation that accompanies the payment.
- 4.12. Laws Prohibiting Payments on Alcohol Sales. If a law is in effect on the Effective Date or is later enacted and takes effect during the Term of this Agreement that prohibits or restricts in any way your ability to pay and/or our ability to collect the Royalty Fee or other amounts derived from the sale of alcohol products, then we reserve the right to modify your payment obligations under this Agreement and amend the applicable provisions hereunder in order to provide the same economic effect to both you and us contemplated by this Agreement as if such law had not taken effect (including, for example, by increasing the Royalty Fee and other payments owed on non-alcohol sales). In such event, you agree to execute the documents in the form we require to give effect to such changes to this Agreement.

5. TRAINING AND ASSISTANCE

5.1. <u>Initial Training Program.</u> If this is your or your Affiliates' first Rusty Taco Restaurant, then (i) at least 60 days before the Opening Date of the Restaurant, at least one (1) and up to three (3) people (one of which must be your Operating Principal or Restaurant Manager, if applicable) (the "Required Trainees") must attend and complete to our satisfaction, our initial training program, and (ii) we will not assess a fee for the initial Required Trainees to attend the initial training program. Replacements of your Operating Principal or Restaurant Manager will be allowed to participate in our initial training program subject to compliance with the requirements regarding Additional Training in Section 5.4 below. If the Restaurant is not your or your Affiliates' first Restaurant, we may require you to (i) conduct the initial training program for your Required Trainees who have not previously completed the program, but we reserve the right to conduct it ourselves at our designated training location if we determine that you are unable to properly do so, and (ii) pay us a non-refundable initial training fee of \$8,000 for providing initial training to your Required Trainees. You will be responsible for all travel and living expenses and related costs that

your personnel incur while attending any training programs.

- Affiliates' first Rusty Taco® Restaurant, we will provide on-site Restaurant opening assistance to you; however, the number of personnel we send and the duration of on-site Restaurant opening assistance will be in our discretion. That assistance will be provided at no cost to you, except that you must pay for the costs of your employees and for the food that is used during the opening assistance period. Your Operating Principal or Restaurant Manager must be present at the Restaurant during the opening assistance period. If this is not your or your Affiliates' first Rusty Taco® Restaurant, we may, either at your request or if we deem necessary, provide on-site opening assistance, subject to availability of our personnel. In such event, you agree to pay our then-current per diem fee (which will not exceed \$400 per trainer per day), plus reimburse us for the out-of-pocket costs for travel, lodging, and dining that are incurred by our personnel. We will provide the on-site, opening assistance once you have completed all improvements at the Restaurant, received a certificate of occupancy, applied for your liquor license, and hired enough employees to staff the Restaurant.
- 5.3. Ongoing Consultation. We will provide such ongoing consultation and advice as we deem appropriate, which may include information about new product development, instruction concerning the operation and management of a Rusty Taco® Restaurant, advertising and marketing advice, and financial and accounting advice. We will determine the substance, manner, and frequency of any such consultation and advice.
- 5.4. Additional Training Programs. We may from time to time, either on your request or at our initiative, provide remedial training if, in our determination, your Restaurant is not complying with our Standards because your personnel have been improperly or inadequately trained ("Remedial Training"). Your Required Trainees and other personnel we designate must attend and complete, at your expense, all Remedial Training. We may also, in our discretion and during the Term, provide or make available, and all of your Required Trainees and other personnel we designate must attend and complete, additional courses, seminars, meetings (including our annual meeting of franchisees), and other training programs regarding the operation of Rusty Taco Restaurants (together with the Remedial Training, the "Additional Training Programs").

We reserve the right to require that you pay us a reasonable fee (which will not exceed \$400 per trainer per day) for any Additional Training Programs and reimburse our expenses, including the costs of travel, lodging, and meals, related to any Additional Training Programs or initial training for your replacement Required Trainees if such training is provided at your Restaurant. We will not be responsible for, and you agree to hold us harmless from, the costs of any subscriptions or licenses required to participation in any training program and any travel or living expenses, wages, or benefits owed to, or other costs of, any person attending the initial training program, any Additional Training Programs, any replacement training and any other training program, whether participation in such training is mandatory or voluntary.

All initial training programs we provide or designate will take place at a location (which may be virtual) and at the times that we designate. The timing of completion, scheduling, location, content, length and format of our training programs are at our discretion, and we reserve the right to require that all of your required attendees attend and participate at the same time.

6. **OPERATION OF THE RESTAURANT**

- 6.1. General Operating Requirements. You agree to faithfully, honestly, and diligently perform your obligations under this Agreement and to operate and promote your Restaurant, generally, in accordance with this Agreement, with all Applicable Laws, and in a manner that promotes and enhances the goodwill associated with your Restaurant, the Marks used at your Restaurant, and the Rusty Taco® brand at your Location throughout the Term. You agree to comply with our Standards and acknowledge and agree that we may add to, eliminate, or modify any the Standards applicable to the operation of Rusty Taco Restaurants in our sole discretion, including modifications to accommodate local and regional variations, each of which may require that you make certain modifications to the manner in which your Restaurant is operated. Specifically, you agree, at your expense, to do the following:
 - 6.1.1. operate the Restaurant according to the highest applicable health standards and ratings; timely obtain any and all permits, certificates, or licenses necessary for the lawful operation of the Restaurant; operate the Restaurant according to our then-current Standards and mandatory provisions of the Operations Manual; and maintain, at all times, a high moral and ethical standard in the operation of the Restaurant;
 - 6.1.2. accept in the Restaurant all forms and methods of payment (including procedures for collection of funds relating thereto) that we specify from time to time, and acquire, install and use all hardware and software necessary to do so;
 - 6.1.3. comply with and meet or exceed all industry standards applicable to the protection of customer privacy and credit card information, including standards developed by the Payment Card Industry Data Security Standards (PCI DSS) council or its successor and other regulations and industry standards, the Fair and Accurate Credit Transaction Act (FACTA), and all other data security requirements we prescribe;
 - 6.1.4. notify us immediately by telephone, and confirm in writing within 48 hours of such notice, of any investigation or violation, actual or alleged, concerning any health, liquor, or narcotics laws or regulations or concerning any data breaches or data security threats, and notify us in writing within 72 hours of the commencement of any investigation, action, suit, or proceeding or the issuance of any order, writ, injunction, award, or decree of any court, agency, or other government instrumentality, which may adversely affect the operation or financial condition of the Restaurant;
 - 6.1.5. upon the occurrence of a Crisis Management Event, immediately inform our President (or as otherwise instructed in the Operations Manual) by telephone, or as set forth in this Agreement, of such event and cooperate fully with us and with the appropriate authorities with respect to the response to, investigation, mitigation and resolution of, and public statements and communications regarding the Crisis Management Event;
 - 6.1.6. process and handle all customer complaints connected with or relating to the Restaurant and promptly notify us of all: (a) food related illnesses; (b) safety or health violations; (c) claims exceeding \$1,000; and (d) any other material claims against or losses suffered by Restaurant. You shall maintain any communications with governmental authorities affecting the Restaurant during the Term of this Agreement and for one year after the expiration or earlier termination hereof. If any customer of the Restaurant contacts us to report a complaint about your Restaurant, you agree that we may, in our discretion, compensate the customer in such manner as we determine appropriate, and you agree to reimburse us the amount of such compensation upon our demand;
 - 6.1.7. attend, and cause your Operating Principal to attend, at your cost, our annual meeting of franchisees:

- 6.1.8. purchase and use all operations-related programs or materials we require, in our sole discretion, and if such programs or materials are supplied by third-party vendors, pay all associated costs and fees; and
- 6.1.9. comply with all Applicable Laws pertaining to sending emails, including the Controlling the Assault of Non-Solicited Pornography and Market Act of 2003 (CAN-SPAM Act).

6.2. Operating Principal; Restaurant Manager.

- 6.2.1. If the Restaurant is your or your Affiliates' only Rusty Taco Restaurant, your Operating Principal must engage in and have control over the on-premises day-to-day supervision of the Restaurant. The Operating Principal will be (i) you if you are a natural person, (ii) your general partner (or an owner of your general partner) if you are a partnership, (iii) a natural person having at least a 10% direct or indirect ownership interest in you if you are any other kind of legal entity. If you are not a natural person, your Operating Principal must be approved by us. Unless a Restaurant Manager is appointed, as described below, the Operating Principal shall devote full-time efforts to the operation of the Restaurant and shall not engage in any other business or activity, directly or indirectly, that requires substantial management responsibility or time commitment. If the Operating Principal ceases to serve in, or no longer qualifies for, such position, or transfers his or her equity interest, you must designate another qualified person to serve as your Operating Principal within 30 days after the date the prior Operating Principal ceases to serve or no longer qualifies to serve. Any proposed replacement Operating Principal is subject to our written approval and must successfully complete, to our satisfaction, all training we require before becoming the Operating Principal and, in no event, later than 90 days after the previous Operating Principal ceased to serve in such position.
- 6.2.2. If, at any time, you or your Affiliates operate more than one Rusty Taco® Restaurant, then, in addition to the Operating Principal, you must appoint an individual, who need not have an ownership interest in you, to serve as Restaurant Manager. The Restaurant Manager must be approved by us and, once approved by us to assume the role, shall have full control over day-to-day Restaurant management and operations. Your Restaurant Manager shall devote full-time efforts to the management and operation of the Restaurant and shall not engage in any other business or activity, directly or indirectly, that requires substantial management responsibility or time commitment. If the Restaurant Manager ceases to serve in, or no longer qualifies for, such position, you shall designate another qualified person to serve as your Restaurant Manager within 30 days after the date the prior Restaurant Manager ceases to serve or no longer qualifies to serve. Any proposed replacement Restaurant Manager is subject to our approval and must successfully complete, to our satisfaction, all training we require before becoming the Restaurant Manager and, in no event, later than 90 days after the previous Restaurant Manager ceased to serve in such position.
- 6.3. Employee Policy; Uniforms and Employee Appearance. You shall maintain a competent, conscientious, and trained staff, and shall take such steps as are necessary to ensure that your employees preserve good customer relations, render competent, prompt, courteous, and knowledgeable service, and meet the minimum Standards that we mandate from time to time. You shall cause all employees, while working at the Restaurant, to wear uniforms of such color, design, and other specifications as we may designate from time to time, and present a neat and clean appearance. In no case shall any of your employees, while working for you at any business other than the Restaurant, dress in the manner they are required to dress when working in the Restaurant. You acknowledge and agree that you are solely responsible for all decisions relating to employees, agents, and independent contractors that you hire to assist in the operation of your Restaurant. You agree that any employee, agent or independent contractor that you hire will be your not our employee, agent or independent contractor. You also agree that you are exclusively responsible for the terms and conditions of employment of your employees, including recruiting, hiring, firing, training, compensation, work hours and schedules, work assignments, safety and

security, discipline, and supervision. You agree to manage the employment functions of your Restaurant in compliance with federal, state, and local employment laws.

6.4. Authorized Menu Offerings. You must offer and sell all menu items that we require and only those menu items that we have approved as of the time you offer and sell them. You must prepare, package, and serve all menu items in accordance with our Standards, which may include use of specific ingredients, adherence to recipes, use of specific containers and paper goods, packaging procedures, requirements or prohibitions relating to "combo meals," product holding times, and other standards for displaying for sale menu items and other merchandise. You must participate in all product and market research programs we require (which may include test-marketing new products) and provide us with timely reports and test results in accordance with our guidelines for each program.

6.5. Purchases from Designated Sources.

- 6.5.1. We reserve the right, in our sole discretion, to approve or designate, and to revoke our approval or designation of, all suppliers, vendors, and distributors (each a "**Designated Supplier**") of any and all products, goods, services, materials, and other items you use in the development of your Restaurant and in the operation of your Restaurant, including in the preparation, promotion, offer and sale of goods and services (each an "**Item**"). You agree to purchase all Items only from Designated Suppliers. We and our Affiliates may be a Designated Supplier and may be the only Designated Supplier for certain Items. We and our Affiliates, as Designated Suppliers, may also establish, own and operate commissaries and distribution facilities. We and our Affiliates may derive revenue and other consideration from the sale of Items to you and from other Designated Suppliers based on your purchases of Items from them. We and our Affiliates have the right to retain and use, or not use, in our and their discretion, all such revenue and other considerations.
- 6.5.2. By designating suppliers, we are not guaranteeing or making any warranties regarding the performance of Designated Suppliers or a particular price or continuous supply of any Item nor are we assuming responsibility for any delays, damages, or defects related to Items or the quality of Items you purchase or are required to purchase from Designated Suppliers. You hereby waive all such claims against us and our Affiliates.
- 6.5.3. Subject to our ongoing right to approve and designate Designated Suppliers, you may purchase any Items for which we have not approved or designated a Designated Supplier from any supplier, vendor or distributor as long as the Items meet our Standards and specifications for such Items. These specifications may include certain designated brands of such Items ("**Approved Brands**"), in which case, you may purchase and use only the Approved Brands that meet any other Standards and specifications.
- 6.5.4. We are not required to consider alternative suppliers, vendors or distributors for any Item for which we have approved or designated a Designated Supplier. However, if you would like us to consider approving or designating a supplier, vendor or distributor for Items for which there is not then a Designated Supplier or an alternative to a Designated Supplier, you shall submit a written request for our approval. If we elect, in our discretion, to investigate your request, we reserve the right to charge you or the proposed supplier, vendor or distributor a reasonable fee and require that you or they reimburse the costs and expenses we incur in investigating your request. We will determine the scope and parameters of our investigation and the conditions we may impose should we decide, in our discretion, to approve your proposed supplier, vendor or distributor.

6.6. The Restaurant Facilities; Vehicles.

6.6.1. You shall, at your expense, maintain the Restaurant (including adjacent public areas) in a clean, orderly, condition and in excellent repair, and equipped and outfitted, in every respect, in accordance with

our Standards. You shall, at your expense, make such additions, alterations, repairs, and replacements as may be required for that purpose, including such periodic repainting, repairing, and replacing of obsolete or deteriorated signs, furnishings, fixtures, equipment, and decor as we may reasonably direct. You must use such systems and devices as we require, and you will not play or show any music, programming, or other entertainment that we do not approve.

- 6.6.2. At our request, you shall, at your own expense, refurbish the Restaurant, replace or renovate equipment, remodel and redecorate the Restaurant and its premises, and make such modifications to existing improvements and reasonable structural changes that we may reasonably require or that may be required by law, in each case, as necessary to conform to the building design, trade dress, color schemes, and presentation of the Marks in a manner consistent with the then-current public image for new or remodeled Rusty Taco Restaurants in the System.
- 6.6.3. You shall purchase and install, at your expense, all fixtures, furnishings, equipment, decor, signs, and other items as we may reasonably direct from time to time in accordance with our Standards; and shall refrain from installing or permitting to be installed on or about the Restaurant, any fixtures, furnishings, equipment, decor, signs, vending, or game machines, pay telephones, automatic teller machines, Internet kiosks, or other similar items not previously approved in writing as meeting our Standards.
- 6.6.4. At our request, you shall make such alterations as may be necessary to reflect new product offerings and marketing incentives, including updating or replacement of menu boards and the purchase and use of new interior signage, graphics, and/or point of sale materials.
- 6.6.5. Any vehicle used in connection with the operation of the Restaurant must meet our Standards. You shall place such signs and decor items on the vehicle as we require and keep the vehicle clean and in good working order. No one under the age of 18 and no one without a valid driver's license may be permitted to operate a vehicle used in connection with the Restaurant. Each individual who operates a vehicle used in connection with Restaurant operations must comply with all Applicable Laws and rules of the road and use due care and caution in the operation and maintenance of the vehicles. Except as stated herein, we do not exercise any control over any motor vehicle used in the operation of your Restaurant.
- 6.7. **Days and Hours of Operation.** You shall cause the Restaurant to be open and operating on the days (currently seven days per week) and during the hours that we prescribe. You agree that, in the event an executive order is issued that directly affects the operation of your Restaurant, you will not close your Restaurant unless: (1) you obtain our prior written consent; and (2) it is specifically in response to an applicable executive order and not based on recommendations or guidelines issued by a governmental authority or agency or the Centers for Disease Control and Prevention.
- 6.8. <u>Inspections; Testing.</u> We have the right to enter the Restaurant premises, including conduct "virtual" Restaurant visits via in-Restaurant video camera systems or other developing technology, during regular business hours to inspect the Restaurant for quality assurance purposes. You may be required to participate in various food safety, cleanliness, or other programs offered by us or third parties, which may require in-Restaurant visits by such third parties, and you may be required to pay the fees associated with such third-party visits. Such inspections are for the purpose of enforcing the Standards. They are not intended to exercise, nor do they constitute, control over your day-to-day operation of your Restaurant. You shall allow us from time to time to obtain samples of ingredients, products, and supplies, without charge therefore, for testing for quality assurance purposes. If notified of a deficiency in our Standards, you must promptly cure the deficiency.

- 6.9. Pricing. Unless prohibited by Applicable Laws, we may periodically set a maximum or minimum price that you may charge for products and services offered by your Restaurant, including restriction on the use of coupons/discounts and participation in company discount programs. If we impose such a maximum or minimum price, you may charge any price for the product or service up to and including our designated maximum price or down to and including our designated minimum price. The designated maximum and minimum prices for the same product or service may, at our option, be the same. For any product or service for which we do not impose a maximum or minimum price, we may require you to comply with an advertising policy adopted by us which will prohibit you from advertising any price for a product or service that is different than our suggested retail price. Although you must comply with any advertising policy we adopt, you will not be prohibited from selling any product or service at a price above or below the suggested retail price unless we impose a maximum price or minimum price for such product or service.
- 6.10. Technology Fee. We have the right to impose, and you agree to pay to us, a recurring technology fee in the amount that we periodically require (subject to a maximum of \$1,000 per month, plus to the actual technology and related out of pocket expenses we incur) for identifying and/or developing certain technology for the Rusty Taco Restaurants and for providing certain related maintenance and support services to you. In addition to charging the technology fee, we also have the right to collect from you and remit to any third-party technology providers any amounts you owe them. In addition to paying us the technology fee, we you may required to directly purchase or lease certain technology products and related services directly from our designated or approved third-party vendors.
- 6.11. Contact Identifiers and Listings. You agree that each Contact Identifier will be used solely to identify your Restaurant in accordance with this Agreement. You acknowledge and agree that, as between us and you, we have the sole rights to, and interest in, all Contact Identifiers and all Online Presences. You hereby authorize us and irrevocably appoint us or our designee as your attorney-in-fact to direct the telephone company, postal service, registrar, Internet Service Provider and all listing agencies to transfer such Contact Identifiers to us.
- 6.12. Data Privacy and Information Security. You may from time to time have access to information that can be used to identify an individual, including names, addresses, telephone numbers, email addresses, employee identification numbers, signatures, passwords, financial information, credit card information, biometric data, government-issued identification numbers and credit report information ("Personal Information"). You may gain access to such Personal Information from us, our Affiliates, our vendors, or your own operations. You acknowledge and agree that all Personal Information (other than Restricted Data, defined below) is our Confidential Information and is subject to the protections granted our Confidential Information under this Agreement.

During and after the Term, you (and your Owners) agree to, and to cause your respective current and former employees, representatives, Affiliates, successors and assigns to: (a) process, retain, use, collect, and disclose all Personal Information only in strict accordance with all applicable laws, regulations, orders, the guidance and codes issued by industry or regulatory agencies, and the privacy policies and terms and conditions of any applicable Online Presence; (b) assist us and our Affiliates in complying with all our and their obligations under all applicable laws and regulations relating to Personal Information, including the guidance and codes of practice issued by industry or regulatory agencies; and (c) promptly notify us of any communication or request from any customer or other data subject to access, correct, delete, opt-out of, or limit activities relating to any Personal Information.

If you become aware of a suspected or actual breach of security or unauthorized access involving Personal Information, you will notify us immediately of the breach or unauthorized access and specify the extent to which Personal Information was compromised or disclosed, and your plans to correct and prevent any further breach or unauthorized access. You must also promptly notify us if you receive any complaint,

notice, or communication, whether from a governmental agency, customer or other person, relating to any Personal Information, or your compliance with your obligations relating to Personal Information under the Agreement, and/or if you have any reason to believe you will not be able to satisfy any of your obligations relating to Personal Information under the Agreement. You will allow us, in our discretion, to participate in any response or corrective action.

Notwithstanding anything to the contrary in the Agreement or otherwise, you agree that we do not control or own any Restricted Data. Regardless of any guidance we may provide generally and/or any specifications that we may establish for other Personal Information, you have sole and exclusivity responsibility for all Restricted Data, including establishing protections and safeguards for such Restricted Data; provided, that in each case you agree to comply with all applicable laws, regulations, orders, and the guidance and codes issued by industry or regulatory agencies applicable to such Restricted Data.

7. MARKS, STANDARDS AND COPYRIGHTS

- 7.1. Acknowledgments. You agree that: (i) as between us and you, we and our Affiliate own all right, title, and interest in and to, and the goodwill associated with, the Marks and the Standards, (ii) you have no ownership interest in the Marks and the Standards, (iii) you will not contest or assist anyone else in contesting our or our Affiliate's ownership or validity of the Marks or any registration thereof or engage in any conduct that adversely affects the ownership or registration of the Marks or our right to use or to sublicense them, (iv) any and all goodwill associated with the Restaurant and identified by the Marks is our Affiliate's property and will inure directly and exclusively to the benefit of that entity, (v) on the expiration or termination of this Agreement for any reason, no monetary amount will be assigned to any goodwill associated with your use of the Marks, and (vi) your right to use the Marks and the Standards is limited to the scope of the license granted herein does not extend beyond the termination or expiration of this Agreement. You agree to execute all documents that we or our Affiliate requests in order to protect the Marks, the Standards, or to maintain their validity and enforceability.
- 7.2. <u>Limitations on Your Use of the Marks.</u> You agree to use only those Marks and Standards that we designate from time to time and to use them only in the operation and promotion of the Restaurant in accordance with this Agreement and in the manner and for the purposes we authorize. Any unauthorized use of the Marks other will constitute an infringement of our and our Affiliate's rights in and to the Marks.
 - 7.2.1. You may not use the Marks or any part thereof in your corporate name, and you may not use them to incur any obligation or indebtedness on our or our Affiliates' behalf.
 - 7.2.2. Unless we separately provide written authorization, you may not use or display the Marks, any part or derivative thereof, or any Copyrighted Works in or in connection with any Online Presence. You represent that you do not own or control, and expressly disclaim ownership of, any domain name or Internet address that contains a Mark or any part or derivative thereof.
 - 7.2.3. You must not use any Mark that implies that you are the owner of it. You shall identify yourself as an independent franchise owner of the Restaurant in the manner we specify from time to time. You agree to refrain from identifying yourself in or signing any business or legal documents as "Rusty Taco." All such documents shall reference and be signed only in your legal name. If you are required under Applicable Laws to file an assumed name registration with any governmental agency, you must first clear the assumed name with us. Your assumed name shall never be simply "Rusty Taco."
- 7.3. <u>Infringement.</u> You shall promptly notify us of any suspected unauthorized use of, or any challenge to the validity of the Marks, the Standards, the Copyrighted Works, or any challenge to our or our Affiliates' ownership of, our license to use and to license others to use, or your right to use, the Marks, the Standards, or Copyrighted Works licensed under this Agreement. We will have complete discretion to take any action

we deem appropriate in connection with any infringement of, or challenge or claim to, any Mark or Copyrighted Work and the right to control exclusively, or to delegate control of, any legal, regulatory, or administrative proceeding, including the settlement thereof, regarding any Marks or Copyrighted Works. You shall execute any and all documents and do such acts as may, in our opinion, be necessary to carry out such defense or prosecution, including, but not limited to, becoming a nominal party to any legal action. Except to the extent that such litigation is the result of your use of the Marks or Copyrighted Works in a manner inconsistent with the terms of this Agreement, we will reimburse you for your associated costs.

- 7.4. Changes to and Discontinuance of Marks. We reserve the right, in our sole discretion, to designate one or more new, modified, or replacement Marks for your use and to require your use of any such new, modified, or replacement Marks in addition to or in lieu of any previously designated Marks. You must comply with any such directive relating to the use of new, modified, or replacement Marks within the time periods we specify depending on the reason for the change. We need not reimburse you for your expenses of complying with our direction in this regard or for any loss of revenue due to any modified or discontinued Mark. We may exercise these rights at any time and for any reason, and you waive any claims, demands or damages arising therefrom.
- 7.5. Non-Disparagement. You agree not to (and to use your best efforts to cause your current and former owners, officers, directors, agents, employees, representatives, attorneys, spouses, Affiliates, successors and assigns not to) disparage or otherwise speak or write negatively, directly or indirectly, of us, our Affiliates, any of our or our Affiliates' directors, officers, employees, representatives, current and former franchisees or developers, the Rusty Taco® brand, the System, any Rusty Taco Restaurants (including your own), any business using the Marks, any other brand or concept of us or our Affiliates, or which would subject the Rusty Taco® brand or such other brands to ridicule, scandal, reproach, scorn, or indignity, or which would negatively impact the goodwill of the Marks or the System.

8. SYSTEM, OPERATIONS MANUAL, AND CONFIDENTIAL INFORMATION

- 8.1. **Operations Manual.** We will provide you online access to the Operations Manual during the Term. The Operations Manual is part of our Confidential Information and will, at all times, remain our sole property. If any dispute arises as to the contents of the Operations Manual, the terms of the master copy of the Operations Manual we maintain shall be controlling.
- 8.2. **System Modification.** We may, from time to time, modify and vary the requirements applicable to the System, the Standards, the Operations Manual, individual Rusty Taco Restaurants, and the products and services that may be offered by Rusty Taco Restaurants (including the addition, deletion, and modification of menu items, operating procedures, products, services, decor, and trade dress). You agree to comply with all such modifications of mandatory provisions that we communicate to you. Further, we reserve the right to materially vary the form of franchise agreement that we use to grant Franchises for other Rusty Taco Restaurants. You have no right to require us to disclose any variation that we grant to other Rusty Taco Restaurants or to grant the same or a similar variation to you.
- 8.3. Confidentiality. You agree that your relationship with us does not vest in you any interest in the Confidential Information other than the right to use it in the development and operation of your Restaurant in accordance with this Agreement, and that the unauthorized use, duplication or improper distribution or publication of the Confidential Information in any case would constitute an unfair method of competition. You acknowledge and agree that the Confidential Information is proprietary, includes trade secrets belonging to us and is disclosed to you or authorized for your use solely on the condition that you agree, and you therefore do agree, that you will (and cause your Owners, employees, and other personnel to):

 (1) process, retain, use, collect, and disclose our Confidential Information strictly as necessary for the development and operation of the Restaurant in accordance with this Agreement and in accordance with the privacy policies and Standards we establish from time to time, and not for any other purpose of any

kind; (2) maintain the absolute confidentiality of the Confidential Information during and after the Term; (3) not make unauthorized copies of, or improperly disclose or publish any portion of, the Confidential Information however and in whatever form or format disclosed to you; (4) adopt and implement all reasonable procedures we periodically prescribe to prevent unauthorized use or disclosure of the Confidential Information, including, by establishing reasonable security and access measures, imposing restrictions on disclosure to your key employees, and requiring the execution of nondisclosure agreements we may approve or prescribe for employees or others who have access to the Confidential Information; and (5) not sell, trade or otherwise profit in any way from the Confidential Information, except using methods approved by us. The foregoing restrictions do not apply to: (i) disclosure or use of information that is lawfully known and used in the food service industry or by the public generally (as long as the availability is not because of a violation of Applicable Laws or an obligation to us or our Affiliates by you), provided that you have first given us written notice of your intended disclosure and/or use; (ii) disclosure of the Confidential Information in judicial or administrative proceedings when and only to the extent you are legally compelled to disclose it, provided that you have first given us the opportunity to obtain an appropriate protective order or other assurance satisfactory to us that the information required to be disclosed will be treated confidentially; and (iii) disclosure of your Restaurant's operating results and financial performance to your existing and prospective lenders, and, provided they are bound by confidentiality obligations, to potential investors in you or purchasers of your Restaurant.

8.4. <u>Innovations.</u> You must promptly disclose to us all ideas, concepts, methods, techniques, and products conceived or developed by you and/or any of your Affiliates, owners, agents, representatives, contractors, or employees during the Term relating to the development or operation of your Restaurant or other Rusty Taco Restaurants ("Innovations"), whether or not protectable intellectual property and whether created by or for you or your owners or employees. All Innovations are our sole and exclusive property and works made-for-hire for us and shall constitute our Confidential Information. To the extent any Innovation does not qualify as a work made-for-hire for us, you assign ownership of that Innovation, and all related rights to that Innovation, to us and agree to sign (and to cause your owners, employees, and contractors to sign) whatever assignment or other documents we request to evidence our ownership or to help us obtain intellectual property rights in the Innovation. We and our Affiliates have no obligation to make any payments to you or any other person with respect to any Innovations. You may not use any Innovation in operating your Restaurant or otherwise without our prior written approval.

9. **ADVERTISING AND MARKETING**

9.1. **General; Marketing Cap.**

9.1.1. In this Article 9, we require that you engage in and contribute to certain types of marketing activities, including a grand opening program, contributions to a brand development Fund, spending on local marketing activities and contributions to a local marketing co-op. Required contributions to the Fund (Section 9.3), minimum Local Advertising expenditure requirement (Section 9.4) and required contributions to a local marketing co-op (Section 9.5) are collectively referred to herein as the "Required Marketing Spend." Excluding the amounts you are required to spend on the grand opening program under Section 9.2 below, we will not require that you spend on all Required Marketing Spend during any calendar quarter more than six percent (6%) of the Restaurant's Gross Sales (the "Marketing Cap"). It is your responsibility to provide us with written notice if the Required Marketing Spend exceeds the Marketing Cap, and until we receive your written notice (the "Marketing Notice"), you will fully comply with the Required Marketing Spend, and no excess amounts will be refunded to you. If the Required Marketing Spend exceeds the Marketing Cap, you may, after we receive your Marketing Notice, reduce the local advertising expenditure, but only to the extent and for the time necessary to not exceed the Marketing Cap. You must immediately return to full compliance with Required Marketing once the Marketing Cap is no longer exceeded.

- 9.1.2. We may require you to spend or contribute the amounts we specify from time to time on the different elements of Required Marketing Spend as long as we do not require you to spend or contribute, on all such activities, more than the Marketing Cap during any calendar quarter. We will notify you 60 days in advance of any change to your required contribution to any component of the Required Marketing Spend. We will not count towards your Marketing Cap the cost of free or discounted products or services, coupons, special offers or price reductions that you provide as a promotion, signs, personnel salaries, administrative costs, employee incentive programs, or other amounts that we, in our reasonable judgment, deem inappropriate for meeting the Marketing Cap. We may periodically review your books and records and require you to submit reports to determine the amounts you have spent on Required Marketing Spend. If you fail to spend (or prove that you spent) the required amount on Local Advertising in any quarter, then we may, in addition to and without limiting its other rights and remedies, require you to pay us the shortfall as, in our discretion, either an additional Fund contribution or for us to spend on Local Advertising for the Restaurant.
- 9.1.3. All of your promotional and marketing materials must be presented in a dignified manner and must conform to our Standards related to advertising, marketing, and trademark use. You shall submit to us samples of any proposed promotional and marketing materials that we have not previously approved or prepared and notify us of the intended media before first publication or use. You may not use the promotional or marketing materials until we expressly approve the materials and proposed media. Once approved, you may use the materials only in connection with the media for which they were approved. We may disapprove your promotional or marketing materials, or the media for which they were approved, at any time, and you must discontinue using any disapproved materials or media upon your receipt of written notice of disapproval.
- 9.2. Grand Opening Marketing Campaign. In addition to all other advertising and promotion obligations described in this Article 9, you must conduct an initial advertising and marketing campaign to promote the opening of the Restaurant. The initial advertising and marketing campaign must comply with our Standards, which currently require that you spend at least \$10,000 on the campaign, and which may include specifications regarding, among other things, use of certain materials and media, the length and timing of the promotion, and the amount spent on each. You must obtain our approval of all advertising items, methods, and media you use in connection with grand opening advertising program. You must submit to us an accurate accounting of all such expenditures upon our request.
- 9.3. **Brand Development Fund.** We have established a brand development fund (the "**Fund**") to be used to promote awareness of the Rusty Taco[®] brand and System generally. You agree to contribute to the Fund in the amounts we specify, in writing, from time to time (currently 2% of the Restaurant's Gross Sales), subject to the Marketing Cap. You must submit payments to the Fund with and in the same manner as your payments of Royalty Fee.
 - 9.3.1. We or our designees will control all activities of the Fund. We have the right to use Fund monies, in our sole discretion, to pay for: creative development services (including creation and modification of Restaurant design and trade dress, logos, menu design, graphics and vehicle wraps, and advertising and promotional items, including the cost of retiring expired or obsolete printed items and materials, photography services, and design software); preparing and procuring market studies, providing or obtaining marketing services (including new product development; conducting customer surveys, focus groups, and marketing-related mystery shops and customer interviews); developing, promoting, implementing and executing limited time offerings; employing advertising and/or public relations agencies; developing, producing, distributing and placing advertising (including preparing and conducting media advertising campaigns in various media, local Restaurant advertising and promotion in a particular area or market, or for the benefit of a particular Restaurant or Restaurants in connection with Restaurant opening promotions or otherwise); conducting and administering in-Restaurant promotions; preparing and executing direct mail advertising, and developing, producing and purchasing point-of-sale

advertising, menus and menu boards, and other sales aids and promotional items and materials); research and development and menu development (including purchase of equipment (including ovens), smallwares, and packaging required for new product testing and development, hiring culinary talent and/or engaging culinary consultants); developing, updating, and hosting our website (including development of locator programs) and/or an intranet or extranet system; obtaining sponsorships and endorsements; preparing and conducting sweepstakes and other promotions; developing, administering, and distributing coupons, certificates and Restaurant value card programs, and the cost of products associated with the redemption of free coupons, gift certificates and Restaurant value cards; developing and administering other customer loyalty programs; developing and administering online ordering platforms and related software applications; providing and procuring public relations services; conducting public relations activities; charitable donations; and membership fees in international, national, regional, and/or local trade or other associations or organizations. From time to time, we may engage in initiatives, new or otherwise, relating to the brand development activities stated above, and in order to implement or maintain the new or existing initiatives, we may use Fund monies to pay for the initial and recurring costs of new or existing initiatives. We reserve the right to shift all or a portion of an initiative's recurring costs from the Fund to the franchisees, as authorized under this Agreement. We also may use Fund monies to reimburse our and our Affiliates' costs of personnel and other administrative and overhead costs (whether internal or third party) associated with providing the services described in this Section 9.3.

- 9.3.2. You acknowledge and agree that we or our Affiliates own all rights, and retain all copyrights, in all design and content developed using Fund monies, and that we will have sole control over the creative concepts, content, form, and media placement of all advertising and promotional materials developed with Fund monies, and the allocations of Fund monies to production, placement, and other costs. We will own all copyright in any works created using Fund monies. You acknowledge and agree that we are not obligated to expend Fund monies for placement of advertising in your trading area, or to ensure that your Restaurant benefits directly or *pro rata* from the expenditure of Fund monies. We will not use Fund monies for creating or placing any advertisement that is principally a solicitation for new franchisees, but may include in all advertising prepared using Fund monies (including Internet advertising) information concerning franchise opportunities, and a portion of Fund monies may be used to create and maintain one or more pages on our website devoted to advertising franchise opportunities and identifying and screening inquiries and applications submitted by franchise candidates. We have no fiduciary duty to you or to any other person with respect to the collection or expenditure of Fund monies. Upon your reasonable request, we will provide you with an annual statement of Fund contributions and expenditures.
- 9.3.3. Although the Fund is intended to be perpetual, we may terminate the Fund at any time. On termination, all remaining Fund monies will, in our discretion, be spent as provided in this Section 9.3 or returned to the Fund contributors in the manner we deem fair and equitable.
- 9.3.4. We are not required to segregate Fund contributions from our other funds, but we will account for the Fund separately from our other funds and, except as permitted herein, we will not use the Fund for any of our general operating expenses. The Fund will not be our asset. We will hold all Fund contributions for the benefit of the contributors and use contributions for the purposes described in this Section 9.3. We do not owe any fiduciary obligation to you for administering the Fund or any other reason. The Fund may, in our discretion, spend in any fiscal year more or less than the total Fund contributions in that year, borrow from us or others (paying reasonable interest) to cover deficits, or invest any surplus for future use. We may use all interest earned on the Fund contributions to pay costs before using the Fund's other assets.
- 9.4. <u>Local Advertising.</u> We reserve the right to require you to spend the amount that we periodically specify, subject to the Marketing Cap, on local brand development and advertising activities that we approve ("Local Advertising"). Any amounts spent on grand opening advertising under Section 9.2 shall not be

credited toward satisfaction of your Local Advertising expenditure obligations under this Section 9.4.

- 9.5. **Franchisee Advertising Cooperatives.** When there are multiple Rusty Taco Restaurants operating in the same geographical area (as we define), we may establish or direct the establishment of a local advertising cooperative ("**Local Advertising Co-op**"), the purpose of which is, with our approval, to administer coordinated advertising programs and develop advertising, marketing and promotional materials for the area covered by the Local Advertising Co-op. The Local Advertising Co-op will be organized and governed in a form and manner, and begin operating on a date that we determine. We may change, dissolve and merge Local Advertising Co-ops. You agree to sign any documents we require to become a member of a Local Advertising Co-op and, subject to the Marketing Cap, to contribute to and participate in the designated Local Advertising Co-op.
- 9.6. Loyalty Programs, Prize Promotions, Meal Deals, and Promotional Literature. You shall participate in and offer to your customers: (a) all customer loyalty and reward programs; (b) all contests, sweepstakes, and other prize promotions; (c) all meal deals, which we may require from time to time; and (d) customized promotional receipts programs. We will communicate to you in writing the details of each such program, promotion, and meal deal, and you shall promptly display all point-of-sale advertising and promotion-related information at such places within the Restaurant as we may designate. You shall purchase and distribute all coupons, clothing, toys, and other collateral merchandise (and only the coupons, clothing, toys, and collateral merchandise) we designate for use in connection with each such program, promotion, or meal deal. If any such program requires that you acquire and use specialized computer software and hardware necessary to process their sale and redemption, you will do so and be solely responsible for any service charges related to such processing. You also shall display at the Restaurant all promotional literature and information regarding such programs as we may reasonably require from time to time. This may include, among other things, displaying signage or other literature containing information about the Rusty Taco franchise offering.
- 9.7. Participation in Marketing Programs. You shall at all times cooperate with us and other operators of Rusty Taco Restaurants and shall actively participate in any and all sales, public relations, advertising, cooperative advertising, and purchasing programs or promotional programs (including product give-away promotions) as we specify from time to time. Participation may include purchasing (at your expense) and using: (a) designated point-of-sale materials; (b) counter cards, displays, and give-away items promoting loyalty programs, prize promotions, movie tie-in promotions, and other marketing campaigns and programs; (c) product mix and ingredients for product giveaways; and (d) equipment necessary to administer loyalty programs and to prepare and print customized purchase receipts, coupons.
- 9.8. **Business Directory Listings.** You shall place and pay the cost of business listings in print and online directories and categories, whether electronic or hard copy, as we may require from time to time. Amounts paid for business directory listings will not be credited toward the Marketing Cap.
- 9.9. Websites and Online Presences. We may establish, develop and update Online Presences to advertise, market, and promote Rusty Taco Restaurants, the products and services that they offer and sell, or the Rusty Taco franchise opportunity (each a "System Website"). We may, but are not obligated to, provide you with a webpage or other Online Presence that references your Restaurant on any System Website, upon which, you must: (i) provide us the information and materials we request to develop, update, and modify the information about your Restaurant; (ii) notify us whenever any information about your Restaurant is not accurate; and (iii) pay our then current initial technology fee for the Online Presences that are dedicated to your Restaurant. You acknowledge that we have final approval rights over all information on any System Website.
 - 9.9.1. If you default under this Agreement, we may, in addition to our other remedies, temporarily remove references to your Restaurant from any System Website until you fully cure the default, and our

doing so will not be considered a termination of this Agreement. All advertising, marketing, and promotional materials that you develop for your Restaurant must contain notices of the System Website's domain name in the manner we designate.

- 9.9.2. You may not, without our prior written consent, develop, maintain or authorize any Online Presence that mentions your Restaurant, links to any System Website, or displays any of the Marks. You may not, directly or indirectly, through any Online Presence, promote, advertise or sell any products or services without our prior written approval. If we approve the use of any such Online Presence in your Restaurant's operations, you will develop and maintain such Online Presence only in accordance with our guidelines, including our guidelines for posting any messages or commentary on other third-party websites. Unless we specify otherwise, we will own the rights to each such Online Presence. At our request, you agree to grant us access to each such Online Presence, and to take whatever action (including signing assignment or other documents) we request to evidence our ownership of such Online Presence, or to help us obtain exclusive rights in such Online Presence. If we allow you to maintain an Online Presence for your Restaurant, you must prepare and link a privacy policy to such Online Presence. Your Online Presence's privacy policy must comply with all applicable laws, the Standards, and other terms and conditions that we may prescribe in writing.
- 9.9.3. TO THE MAXIMUM EXTENT PERMITTED BY LAW, WE HEREBY EXPRESSLY DISCLAIM ALL WARRANTIES (WHETHER EXPRESS, IMPLIED OR STATUTORY) RELATED TO THE AVAILABILITY AND PERFORMANCE OF A SYSTEM WEBSITE AND YOUR RESTAURANT'S PAGE, INCLUDING ANY IMPLIED WARRANTIES OF NON-INFRINGEMENT, MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE. TO THE MAXIMUM EXTENT PERMITTED BY LAW, WE SHALL NOT BE LIABLE FOR ANY DIRECT OR INDIRECT DAMAGES (INCLUDING ANY CONSEQUENTIAL, PUNITIVE OR INCIDENTAL DAMAGES OR DAMAGES FOR LOST PROFITS OR LOSS OF BUSINESS) RELATED TO THE USE, OPERATION, AVAILABILITY OR FAILURE OF THE SYSTEM WEBSITE OR YOUR RESTAURANT'S PAGE.

10. COMPUTER SYSTEM; ACCOUNTING AND RECORDS; TAXES

- Computer System. You shall acquire and use only the point-of-sale cash registers and computer systems 10.1. and equipment that we prescribe for use by Rusty Taco® Restaurants ("Computer System") and adhere to our requirements for use, including accounting for all Gross Sales through the Computer System. Requirements may include, among other things, connection to remote servers, off-site electronic repositories, and high-speed Internet connections. We may, in our sole discretion, require you to add to your Computer System memory, ports, and other accessories or peripheral equipment or additional, new, or substitute software, replace or upgrade your Computer System (software and hardware) and other computers as we prescribe, and enter into maintenance agreements for the Computer System and other computers. We will provide you with 30 days' advance written notice of any change to our Computer System requirements, and you must comply with all requirements within the time period we require. You must acquire, install, and maintain such anti-virus and anti-spyware software as we require, and you must adopt and implement such Internet user policies as we may prescribe for purposes of avoiding, blocking, and eliminating viruses and other conditions that interfere with operation of the Computer System. We may require you to provide us with all passwords, access keys, and other security devices as necessary to permit our remote access.
- 10.2. <u>Software.</u> You must: (a) use any and all proprietary software programs, system documentation manuals, and other proprietary materials that we require in connection with the operation of the Restaurant; (b) input and maintain in your computer such data and information as we prescribe or as provided in the software programs, documentation, or otherwise; and (c) purchase new or upgraded software programs, system documentation manuals, and other proprietary materials at then-current prices whenever we adopt such new or upgraded programs, manuals, and materials system-wide. You must enter into all software

license agreements, "terms of use" agreements, and software and hardware maintenance agreements, in the form and manner we prescribe, and pay all fees imposed thereunder. We reserve the right to require the implementation and use of a learning management system throughout the Term, which may require you to incur an additional cost (which will not exceed \$1,000 per month).

- 10.3. <u>Independent Access.</u> We may independently poll Gross Sales and other information input and compiled by your Computer System from a remote location, and you agree to do all things necessary to facilitate, and refrain from doing anything to impede, our ability to do so. We may also access your Computer System and retrieve, analyze, download, and use all software, data, and files stored or used on the Computer System. We may access the information system through our intranet, in your Restaurant, or from or through other locations and methods. You must store all data and information that we designate and report data and information in the manner we specify, including through our intranet, file exports, or other online communications or services. You also must maintain a phone line and a high-speed, broadband Internet connection, at our current, minimum bandwidth specification, dedicated to the sole use of allowing the exchange of information between you, us, and any third-party vendors that we require, and any software and/or services designated by us to facilitate these communications.
 - 10.3.1. You may require you to obtain from us and use an email address associated with our registered domain name. If we require you to obtain and use such an email address, you must do so according to our then-current terms and conditions and Standards. You acknowledge and agree that we will have unrestricted access to all such email accounts, and all documents, data, materials, and messages shared from or by such accounts. We may deactivate any such account or limit your or your users' access to it at any time. We reserve the right to charge you a recurring fee (to off-set the cost we incur) for each email address with which we provide you.
- 10.4. Maintenance of Records. You shall prepare and preserve for at least five (5) years from the date of preparation complete and accurate books, records, and accounts according to generally accepted accounting principles and in the form we prescribe. Additionally, concurrently with the execution of this Agreement, you must provide us true and accurate copies of your charter documents and governing documents, and any amendments to the foregoing. You shall promptly provide such additional information as we may from time to time request concerning all persons who may have any direct or indirect financial interest in you.
- 10.5. <u>Submission of Financial Statements and Tax Returns.</u> No later than March 31 of each calendar year, you shall provide to us: (a) a copy of the previous year's annual profit and loss statements; (b) a copy of the previous year's sales tax returns; and (c) if requested, a copy of your federal and state income tax returns for the previous year; provided, however, that if you are a natural person, you may submit only those schedules to your personal tax returns which reflect the revenues and expenses of the Restaurant.
- 10.6. Submission of Performance Reports. You must accurately report to us the Restaurant's Gross Sales and such other financial information as we may reasonably require using the procedures and reporting dates that we prescribe periodically. We may require reports to be signed by your authorized representative, attesting to their accuracy. Within 30 days following the end of each calendar quarter, you must provide us a copy of your profit and loss statements prepared according to generally accepted accounting principles and which accurately reflect your financial information for the applicable accounting periods. We may require that you submit the profit and loss statements in a standardized format, using standardized charts of account that we provide to all of our franchisees. You also must provide us such other reports, computer back-up, and other information that we may reasonably request. We may also require that you periodically update us on the status of any loans (including the outstanding balance, then-current payment amounts, and whether such loan is in good standing) for which the Restaurant or any Items have been used as collateral. You must also deliver to us, within five (5) days after your receipt, copies of any default notices you receive from any such lenders.

- 10.7. Our Right to Inspect Your Restaurant. We (or our designee) may at all times and without prior notice to you inspect and photograph, capture moving images, and observe your Restaurant and its operations for consecutive or intermittent periods we deem necessary; remove samples of any products and supplies; interview your personnel and customers; inspect your Computer System and its components; and inspect and copy any books, records, and documents relating to the operation of your Restaurant. You agree to fully cooperate with us. If we exercise any of these rights, we will not interfere unreasonably with the operation of your Restaurant. You agree to present to your customers the evaluation forms that we may prescribe and to participate in and request your customers to participate in any surveys performed by or for us. Inspections under this Section will be at our expense, but if the inspection reveals failures to comply with our Standards, you will be required to reimburse the expenses we incur in re-inspecting the Restaurant to confirm that the failures have been correct. Similarly, if you or your personnel prevent us from conducting an inspection, you will reimburse us the costs we incur in connect in with our subsequent efforts to inspect your Restaurant.
- 10.8. Audit of Franchisee Records. We (directly or through our designated agent) have the right to audit, examine, and copy your books, records, accounts, and business tax returns at any time. You agree to cooperate fully with us and our representatives in any audit or examination under this Section. If an inspection or audit reveals underpayment of amounts owed to us, you shall immediately pay the understated amount with interest. If an audit or inspection reveals your understatement of Gross Sales by 2% or more for any Accounting Period then, in addition to amounts due on the understatement and interest, you shall promptly reimburse us all costs and expenses we incurred in connection with performing the audit or inspection (including travel, lodging and wage expenses, and attorneys' and accountants' fees). These remedies are in addition to our other remedies and rights under this Agreement and Applicable Laws.
- 10.9. <u>Disclosure of Information.</u> You hereby authorize us to publish information concerning the Restaurant's Gross Sales and other information reported to us or discovered by us in any inspection or audit, including in our franchise disclosure document and in other reports which may be presented to existing other operators of Rusty Taco Restaurants.
- 10.10. <u>Taxes.</u> You shall promptly pay all taxes due and owing based on your operation of the Restaurant, including sales taxes, income taxes, and property taxes.

11. INDEPENDENT CONTRACTOR, INSURANCE, AND INDEMNIFICATION

- 11.1. Independent Contractor. You are an independent contractor, and nothing contained in this Agreement will create or be construed to create a partnership, joint venture, agency or employment relationship between the parties. Neither party has any fiduciary obligations to the other or will be liable for the debts or obligations of the other. Neither party may bind the other, transact business in the other party's name, or in any manner make any promises or representations on behalf of the other party, unless otherwise agreed in writing, between them. We do not participate in the hiring, promoting, disciplining, or discharging of your employees or in setting or paying wages or benefits to your employees, and you acknowledge that we have no power, responsibility, or liability in respect to the hiring, promoting, disciplining, or discharging of employees or in setting or paying their wages. Additionally, you must communicate to all of your employees that you, not us, are their employer; and you must ensure that no payroll checks or other employment-related documents (such as job applications and W-2s) contain or reference the Marks or our name. Each of the parties will file its own tax, regulatory, and payroll reports with respect to its respective employees and operations, saving and indemnifying the other party hereto of and from any liability of any nature whatsoever by virtue thereof.
- 11.2. <u>Insurance Obligations.</u> You shall, at your expense, procure and maintain in full force and effect at all times during the Term of this Agreement, at your expense, one or more insurance policies that provide

that cowerages required by Applicable Laws and any additional coverages (types and amounts of coverage) that comply, in all respects, with our Standards. Your obligation to obtain and maintain insurance under this Section must not be limited in any way by reason of any insurance which we or our Affiliates may maintain, nor will your performance of that obligation relieve you of liability under the indemnity provisions set forth in Section 11.3 of this Agreement. At least 10 days prior to the time you are first required to carry insurance, and thereafter at least 30 days prior to the expiration of any policy, you shall deliver to us a certificate of insurance evidencing your compliance with this Section 11.2. Each certificate of insurance must expressly provide that no less than 30 days' prior written notice will be given to us in the event of material alteration to or cancellation or non-renewal of the coverage evidenced by such certificates. If you fail to procure or maintain these minimum insurance requirements, we or our designee has the right (but is not required) to procure, at your expense, such insurance on your behalf. Such right will be in addition to and not in lieu of any other rights or remedies available to us under this Agreement.

Indemnification. You shall indemnify and hold harmless, to the fullest extent permitted by Applicable 11.3. Laws, us, our Affiliates and our and their respective directors, officers, employees, owners, agents, successors and assigns (collectively, "Indemnitees") from any and all "losses and expenses" incurred in connection with any litigation or other form of adjudicatory procedure, claim, demand, investigation, or formal or informal inquiry (regardless of whether same is reduced to judgment) or any settlement thereof which arises directly or indirectly from, as a result of, or in connection with any agreement between you and a third party, the training you and your personnel receive from us, your employment or other relationship with your employees, contractors, or other third parties, activities that occur at your Restaurant, or your operation of, or failure to operate, the Restaurant including, but not limited to, claims arising as a result of the maintenance and operation of vehicles or the Location (collectively, "event"), and regardless of whether same resulted from any strict or vicarious liability imposed by law on the Indemnitees; provided, however, that this indemnity will not apply as to an Indemnitee in respect of any losses and expenses arising from the gross negligence of such Indemnitee (except to the extent that joint liability is involved, in which event the indemnification provided in this Agreement will extend to any finding of comparative negligence or contributory negligence attributable to you). For the purpose of this Section 11.3, the term "losses and expenses" will be deemed to include compensatory, exemplary, or punitive damages; fines and penalties; attorneys' fees; experts' fees; court costs; costs associated with investigating and defending against claims; settlement amounts; judgments; compensation for damages to our reputation and goodwill; and all other costs associated with any of the foregoing losses and expenses. Under no circumstances will an Indemnitee be required or obligated to seek recovery from third parties or otherwise mitigate its losses in order to maintain a claim under this provision, and its failure to seek such recovery or mitigate its losses and expenses will in no way reduce the amounts recoverable by such Indemnitee under this provision. You must give us prompt notice of any event of which you are aware, for which indemnification is required, and, at your expense and risk, we may elect to assume (but under no circumstance is obligated to undertake) the defense and/or settlement thereof, provided that we will seek your advice and counsel. Any assumption by us will not modify your indemnification obligation. We may, in our sole and absolute discretion, take such actions as we deem necessary and appropriate to investigate, defend, or settle any event or take other remedial or corrective actions with respect thereof as may be, in our sole and absolute discretion, necessary for the protection of the Indemnitees or the System. This provision survives termination or expiration of this Agreement, regardless of the reason for expiration or termination.

12. TRANSFER OF INTEREST

12.1. <u>Transfer by Us.</u> We may freely Transfer, assign or delegate all or any part of our rights or the performance of any of our obligations under this Agreement. If this Agreement is assigned to a third party, the assignee shall expressly assume and agree to perform, and shall become solely responsible for the performance of, all of our obligations under this Agreement from the date of assignment. In addition, and without limitation to the foregoing, you expressly affirm and agree that our owners may Transfer their

ownership interests in us, and we and our Affiliates may (i) sell some or all of their assets, including the Marks, the Copyrighted Works, or the System, (ii) sell securities in a public offering or in a private placement, (iii) merge, acquire other entities, or be acquired by another entity, and (iv) undertake a refinancing, recapitalization, leveraged buy-out, or other economic or financial restructuring. With regard to any of the foregoing, you expressly and specifically waive any claims, demands or damages arising from or related to the loss of our name, the Marks (or any variation thereof), the Copyrighted Works, the System, and/or the loss of association with or identification of us as the franchisor under this Agreement. You agree that we have the right, now or in the future, to purchase, merge, acquire or affiliate with an existing competitive or non-competitive franchise network, chain or any other business regardless of the location of that chain's or business' facilities, and to operate, franchise or license those businesses and/or facilities as Rusty Taco Restaurants operating as part of the System and under the Marks, the Copyrighted Works or any other marks following our purchase, merger, acquisition or affiliation, regardless of the location of these facilities (which you acknowledge may be proximate to the Restaurant). You specifically waive any and all other claims, demands, or damages arising from or related to a merger, acquisition and other business combination activities as described in this Section, including any claim of divided loyalty, breach of fiduciary duty, fraud, breach of contract or breach of the implied covenant of good faith and fair dealing.

- 12.2. <u>Transfer By You or Your Owners.</u> We have entered into this Agreement based on our specific assessment of you and, if applicable, your Owners and, if you are not a natural person, the specific amount of direct or indirect ownership held by each Owner in you. Therefore, you agree that you and, if applicable, your Owners, may not engage in any form of Transfer without our prior written consent. We will not unreasonably withhold our consent to a proposed Transfer, but you agree that it would not be unreasonable to condition our consent on one or more of the following:
 - 12.2.1. You and, if applicable, the transferring Owner, must submit to us a written request for consent to the proposed Transfer and deliver to us a copy of the proposed agreements related to the proposed Transfer, including sale terms. Your request and relevant documents must be submitted at least 60 days prior to the proposed Transfer, and we must have determined that the terms of the Transfer will not materially and adversely affect the post-Transfer viability of the Restaurant or unduly impact the Rusty Taco[®] brand or System;
 - 12.2.2. The transferee must demonstrate to our satisfaction that it meets our then-current educational, managerial and business standards; possesses good moral character, business reputation and credit rating; has the aptitude and ability to operate the Restaurant; and meets our then-current liquidity and net worth requirements;
 - 12.2.3. All of your accrued monetary obligations and all other outstanding obligations to us, our Affiliates, and your third-party suppliers, including your landlord and lender(s), are, or will be at the time of the Transfer, current and fully paid and satisfied, and you must be in full compliance with this Agreement and any other agreements between you and us, our Affiliates, and your suppliers;
 - 12.2.4. You or the proposed transferee must agree to refurbish the Restaurant premises, as we determine to be necessary in order to meet our then-current image requirements for new Rusty Taco Restaurants, or take such measure as necessary to meet our then-current repair and maintenance standards, whichever we choose at our sole discretion:
 - 12.2.5. You and each Owner (and, the proposed transferee if it is already our franchisee or the owner of another franchisee or Rust Taco Restaurant) must have execute a general release, in a form satisfactory to us, of any and all claims against us and our Affiliates and our and their respective officers, directors, shareholders, agents and employees in their corporate and individual capacities, including claims arising under federal, state and local laws, rules and ordinances;

- 12.2.6. You must pay us \$5,000, plus reasonable attorneys' fees, and reimburse us our reasonable expenses incurred to inspect the Restaurant and enforce Section 12.2.4, above;
- 12.2.7. The transferee must execute our then-current form of Franchise Agreement, the terms of which may be materially different than the terms of this Agreement and may include, among other things, a different percentage Royalty Fee and different advertising obligations. The term of such agreement will be the remaining Term of this Agreement at the time of Transfer. If the transferee wishes to increase the term of the Franchise Agreement, then we may elect to require the transferee pay to us the pro-rated amount of the initial franchise fee that corresponds to the number of years being added to the term. If the transferee is a legal entity, then the transferee's Owners each must sign our standard form of Guaranty and Personal Undertaking;
- 12.2.8. You and your transferring owners (and your and their immediate family members) will not, for two years beginning on the transfer's effective date, engage in any of the activities proscribed in Section 15.2 below;
- 12.2.9. The transferee must comply with our then-current initial training requirements;
- 12.2.10. If we introduced the buyer to you, you must pay all fees due us under our then-current franchise resale policy or program; and
- 12.2.11. If we consent to the Transfer contemplated by this Section 12.2, and if for any reason the Transfer does not close, you must reimburse us for our reasonable attorneys' fees incurred in conjunction with the abandoned Transfer.
- 12.3. <u>Transfers Void.</u> Any purported Transfer, by operation of law or otherwise, made without compliance with the requirements of this Article 12, will be considered null and void and will be considered a material breach of this Agreement.

12.4. **Security Interest.**

- 12.4.1. You may use this Agreement as collateral to secure repayment of your loan only to the limited extent permitted by the Uniform Commercial Code. Any such security interest may only attach to an interest in the proceeds of the operation of the Restaurant and may not entitle or permit the secured party to take possession of or operate the Restaurant or to Transfer your interest in the franchise without our consent. It may not restrict, in any way, our rights to collect fees owed under this Agreement or to enforce the provisions of this Agreement. You agree to provide us with a copy of any proposed loan and security agreements at least 10 days prior to their execution.
- 12.4.2. You hereby collaterally assign to us the lease for the premises on which the Restaurant is located and grant us a security interest in all of the Items and all other assets used in the operation of your Restaurant, including but not limited to inventory, accounts, supplies, contracts, cash derived from the operation of your Restaurant and sale of other assets, and proceeds and products of all those assets. You agree to execute such other documents as we may reasonably request in order to further document, perfect and record our security interest. If you default in any of your obligations under the Agreement, we may exercise all rights of a secured creditor granted to us by law, in addition to our other rights under the Agreement and at law. If an approved third-party lender requires that we subordinate our security interest in the assets of your Restaurant as a condition to lending you working capital for the construction or operation of your Restaurant, we will agree to subordinate pursuant to terms and conditions determined by us. This Agreement shall be deemed to be a security agreement and financing statement and may be filed for record as such in the records of any county and state that we deem appropriate to protect our interests.

- 12.5. **Private or Public Offerings.** If you are not a natural person and you intend to issue equity interests pursuant to a public or private offering, you must first obtain our written consent, which consent will not be unreasonably withheld. You must provide to us for our review a copy of all offering materials (whether or not such materials are required by applicable securities laws) at least 60 days prior to such documents being filed with any government agency or distributed to potential investors. No offering will imply (by use of the Marks or otherwise) that we are participating in an underwriting, issuance or offering of your securities, and our review of any offering shall be limited to ensuring compliance with the terms of this Agreement. We may condition our approval on satisfaction of any or all of the conditions set forth in Section 12.2 and on execution of an indemnity agreement, in a form we prescribe, by you and any other participants in the offering. For each proposed offering, you shall pay to us a retainer in an amount we determine, which we shall use to reimburse ourselves for the reasonable costs and expenses we incur (including attorneys' fees and accountants' fees) in connection with reviewing the proposed offering.
- 12.6. **Right of First Refusal.** If you receive a bona fide offer to purchase your interest in this Agreement or all or substantially all of the assets of the Restaurant, or if any Owner receives a bona fide offer to purchase equity interests in you, and you or such Owner wishes to accept such offer, you or the Owner must deliver to us written notification of the offer and, except as otherwise provided herein, we have the option, exercisable within 30 days after receipt of such written notification, to purchase the seller's interest on the same terms and conditions offered by the third party. If the bona fide offer provides for the exchange of assets other than cash or cash equivalents, the bona fide offer must include the fair market value of the assets and you shall submit with the notice an appraisal prepared by a qualified independent third party evidencing the fair market value of such assets as of the date of the offer. Any material change in the terms of any offer prior to closing will constitute a new offer subject to the same right of first refusal by us as in the case of an initial offer. If we elect to purchase the seller's interest, closing on such purchase must occur by the later of: (a) the closing date specified in the third-party offer; or (b) within 60 days from the date of notice to the seller of our election to purchase. Our failure to exercise the option described in this Section 12.6 will not constitute a waiver of any of the conditions regarding the Transfer set forth in this Article 12.
- 12.7. **Transfer Upon Death or Incapacity.** Upon the permanent incapacity (mental or physical) of your Operating Principal or the death of any of your Owners, the executor, administrator, or personal representative of such person shall, subject to the requirements of this Article 12, complete a Transfer of such person's interests to the heirs of such person within six (6) months after such death or mental incapacity, as applicable; provided, however, that if the heirs of such person are unable to meet the conditions of this Article 12, then the executor, administrator or personal representative will have an additional six (6) months following our disapproval of the heirs to complete a Transfer of such interests to a third party who we approve pursuant to this Article 12. Such Transfer, including a Transfer by devise or inheritance, will be subject to the same conditions as an *inter vivos* Transfer, except that the transfer fee will be waived. If the interests are not disposed of within such period, we may, at our option, terminate this Agreement, pursuant to Section 13.5. Nothing herein is intended to alter your obligation to propose a replacement Operating Principal in accordance with Section 6.2.1 above.
- 12.8. <u>Non-Waiver of Claims.</u> Our consent to a Transfer will not constitute a waiver of any claims we may have against the transferring party, and it will not be deemed a waiver of our right to demand strict compliance with any of the terms of this Agreement, or any other agreement to which we and the transferor are parties.

13. **DEFAULT AND TERMINATION**

- 13.1. Termination In the Event of Bankruptcy, Insolvency or Transfers Required on Death or Disability. You shall be deemed to be in default under this Agreement, and we may terminate this Agreement immediately upon notice, if you become insolvent or make a general assignment for the benefit of creditors; if a petition in bankruptcy is filed by you or such a petition is filed and against you and you do not oppose it; if you are adjudicated as bankrupt or insolvent; if a bill in equity or other proceeding for the appointment of a receiver for you or other custodian for your business or assets is filed and consented to by you; if a receiver or other custodian (permanent or temporary) of your assets or property, or any part thereof, is appointed by any court of competent jurisdiction; if proceedings for a composition with creditors under any state or federal law is instituted by or against you; if a final judgment remains unsatisfied or of record for 30 days or longer (unless a supersedeas bond is filed); if you are dissolved; if execution is levied against your business or property; if judicial, non-judicial or administrative proceedings to foreclose any lien or mortgage against the premises or assets or equipment of the Restaurant is instituted against you and not dismissed within 30 days; or if the real or personal property of the Restaurant is sold after levy thereupon by any sheriff, marshal, or constable. We will also have the right to terminate this Agreement, immediately on delivery of written notice of termination, if an approved Transfer is not timely completed as required by Section 12.9 regarding Transfers on death or incapacity.
- Termination With Notice and Without Opportunity To Cure. We have the right to terminate this 13.2. Agreement, which termination will become effective upon delivery of notice, without opportunity to cure, if: (a) your Operating Principal or Restaurant Manager fails to successfully complete training; (b) you fail to open the Restaurant for regular on-going business by the earlier of the Opening Deadline; (c) you (i) permanently close your Restaurant for business or inform us of your intention to cease operation of your Restaurant, (ii) fail to actively operate your Restaurant for three (3) or more consecutive days, or (iii) otherwise abandon or appear to have abandoned your rights under this Agreement; (d) you lose any license required to operate the Restaurant or you lose your right to occupy the Restaurant premises; (e) you or any Owner or Restaurant Manager are convicted of, or plead no contest to, a felony, a crime involving moral turpitude, or any other crime or offense that we believe is reasonably likely to have an adverse effect on your Restaurant or the System; (f) there is any Transfer or attempted Transfer in violation of Article 12. of this Agreement; (g) you or any Owner fails to comply with the confidentiality or noncompetition covenants in Section 15.1 of this Agreement; (h) you or any Owner has made any material misrepresentations in the Application Materials or otherwise in connection with your franchise application; (i) you fail to comply with notification requirements concerning investigations and Crisis Management Events; (j) you understate any payment to us by 3% or more, or understate any such payment in any amount, twice in any two-year period; (k) an imminent threat or danger to public health or safety results from the operation of the Restaurant; (1) you maintain false books or records or submit any false reports or statements to us; (m) you offer unauthorized products or services from the Restaurant premises or in conjunction with the Marks or Copyrighted Works; (n) you purchase Items for which we have identified a Designated Supplier from an unapproved source; (o) you fail to pass two or more quality assurance inspections within any rolling 12-month period; (p) you violate our policies for Restaurant operations, without authorization or permission, on two or more occasions, within any rolling 12-month period, including failure to accurately account for all Gross Sales through the Computer System; (q) you fail to participate in any advertising or marketing program pursuant to Sections 9.6, 9.7, or 9.8, or fail to maintain required hours of operation on two or more occasions without prior written permission, within any rolling 12-month period; (r) you violate any Dram Shop laws; or (t) we deliver to you three or more written notices of default pursuant to this Article 13 within any rolling 12-month period, whether or not the defaults described in such notices ultimately are cured.
- 13.3. <u>Termination With 10-Day Cure Period.</u> We have the right to terminate this Agreement, which termination will become effective upon delivery of written notice of termination, if you fail to cure the following defaults within 10 days after delivery of written notice: (a) failure to obtain or maintain required

insurance coverage; (b) failure to pay any amounts due to us or our Affiliates when due; (c) failure to pay any amounts due to the landlord of the Restaurant's premises, your suppliers, or other trade creditors (unless such amount is subject to a bona fide dispute); (d) failure to pay any amounts for which we have advanced funds for or on your behalf, or upon which we are acting as guarantor of your obligations; (e) violation of any provision of this Agreement concerning the use and protection of the Marks or Copyrighted Works; or (f) violation of any provision of this Agreement concerning the preparation, service, appearance, or quality of products or services.

- 13.4. <u>Termination With 30-Day Cure Period.</u> Except as otherwise provided in this Article 13., we have the right to terminate this Agreement, which termination will become effective upon delivery of written notice of termination, if you fail to cure any other default not specified in Sections 13.1, 13.2, or 13.3 within 30 days after delivery of written notice.
- 13.5. <u>Cross-Default.</u> Any default under any agreement between you or your Affiliates and us or our Affiliates, which you fail to cure within any applicable cure period, will be considered a default under this Agreement and will provide an independent basis for termination of this Agreement.
- 13.6. <u>Additional Remedies.</u> In addition to, or in lieu of, termination of this Agreement, in our sole discretion, we may require the Restaurant to be closed during any cure period relating to a default based on public health and safety concerns.

14. OBLIGATIONS UPON TERMINATION OR EXPIRATION

- 14.1. Cease Use of Intellectual Property. Upon termination or expiration of this Agreement, you shall immediately cease all use of the Marks, Copyrighted Works, and Confidential Information. You shall cancel any assumed name registration containing the Marks. You shall, at our option and request, assign to us all rights to all email addresses, URLs, domain names, Internet listings, business directory listing, and Internet accounts related to the Restaurant. You hereby appoint us as your attorney-in-fact with full power and authority for the sole purpose of assigning these rights to us. This appointment will be deemed to be coupled with an interest and will continue in full force and effect until the termination or expiration of this Agreement.
- 14.2. Assignment of Lease; De-Identification. At our request, you shall assign to us or our designee your interest in the lease, including your interest in all leasehold improvements, without additional compensation. If we do not request assignment of the lease before or as of the date of expiration or termination of this Agreement, then within 10 days after termination or expiration of this Agreement, you shall modify the Restaurant premises (including the changing of the color scheme and other distinctive design features, and the changing of and assigning to us of, the telephone numbers) as may be necessary to distinguish the appearance of the Restaurant from that of other Rusty Taco Restaurants, and shall make such specific additional changes to the Restaurant as we may reasonably request for that purpose. We shall also have the right, at our option and at your expense, to enter the Restaurant premises and take all actions necessary to de-identify the premises as a Rusty Taco Restaurant, including, but not limited to, removing all signs, advertising materials, displays, proprietary equipment and inventory, and any other items which display the Marks or are indicative of trade dress or Rusty Taco Restaurants. Such costs incurred due to our de-identification efforts must be paid by you immediately upon notice.
- 14.3. **Return of Operations Manual and Other Confidential Information.** You shall immediately deliver to us any hard copies or downloaded electronic copies of the Operations Manual and all other manuals, records, correspondence, files, and any instructions containing Confidential Information relating to the operation of the Restaurant that are in your possession; and all copies thereof (all of which are acknowledged to be our property).

- 14.4. Pay Accrued Amounts. You and your Owners must promptly pay all sums owing to us and our Affiliates. Such sums include all damages (including Lost Revenue Damages as described in Section 14.7 below), costs, and expenses, including reasonable attorneys' fees, incurred by us as a result of any default by you, which obligation will give rise to and will remain, until paid in full, a lien in our favor against any and all of the personal property, furnishings, equipment, signs, fixtures, and inventory owned by you and used by you at the Restaurant during the Term of this Agreement.
- 14.5. Our Right to Purchase Tangible Assets. We have the option to purchase any or all of the Restaurant's Items and other furniture, fixtures, equipment, and interior and exterior signs at the lesser of your cost or then-current fair market value and may set off against the purchase price any amounts that you owe us. We shall exercise our option by written notice to you, delivered within 10 days following the date of expiration or termination of this Agreement.
- 14.6. **Trademark Infringement.** If you fail to comply with a written notice of termination sent by us and a court later upholds such termination of this Agreement, if such termination was disputed by you, your operation of the Restaurant, from and after the date of termination stated in such notice, will constitute willful trademark infringement and unfair competition by you, and you shall be liable to us for damages resulting from such knowing and intentional infringement in addition to any fees paid or payable hereunder, including any profits which you derived from such post termination operation of the Restaurant.
- 14.7. Lost Revenue Damages. You agree that we will suffer compensable damages including, among others, the amount of the Royalty Fee and Fund contributions we would have received, and for which we bargained in entering into this Agreement, if you terminate this Agreement without cause or we terminate this Agreement because of your breach (the "Lost Revenue Damages"). You and we acknowledge that, because Royalty Fee and Fund contributions are calculated primarily as a percentage of the Restaurant's Gross Sales, it will be impossible to calculate Lost Revenue Damages once the Restaurant ceases operation. To bring certainty to the actual amount of Lost Revenue Damages, you and we agree that Lost Revenue Damages will equal the net present value of: (1) the lesser of 36 or the number of calendar months remaining on the Term absent the termination, multiplied by (2) the sum of the Royalty Fee and Fund contribution percentages in effect as of the termination date, multiplied by (3) the average monthly Gross Sales of the Restaurant during the 24 full calendar months immediately preceding the earlier of the termination date of this Agreement or the last date on which your Restaurant operated in compliance with this Agreement, minus (4) any cost savings we experienced as a result of the termination; provided, however, that if, as of the termination date of this Agreement or the last date on which your Restaurant operated in compliance with this Agreement, your Restaurant had not commenced operations or had operated for less than 24 months, the average monthly Gross Sales will equal the 24-month average Gross Sales of all Rusty Taco Restaurants that had operated for the full 24 calendar months immediately preceding termination of this Agreement.

You and we acknowledge and agree that (a) our agreement on the calculation of Lost Revenue Damages is a reasonable determination of actual damages we will suffer in the event of a termination as described above and is not a penalty, and (b) Lost Revenue Damages represent only lost Royalty Fees and Fund contributions, and the right to recover such damages is not exclusive of and does not replace any other rights we have under this Agreement or Applicable Laws if this Agreement is terminated as described above, including the right to seek other damages we suffer as a result of a termination as described above or the events on which such termination was based.

15. **COVENANTS**

- 15.1. Noncompetition During Term of Agreement. You and each Owner acknowledge that you and each Owner will receive access to customers of Rusty Taco Restaurants and valuable specialized training and Confidential Information, including information regarding the operational, sales, promotional, and marketing methods and techniques and trade secrets of pertaining to the operation of Rusty Taco Restaurants and the System. You agree that neither you nor any of your Owners, your or your Owners' Affiliates, or the officers, directors, managers or immediate family members of any of the foregoing will either directly or indirectly, for yourselves, or through, on behalf of, or in conjunction with any person, or legal entity:
 - 15.1.1. Divert or attempt to divert any present or prospective customer of a Rusty Taco Restaurant to any competitor, by direct or indirect inducement or otherwise, or do or perform, directly or indirectly, any other act injurious or prejudicial to the goodwill associated with the Marks and the System.
 - 15.1.2. Own, maintain, advise, operate, engage in, be employed by, make loans to, lease property to, invest in, provide any assistance to, or have any interest in (as owner or otherwise) or relationship or association with, a Competitive Business (except that equity ownership of less than five percent (5%) of a Competitive Business whose stock or other forms of ownership interest are publicly traded on a recognized United States stock exchange will not be deemed to violate this subparagraph) (i) anywhere in the world, or (ii) at any location within the United States, its territories or commonwealths, or any other country, province, state, or geographic area in which we or our Affiliates have used, sought registration of, or registered the Marks or similar marks or operate or license others to operate a business under the Marks or similar marks.
- 15.2. Noncompetition After Expiration or Termination of Agreement. Commencing on the later of: (a) a Transfer permitted under Article 12. of this Agreement, expiration of this Agreement (without renewal), or termination of this Agreement (regardless of the cause for termination); or (b) a final court order (after all appeals have been taken) with respect to any of the foregoing events or with respect to enforcement of this Section 15.2, and continuing for an uninterrupted period of two (2) years thereafter, you shall not either directly or indirectly, for yourselves, or through, on behalf of, or in conjunction with any person, persons, or legal entity, own, maintain, advise, operate, engage in, be employed by, make loans to, lease property to, invest in, provide any assistance to, or have any interest in (as an owner or otherwise) or relationship or association with any Competitive Business that: (i) is, or is intended to be, located at the Location; (ii) is within a five-mile radius of the Location; or (iii) is within a five-mile radius of any other Rusty Taco Restaurant operating under the System and Marks, that is in existence or under development at any location within the United States, its territories, or commonwealths, or any other country, province, state, or geographic area in which we or our Affiliates have used, sought registration of, or registered the Marks or similar marks, or operate or license others to operate a business under the Marks or similar marks, at the time of such expiration, termination, or Transfer. If any Owner ceases to be an Owner of the Franchisee for any reason during the Term of this Agreement, the foregoing covenant will apply to the departing Owner for a two-year period beginning on the date such person ceases to meet the definition of an Owner. The time period during which the post-term non-competition obligations in this Section apply will be tolled for any period of noncompliance.
- 15.3. Additional Provisions. You agree that we have the right, in our sole discretion, to reduce the scope of any covenant set forth in Sections 15.1 and 15.2, or any portion thereof, without your consent or the consent of any Owner, effective immediately upon delivery of written notice to the affected party; and you and each Owner agree that you and such person shall comply forthwith with any covenant as so modified. You and each Owner expressly agree that the existence of any claims you may have against us, whether or not arising from this Agreement, will not constitute a defense to our enforcement of the covenants in this Article 15. You agree to pay all costs and expenses (including reasonable attorneys'

fees) we incur in connection with the enforcement of this Article 15.

- 15.4. <u>Covenants from Individuals.</u> Unless prohibited by Applicable Laws, each individual who attends our training program, has access to our Confidential Information, or whom we otherwise designate, must sign a confidentiality agreement in a form we approve.
- 15.5. **Breach of Covenants Causes Irreparable Injury.** You acknowledge that the violation of any covenant of this Article 15. would result in irreparable injury to us for which no adequate remedy at law may be available, and you consent to the issuance of, and agree to pay all court costs and reasonable attorneys' fees incurred by us in obtaining, without the posting of any bond, an *ex parte* or other order for injunctive or other legal or equitable relief with respect to such conduct or action.
- 15.6. **Exception for Publicly Held Companies.** The foregoing restrictions will not apply to your ownership or any Owner's ownership of less than a 5% beneficial interest in the outstanding equity securities of any company registered under the Securities Act of 1933 or the Securities Exchange Act of 1934.

16. **REPRESENTATIONS**

- 16.1. Our Representations. We represent and warrant that: (a) we are duly organized and validly existing under the law of the state of our formation; (b) we are duly qualified and authorized to do business in each jurisdiction in which our business activities or the nature of the properties we own requires such qualification; and (c) the execution of this Agreement and the performance of the transactions contemplated by this Agreement are within our corporate power and have been duly authorized.
- 16.2. **Representations of Franchisee.** You represent and warrant to us that:
 - 16.2.1. the ownership information set forth in <u>Attachment A</u> is accurate and complete in all material respects, and you will notify us in writing prior to any change in such information;
 - 16.2.2. (a) you are duly organized and validly existing under the law of the state of your formation; (b) you are duly qualified and authorized to do business in each jurisdiction in which your business activities or the nature of the properties you own require such qualification; (c) your organizational charter and governing documents will at all times provide that your activities are confined exclusively to the operation of the Restaurant; (d) neither you nor any of your Affiliates or Owners own, operate, or have any financial or beneficial interest in any business in Competitive Business; and (e) the execution of this Agreement and the performance of the transactions contemplated by this Agreement are permitted under your charter and governing documents; and
 - 16.2.3. neither you nor your Owners, officers, directors, managers, partners, agents, or employees, or their respective interests therein is now and will not be, during the Term, (a) identified, either by name or an alias, pseudonym, or nickname, on any of the lists of restricted or denied parties maintained by the United States government, or (b) in violation of any law prohibiting money laundering or the aid or support of persons who conspire to commit acts of terror against any person or government, including acts prohibited by the U.S. Patriot Act (http://epic.org/privacy/terrorism/hr3162.html), U.S. Executive Order 13224 (http://www.treasury.gov/resource-center/sanctions/Programs/Documents/terror.pdf), or any similar law.

17. **NOTICES**

17.1. Notices. All written notices, reports, and payments permitted or required to be delivered by this Agreement or the Operations Manual will be deemed to be delivered on the earlier of the date of actual delivery or one of the following: (i) at the time delivered by hand, (ii) at the time delivered to the computer server in case of electronic transmission and, in the case of the Royalty Fees, Fund contributions, and other amounts due, at the time we actually receive electronic payment, or (iii) one (1) business day after being placed in the hands of a nationally recognized commercial courier service for next business day delivery. Notices must be sent to you as shown on the attached Data Sheet or the most current principal business address of which the notifying party has notice; except that, it will always be deemed acceptable during the Term to send notices to you at the address of the Restaurant. Notices to us must be sent to the address shown on the first page of this Agreement, Attention: President, with a copy (which shall not constitute notice) to Legal Department.

18. **CONSTRUCTION OF AGREEMENT**

18.1. Entire Agreement. This Agreement, and any other agreements executed by the parties concurrently with the parties' execution of this Agreement, represents the entire fully integrated agreement between the parties and supersedes all other negotiations, agreements, representations, and covenants, oral or written. Notwithstanding the foregoing, nothing in this Agreement will disclaim or require you to waive reliance on any representation that we made in the franchise disclosure document (including its exhibits and amendments) that we delivered to you in connection with this franchise offering. Except for those changes permitted to be made unilaterally by us hereunder, no amendment, change, or variance from this Agreement will be binding on either party unless mutually agreed to in writing by the parties and executed by their authorized officers or agents.

The following provision applies if you or the franchise granted hereby are subject to the franchise registration or disclosure laws in California, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Virginia, or Wisconsin: No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

- 18.2. **No Waiver.** No delay, waiver, omission, or forbearance on our part to exercise any right, options, duty, or power arising out of any breach or default by you or any of your Owners under this Agreement will constitute a waiver by us to enforce any such right, option, duty, or power against you or your Owners, or as to a subsequent breach or default by you or any of your Owners.
- 18.3. **Severability.** The invalidity or unenforceability of any provision of this Agreement will not affect the validity or enforceability of any other provision of this Agreement.
- 18.4. <u>Survival of Terms.</u> Any provision or covenant of this Agreement which expressly or by its nature imposes obligations beyond the expiration or termination of this Agreement will survive such expiration or termination.
- 18.5. **Definitions and Captions.** Unless otherwise defined in the body of this Agreement, capitalized terms have the meaning ascribed to them in <u>Attachment B</u> ("Additional Definitions"). All captions in this Agreement are intended for the convenience of the parties, and none will be deemed to affect the meaning or construction of any provision of this Agreement.

- 18.6. **Persons Bound.** This Agreement will be binding on the parties and their respective successors and assigns. Each Owner shall execute our then-current form of Guaranty and Personal Undertaking. Our current form is attached as <u>Attachment C</u>. Failure or refusal to do so will constitute a breach of this Agreement. You and each Owner shall be joint and severally liable for each person's obligations hereunder and under the Guaranty and Personal Undertaking.
- 18.7. **Rules of Construction.** Neither this Agreement nor any uncertainty or ambiguity in this Agreement shall be construed or resolved against the drafter of this Agreement, whether under any rule of construction or otherwise. Terms used in this Agreement shall be construed and interpreted according to their ordinary meaning. If any provision of this Agreement is susceptible to two or more meanings, one of which would render the provision enforceable and the other(s) which would render the provision unenforceable, the provision will be given the meaning that renders it enforceable.
- 18.8. **Business Judgment.** Whenever we reserve discretion in a particular area or where we agree to exercise our rights reasonably or in good faith, we will satisfy such obligations whenever we exercise reasonable business judgment in making our decision or exercising our rights. Our decisions or actions will be deemed to be the result of reasonable business judgment, even if other reasonable or even arguably preferable alternatives are available.
- 18.9. <u>Counterparts.</u> This Agreement may be executed in one or more counterparts, each of which will be deemed an original, but all of which together will constitute one agreement.
- 18.10. Remedies Cumulative. All rights and remedies of the parties to this Agreement are cumulative and not alternative or in addition to and are not exclusive of any other rights or remedies which are provided for herein or which may be available at law or in equity in case of any breach, failure, or default or threatened breach, failure, or default of any term, provision, or condition of this Agreement or any other agreement between you and any of your Affiliates and us and any of our Affiliates. The rights and remedies of the parties to this Agreement will be continuing and will not be exhausted by any one or more uses thereof and may be exercised at any time or from time to time as often as may be expedient; and any option or election to enforce any such right or remedy may be exercised or taken at any time and from time to time. The expiration, earlier termination, or exercise of our rights pursuant to Article 13 of this Agreement will not discharge or release you or any of your Owners from any liability or obligation then accrued, or any liability or obligation continuing beyond, or arising out of, the expiration, the earlier termination, or the exercise of such rights under this Agreement. Additionally, you and your Owners will pay all courts costs and reasonable attorneys' fees and costs we incur in obtaining any remedy available to us for any violation of this Agreement.
- 18.11. **No Third-Party Beneficiary.** Except as expressly provided to the contrary herein, nothing in this Agreement is intended, nor will be deemed, to confer rights to or upon any person or legal entity who is not a party to this Agreement.
- 18.12. No Liability for Related Parties. You agree that none of our past, present, or future directors, officers, employees, incorporators, members, partners, stockholders, subsidiaries, Affiliates, controlling parties, entities under common control, ownership, or management, vendors, service providers, agents, attorneys, or representatives will have or will have any liability for: (a) any of our obligations or liabilities relating to or arising from this Agreement; (b) any claim against us based on, in respect of, or by reason of, the relationship between you and us; or (c) any claim against us based on any of our alleged, unlawful acts or omissions.

19. APPLICABLE LAWS; DISPUTES; ARBITRATION

- 19.1. Choice of Law. Except to the extent governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. Sections 1051 *et seq.*) or other federal law, this Agreement will be governed by and construed in accordance with the laws of the State of Delaware without regard to its conflict of law rules, except that any law regulating the offer or sale of franchises, business opportunities, or similar interests or governing the relationship between you and us will not apply unless its jurisdictional requirements are met independently without reference to this Section.
- 19.2. **Requirement to Submit Disputes to Arbitration.** We and you agree that all controversies, disputes, or claims between us or any of our Affiliates, and our and their respective owners, officers, directors, agents, and employees, on the one hand, and you (and your owners, guarantors, Affiliates, and employees), on the other hand, arising out of or related to:
 - (a) this Agreement or any other agreement between you (or any of your owners) and us (or any of our Affiliates);
 - (b) our relationship with you;
 - (c) the scope or validity of this Agreement or any other agreement between you (or any of your owners) and us (or any of our Affiliates) or any provision of any of such agreements (including the validity and scope of the arbitration provision under this Section 19.3, which we and you acknowledge is to be determined by an arbitrator, not a court); or
 - (d) any Standard, must be submitted for binding arbitration, on demand of either party, to the American Arbitration Association (the "AAA"). The arbitration proceedings will be conducted by one arbitrator and, except as this Section otherwise provides, according to the AAA's then-current Commercial Arbitration Rules. All proceedings will be conducted at a suitable location chosen by the arbitrator that is within 50 miles of our or, as applicable, our successor's or assign's then-current principal place of business (currently, Plano, Texas). All matters relating to arbitration will be governed by the Federal Arbitration Act (9 U.S.C. §§ 1 et seq.). The interim and final awards of the arbitrator shall be final and binding upon each party and may be enforced in any court of competent jurisdiction.

The arbitrator has the right to award or include in an award any relief which he or she deems proper, including money damages, pre- and post-award interest, interim costs and attorneys' fees, specific performance, and injunctive relief, provided that the arbitrator may not declare any of the trademarks owned by us or our affiliates generic or otherwise invalid, or award any punitive or exemplary damages against any party to the arbitration proceeding (we and you hereby waiving to the fullest extent permitted by law any such right to or claim for any punitive or exemplary damages against any party to the arbitration proceeding). In any arbitration brought pursuant to this Section 19.2, and in any action in which a party seeks to enforce compliance with this provision, the prevailing party shall be awarded its costs and expenses, including attorneys' fees, incurred in connection therewith. We and you agree to be bound by the provisions of any applicable contractual or statutory limitations provision, whichever expires earlier. We and you further agree that, in any arbitration proceeding, each party must submit or file any claim which would constitute a compulsory counterclaim (as defined by Rule 13 of the Federal Rules of Civil Procedure) within the same proceeding. Any claim which is not submitted or filed as required will be forever barred. The arbitrator may not consider any settlement discussions or offers that might have been made by either you or us.

ARBITRATION PROCEEDINGS WILL BE CONDUCTED ON AN INDIVIDUAL BASIS. NO ARBITRATION PROCEEDING MAY BE: (I) CONDUCTED ON A CLASS-WIDE BASIS, (II)

COMMENCED, CONDUCTED OR CONSOLIDATED WITH ANY OTHER ARBITRATION PROCEEDING, (III) JOINED WITH ANY SEPARATE CLAIM OF AN UNAFFILIATED THIRD-PARTY, OR (IV) BROUGHT ON BEHALF OF ANY PARTY BY ANY ASSOCIATION OR AGENT. Notwithstanding the foregoing, if any court or arbitrator determines that all or any part of the preceding sentence is unenforceable with respect to a dispute, controversy or claim that otherwise would be subject to arbitration under this Section, then all parties agree that this arbitration clause shall not apply to that dispute, controversy or claim and that such dispute, controversy or claim shall be resolved in a judicial proceeding in accordance with the dispute resolution provisions of the Agreements.

We and you agree that, in any arbitration arising as described in this Section, the arbitrator shall have full authority to manage any necessary exchange of information among the parties with a view to achieving an efficient and economical resolution of the dispute. The parties may only serve reasonable requests for documents, which must be limited to documents upon which a party intends to rely or documents that are directly relevant and material to a significant disputed issue in the case or to the case's outcome. The document requests shall be restricted in terms of time frame, subject matter and persons or entities to which the requests pertain and shall not include broad phraseology such as "all documents directly or indirectly related to." You and we further agree that no interrogatories or requests to admit shall be propounded, unless the parties later mutually agree to their use.

The provisions of this Section are intended to benefit and bind certain third-party non-signatories.

The provisions of this Section will continue in full force and effect subsequent to and notwithstanding the expiration or termination of the Agreement.

Any provisions of this Article 19 that pertain to judicial proceedings shall be subject to the agreement to arbitrate contained in this Section.

- 19.3. **Non-Exclusivity of Remedy.** No right or remedy conferred upon or reserved to us or you by this Agreement is intended to be, nor will be deemed, exclusive of any other right or remedy in this Agreement or by law or equity provided or permitted, but each will be cumulative of every other right or remedy.
- 19.4. Waiver of Jury Trial. WE AND YOU IRREVOCABLY WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING, OR COUNTERCLAIM, WHETHER AT LAW OR IN EQUITY, BROUGHT BY EITHER OF US AGAINST THE OTHER, WHETHER OR NOT THERE ARE OTHER PARTIES IN SUCH ACTION OR PROCEEDING.
- 19.5. Waiver of Punitive Damages. EXCEPT FOR YOUR OBLIGATION TO INDEMNIFY US FOR THIRD PARTY CLAIMS UNDER SECTION 11.3, THE PARTIES HEREBY WAIVE TO THE FULLEST EXTENT PERMITTED BY LAW ANY RIGHT TO OR CLAIM OF ANY PUNITIVE OR EXEMPLARY DAMAGES AGAINST THE OTHER AND AGREE THAT IN THE EVENT OF A DISPUTE BETWEEN THEM EACH SHALL BE LIMITED TO THE RECOVERY OF ANY ACTUAL DAMAGES SUSTAINED BY IT.
- 19.6. **Right to Injunctive Relief.** Nothing contained in this Agreement will bar our right to seek injunctive relief from any court of competent jurisdiction; and you agree to pay all costs and reasonable attorneys' fees we incur in obtaining such relief.
- 19.7. <u>Attorneys' Fees.</u> If either party commences a legal action (including an arbitration) against the other party arising out of or in connection with this Agreement, the prevailing party shall be entitled to have and recover from the other party its reasonable attorneys' fees and costs of suit.

Limitation of Claims & Class Action Bar. You and your owners agree not to bring any claim asserting 19.8. that any of the Marks are generic or otherwise invalid. EXCEPT FOR CLAIMS ARISING FROM YOUR NON-PAYMENT OR UNDERPAYMENT OF AMOUNTS YOU OWE US. ANY AND ALL CLAIMS ARISING OUT OF OR RELATING TO THE AGREEMENT OR OUR RELATIONSHIP WITH YOU WILL BE BARRED UNLESS A JUDICIAL OR ARBITRATION PROCEEDING IS COMMENCED IN ACCORDANCE WITH THE AGREEMENT WITHIN ONE (1) YEAR FROM THE DATE ON WHICH THE PARTY ASSERTING THE CLAIM KNEW OR SHOULD HAVE KNOWN OF THE FACTS GIVING RISE TO THE CLAIMS. The parties understand that such time limit might be shorter than otherwise allowed by law. You and your owners agree that your and their sole recourse for claims arising between the parties shall be against us or our successors and assigns. You and your owners agree that our and our Affiliates' members, managers, shareholders, directors, officers, employees, and agents shall not be personally liable nor named as a party in any action between us or our Affiliates and you or your owners. No previous course of dealing shall be admissible to explain, modify, or contradict the terms of the Agreement. No implied covenant of good faith and fair dealing shall be used to alter the express terms of the Agreement.

WE AND YOU AGREE THAT ANY PROCEEDING WILL BE CONDUCTED ON AN INDIVIDUAL BASIS AND THAT ANY PROCEEDING BETWEEN US AND ANY OF OUR AFFILIATES, OR OUR AND THEIR RESPECTIVE SHAREHOLDERS, OFFICERS, DIRECTORS, AGENTS, AND EMPLOYEES, ON THE ONE HAND, AND YOU (OR YOUR OWNERS, GUARANTORS, AFFILIATES, AND EMPLOYEES), ON THE OTHER HAND, MAY NOT BE: (I) CONDUCTED ON A CLASS-WIDE BASIS, (II) COMMENCED, CONDUCTED OR CONSOLIDATED WITH ANY OTHER PROCEEDING, (III) JOINED WITH ANY CLAIM OF AN UNAFFILIATED THIRD-PARTY, OR (IV) BROUGHT ON YOUR BEHALF BY ANY ASSOCIATION OR AGENT.

19.9. Consent to Jurisdiction. Subject to Section 19.2 above and the provisions below, we and you agree that all controversies, disputes, or claims between us or any of our affiliates (and our and their respective shareholders, officers, directors, agents, and employees), on the one hand, and you (and your owners, guarantors, affiliates, and employees), on the other hand, arising out of or related to this Agreement or any other agreement between you (or any of your owners) and us (or any of our affiliates) or our relationship with you must be commenced exclusively in state or federal court closest to our (or our successor's or assign's, as applicable) then-current principal place of business (currently, Plano, Texas), and the parties irrevocably consent to the jurisdiction of those courts and waive any objection to either the jurisdiction of or venue in those courts. Nonetheless, the parties agree that any of us may enforce any arbitration orders and awards in the courts of the state or states in which you are or the Restaurants is located.

[Signature Page to Follow]

IN WITNESS WHEREOF, the parties have executed and delivered this Agreement effective on the Effective Date.

RUSTY TACO FRANCHISING, LLC

[FRANCHISEE]

By:	By:
Name:	Name:
Title:	Title:
Date*:	Date:
*This is the Effective Date	

ATTACHMENT A DATA SHEET

1.	Effective Date o	of Agreement:		
2.	Franchisee:			
Name	<u> </u>			
Addre	ess:			
Atten				
	Address:			
Phone				
	of Entity: of Formation:			
	of Formation:			
	ting Principal:			
3.	Franchisee Own	ers [Section 16.2.1]:	T 6 I4	D4 II.11
Name	2	Address	Type of Interest	Percentage Held
4.	Initial Franchise	Fee [Section 4.1]: \$		
5.	Brand Developm	ment Fund Contribution (percentage	of Gross Sales) [Section 9	0.3]:
6. Initial Restaurant Manager:				
7.	Location Address	ss [Section 1.1]:		
8.	Site Selection A	rea [Section 3.1]: (check one):		
	□ N/A			
☐ Development Area identified in the Multi-Unit Development Agreement dated, between us and				
	☐ The area shown on the following map:			
		[insert map]		
9.	Protected Territo	ory [Section 1.2]: (check one):		
☐ Two-mile radius surrounding the Location with the front door of your Restaurant serving as the center point of the circle; or				your
	☐radius for Locations located in densely populated urban area; or			or
☐ No Protected Territory (Non-Traditional Location)				

10. Opening Date:				
11. <u>Additional Terms or Modifications to the Agreement</u> . The following terms, if any, supplement or amend the provisions of the Agreement and will control in the event of any conflicts:				
[insert as applicable]				
RUSTY TACO FRANCHISING, LLC	[FRANCHISEE]			
By:	Ву:			
Name:	Name:			
Title:	Title:			
Date*:	Date:			
*This is the Effective Date				

ATTACHMENT B ADDITIONAL DEFINITIONS

- "Affiliate" means an affiliate of a named person identified as any person or entity that Controls, is Controlled by, or is under common Control with such named person.
- "Agreement" means the Franchise Agreement.
- "Applicable Laws" means all federal, state and local laws, ordinances, regulations, and governmental orders related to the development or operation of your Restaurant.
- "Application Materials" means the information you and, if applicable, your owners provide to us to support your request for a Franchise including information about your and, if applicable, your owners' background, experience, skills, financial condition, and resources available to acquire a Franchise and to develop and operate a Rusty Taco Restaurant.
- "Business Day" means each day other than a Saturday, Sunday, U.S. holiday or any other day on which the Federal Reserve is not open for business in the United States.
- "Competitive Business" means any restaurant, food service or other business (other than a Rusty Taco Restaurant that is operating pursuant to a franchise agreement with us or our Affiliates) (i) deriving more than 10% of its revenue (excluding revenue derived from the sale of alcoholic beverages) from the sale of Mexican-inspired food, or (ii) that grants franchises or licenses for the operation of any of the foregoing or provides services to the franchisor or licensor of any of the foregoing.
- "Confidential Information" means all non-public information about the System and the operation of Rusty Taco Restaurants (including your Restaurant), some of which may constitute trade secrets under Applicable Law, regardless of whether it is marked "confidential," including the Standards (and other methods, formats, specifications, standards, systems, procedures, techniques, market research, customer data, knowledge, and experience used in developing, promoting and operating Rusty Taco Restaurants and the products they offer and sell), and other elements of the System; all customer information; all information contained in the Operations Manual; site selection criteria; growth and development plans, strategies and forecasts related to the System; training and operations materials; knowledge of and specifications for, and vendors of, Items and other products and supplies; any software or other technology which is proprietary to us, our Affiliates, or the System, including digital passwords and identifications and any source code of, and data, reports, and other materials generated by, the software or other technology; knowledge of the operating results and financial performance of Rusty Taco Restaurants (including your Restaurant). Our Confidential Information does not include Restricted Data.
- "Contact Identifiers" means each telephone or facsimile number, directory listing, and any other type of contact information used by or that identifies or is associated with your Restaurant.
- "Control" means the ability, either alone or as part of a group that has agreed to act in concert, to determine the outcome of decisions regarding the management or policies of an entity, whether by voting power, Applicable Laws, or authority granted in the entity's governing documents.
- "Controlling Interest" means: (a) if you are a corporation or a limited liability company, that the Owners, either individually or cumulatively (i) directly or indirectly own at least 51% of the shares of each class of the franchisee entity's issued and outstanding capital stock or membership units, as applicable; and (ii) are entitled, under its governing documents and under any agreements among the Owners, to cast a sufficient number of votes to require such entity to take or omit to take any action which such entity is required to take or omit to take under this Agreement; or (b) if you are a partnership, that the Owners (i) own at least 51% interest in the operating profits and operating losses of the partnership as well as at least 51% ownership interest in the partnership (and at least

51% interest in the shares of each class of capital stock of any corporate general partner); and (ii) are entitled under its partnership agreement or Applicable Laws to act on behalf of the partnership without the approval or consent of any other partner or be able to cast a sufficient number of votes to require the partnership to take or omit to take any action which the partnership is required to take or omit to take under this Agreement. Any interest less than 51% of the corporation, limited liability company, or partnership is a "Non-Controlling Interest."

"Copyrighted Works" means works of authorship which are owned by us or our Affiliates and fixed in a tangible medium of expression including the content of the Operations Manual, the design elements of the Marks, menus, bulletins, correspondence and communications with our franchisees, training, advertising and promotional materials, and the content and design of System Websites.

"Crisis Management Event" means any event that occurs at or about the Restaurant premises or in connection with the operation of the Restaurant that has or may cause harm or injury to customers or employees, such as food contamination, food spoilage/poisoning, food tampering/sabotage, contagious diseases, natural disasters, terrorist acts, shootings or other acts of violence, data breaches, real or threatened, or any other circumstance which may materially and adversely affect the System or the goodwill symbolized by the Marks.

"Development Agreement" means the form of agreement prescribed by us and used to grant to our developers the right to acquire multiple Franchises for the development of Rusty Taco Restaurants within a certain geographical area, including all attachments, exhibits, riders, guarantees, or other related documents, all as amended from time to time.

"Dram Shop Laws" means laws that regulate the sale of alcohol to minors and/or to those who have consumed or appear to have consumed alcohol in an amount above the legal limits.

"Franchise" means the right and license to develop, own and operate a Rusty Taco Restaurant pursuant to a Franchise Agreement.

"Franchise Agreement" means the form of agreement we use from time to time to grant to franchisees the right to own and operate a single Rusty Taco Restaurant, including all attachments, exhibits, riders, guarantees, or other related instruments, all as amended from time to time.

"Gross Sales" means the total revenue attributable to, and receipts from, all services and products and all income of every other kind and nature related to your Restaurant, whether or not in compliance with the Franchise Agreement, including income related to catering operations and special events and the full value of meals provided to your bona fide employees as a benefit of their employment (except you may deduct from Gross Sales the value of any employee discounts that are given during the week in which the meals are provided), whether for cash or credit, barter or exchange, and regardless of collection in the case of credit. Gross Sales does not include (a) sales (or other similar) taxes that you collect from your customers, if you transmit them to the appropriate taxing authority, promotions, voids, and discounts; (b) proceeds from isolated sales of trade fixtures that are not part of the products and services you offer and that do not have any material effect on the operation of your Restaurant; (c) tips or gratuities that Restaurant customers pay directly to your employees or to you which you then turn over to these employees in lieu of direct tips or gratuities; or (d) returns to shippers or manufacturers. Revenue from the purchase or redemption of gift certificates, gift cards, loyalty or similar programs is calculated as part of Gross Sales in accordance with our then-current guidelines for such programs. Gross Sales also includes all insurance proceeds you receive to replace revenue that you lose from the interruption of the operation of your Restaurant.

"Including" means including, without limitation.

"Incapacity" means physical, emotional, or mental injury or illness which would prevent a person from performing the obligations set forth in this Agreement or in the guaranty made part of this Agreement for at least 90 consecutive days and from which the person suffering the incapacity is not likely to recover within 90 the date of determination of the incapacity, as determined by a licensed practicing physician upon examination of the

person, or if the person refuses to submit to an examination, then such person automatically will be deemed to be incapacitated as of the date of such refusal.

- "Marks" means certain trade names, service marks, trademarks, logos, emblems, and indicia of origin, including, but not limited, to the mark "Rusty Taco®" and such other trade names, service marks, and trademarks as are now designated and which we may hereafter designate, in writing, for use in connection with the System.
- "Non-Traditional Location" means a physical location that (1) is part of a larger venue or facility, (2) is not generally and easily accessible to the general public, or (3) whose operating hours are limited to those imposed by the owner or operator of the venue in which it sits (for example, military bases, shopping malls, hotels, school and college campuses, airports, train stations, travel plazas, toll roads, beaches, parks and other seasonal facilities, government buildings and establishments, prisons, hospitals, convenience stores, trucks, casinos, sports or entertainment venues or stadiums, and retail shop locations being sublet under a lease to a master concessionaire).
- "Online Presence" means use of any Marks (a) on the Internet; (b) as part of any URL or domain name; (c) as part of a user name on any gaming website, social networking website, or discounting website; (d) in any unauthorized email address; (e) as part of any unauthorized software application; (f) on any branded merchandise, whether sold or given away; of (g) as part of any other online presence or presence on any electronic, virtual, or digital medium of any kind.
- "Opening Date" means the date on which your Restaurant generates revenue from operations whether or not in compliance with this Agreement. The Opening Date may be reflected in <u>Attachment A</u> or in any other form of written communication from us that defines the Opening Date.
- "Opening Deadline" means 120th day following the date on which you take or by which you were required to have taken possession of the Location.
- "Operations Manual" means the compilation of information and knowledge that is necessary and material to the System. The term "Manual," as used in this Agreement, includes all publications, materials, drawings, memoranda, videotapes, CDs, DVDs, MP3s, and other electronic media that we from time to time may loan or otherwise make available to you. The Manual may be supplemented or amended from time to time by letter, electronic mail, bulletin, videotape, CD, DVD, MP3, or other communications concerning the System to reflect changes in the image, specifications, and Standards relating to developing, equipping, furnishing, and operating a Rusty Taco® Restaurant.
- "Operating Principal" means the person identified as the operating principal on Attachment A.
- "Owner" means each individual or entity holding a beneficial ownership in the franchisee. It includes all shareholders of a corporation, all members of a limited liability company, all general and limited partners of a limited partnership, and the grantor and the trustee of the trust.
- "Proprietary Products" means recipes and menu items that incorporate our trade secrets and proprietary information and products and ingredients that are manufactured according to our proprietary specifications.
- "Protected Territory" means the area around your Restaurant as described on Attachment A hereto or, if your Restaurant is not approved when you and we sign this Agreement, they area that we describe in our notice of approval of your proposed Location.
- "Restaurant Manager" means the person who is in charge of the day-to-day, on-site operations of the Restaurant.
- "Restricted Data" means (a) any Personal Information of the employees, officers, contractors, owners or other personnel of you, your Affiliates, or the Restaurant; and (b) such other Personal Information as we from time to

time expressly designate as Restricted Data.

"Standards" means regulations and requirements we specify from time to time, which may regulate every aspect of the development and operation of a Rusty Taco Restaurant, including: specified business formats, methods, procedures, designs, layouts, standards, menus, ingredients, recipes, packaging materials, and other specifications; the geographical area in which you can provide catering and delivery services; sales, marketing, advertising and promotional programs and materials and media used in these programs; staffing levels and employee qualifications, training, dress and appearance (although employee selection and promotion, hours worked, rates of pay and other benefits, work assigned and working conditions are your sole responsibility); use and display of the Marks; the appearance of your Restaurant (including its interior, exterior and perimeter) and any vehicles used in the operation or promotion of your Restaurant; days and hours of operation; methods of payment that your Restaurant may accept from customers; participation in market research and product and service testing programs; participation in gift and loyalty card programs; menus, including product offerings, appearance, and inclusion of nutrition information; bookkeeping, accounting, data processing and record keeping systems and forms; formats, content and frequency of reports to us of sales, revenue, and financial performance and condition; participation in quality assurance and customer satisfaction programs; types, amounts, terms and conditions of insurance coverage required for your Restaurant, including criteria for your insurance carriers; approved music systems, wi-fi and other wireless Internet and communications systems, and interactive displays and screens that may be installed in the Restaurant premises; responses to and handling of Crisis Management Events; and any other aspects of operating and maintaining your Restaurant that we determine to be useful to preserve or enhance the efficient operation, image or goodwill of the Marks and Rusty Taco Restaurants.

"System" means the entire system of Rusty Taco Restaurants.

"**Term**" means, unless sooner terminated as provided in this Agreement, the period beginning on the Effective Date and ending on the 10th anniversary of the Opening Date.

"Transfer" means the appointment of a third-party manager to assume day-to-day responsibility for the operation or management of the Restaurant or the sale, assignment, transfer, conveyance, give-away, pledge, mortgage, or other disposition or encumber of any direct or indirect interest in the Agreement (including any or all of your rights or obligations under it), your Restaurant or its assets (other than in the ordinary course of business), your right to possession of the premises on which the Restaurant is located, or any direct or indirect ownership interest in you (regardless of its size).

RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT

ATTACHMENT C GUARANTY AND PERSONAL UNDERTAKING

1.	I have read the Franchise Agreement between Rusty Taco Franchising, LLC ("") and
	("Franchisee"), having an Effective Date of

- 2. I own a beneficial interest in Franchisee and would be considered an "Owner" within the definition contained in the Franchise Agreement.
- 3. I understand that were it not for this Guaranty and Personal Undertaking ("**Guaranty**"), Franchisor would not have agreed to enter into the Franchise Agreement with Franchisee.
- 4. I will comply with all of the provisions contained in Article 7. of the Franchise Agreement concerning the Franchisee's use of the Marks and Copyrighted Works (as those terms are defined in the Franchise Agreement). I understand that, except for the license granted to Franchisee, I have no individual right to use the Marks or Copyrighted Works, and I have no ownership interest in the Marks or Copyrighted Works.
- 5. I will comply with all of the provisions contained in Article 8. of the Franchise Agreement concerning the use of the Confidential Information. I will maintain the confidentiality of all Confidential Information disclosed to me. I agree to use the Confidential Information only for the purposes authorized under the Franchise Agreement. I further agree not to disclose any of the Confidential Information, except: (a) to Franchisee's employees on a need-to-know basis; (b) to Franchisee's and my legal and tax professionals to the extent necessary for me to meet my legal obligations; and (c) as otherwise may be required by law.
- 6. I will comply with all of the provisions contained in Article 12. of the Franchise Agreement concerning the transfer of my ownership interest in the Franchise Agreement and/or in Franchisee.
- 7. While I am an Owner of Franchisee and, for a two-year period after I cease to be an Owner (or two years after termination or expiration (without renewal) of the Franchise Agreement, whichever occurs first), I will not:
- (a) Divert or attempt to divert any present or prospective customer of a Rusty Taco® Restaurant to any competitor by direct or indirect inducement or otherwise, or do or perform, directly or indirectly, any other act injurious or prejudicial to the goodwill associated with the Marks and the System; or
- (b) Own, maintain, advise, operate, engage in, be employed by, make loans to, invest in, provide any assistance to, or have any interest in (as owner or otherwise) or relationship or association with, restaurant or other food service business deriving more than 10% of its revenue (excluding revenue derived from the sale of alcoholic beverages) from the sale of Mexican food, other than a Rusty Taco Restaurant operated pursuant to a then-currently effective Franchise Agreement. This restriction will apply, while I am an Owner everywhere in the world. This restriction will apply for two years after I cease to be an Owner (or two years after termination or expiration of the Franchise Agreement, whichever occurs first) to any location that: (i) is, or is intended to be, located at the location of the Restaurant; (ii) is within a five-mile radius of the Restaurant; or (iii) is within a five-mile radius of any other Rusty Taco Restaurant operating under the System and Marks, that is in existence or under development at any location within the United States, its territories, or commonwealths, or any other country, province, state, or geographic area in which Franchisor or its

Affiliates have used, sought registration of, or registered the Marks or similar marks, or operate or license others to operate a business under the Marks or similar marks, at the time of such expiration, termination, or transfer. The time period relating to the restrictions set forth in this Section will be tolled during any period of my noncompliance.

- 8. I agree that each of the foregoing covenants will be construed as independent of any other covenant or provision of this Guaranty. If all or any portion of a covenant in this Guaranty is held unreasonable or unenforceable by a court or agency having valid jurisdiction in an un-appealed final decision to which Franchisor is a party, I expressly agree to be bound by any lesser covenant subsumed within the terms of such covenant that imposes the maximum duty permitted by law, as if the resulting covenant were separately stated in and made a part of this Guaranty.
- 9. I understand and acknowledge that Franchisor shall have the right, in its sole discretion, to reduce the scope of any covenant set forth in this Guaranty, or any portion thereof, without my consent, effective immediately upon receipt by me of written notice thereof, and I agree to comply forthwith with any covenant as so modified.
- 10. I agree that the provisions contained in Article 19. of the Franchise Agreement will apply to any dispute arising out of or relating to this Guaranty. If Franchisor brings any legal action to enforce its rights under this Guaranty, I will reimburse Franchisor its reasonable attorneys' fees and costs.
- 11. I hereby guarantee the prompt and full payment of all amounts owed by the Franchisee under the Franchise Agreement.
- 12. I will pay all amounts due under this Guaranty within 14 days after receiving notice from Franchisor that the Franchisee has failed to make the required payment. I understand and agree that Franchisor need not exhaust its remedies against the Franchisee before seeking recovery from me under this Guaranty.
- 13. No modification, change, impairment, or suspension of any of Franchisor's rights or remedies will in any way affect any of my obligations under this Guaranty. If the Franchisee has pledged other security or if one or more other persons have personally guaranteed performance of the Franchisee's obligations, I agree that Franchisor's release of such security will not affect my liability under this Guaranty.
- 14. I WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING, OR COUNTERCLAIM, WHETHER AT LAW OR IN EQUITY, INVOLVING FRANCHISOR, WHICH ARISES OUT OF OR IS RELATED IN ANY WAY TO THE FRANCHISE AGREEMENT, THE PERFORMANCE OF ANY PARTY UNDER THE FRANCHISE AGREEMENT, AND/OR THE OFFER OR GRANT OF THE FRANCHISE.
- 15. I ACKNOWLEDGE AND AGREE THAT THE FRANCHISE AGREEMENT REQUIRES THAT ANY DISPUTES ARISING OUT OF THE FRANCHISE AGREEMENT BE SUBMITTED TO ARBITRATION AND THAT THOSE PROVISIONS ALSO APPLY TO DISPUTES REGARDING THIS GUARANTY.
- 16. I understand that Franchisor's rights under this Guaranty will be in addition to, and not in lieu of, any other rights or remedies available to Franchisor under Applicable Laws. I further understand and agree that my obligations under this Guaranty are joint and several as to other signatories to this Guaranty.
- 17. I agree that any notices required to be delivered to me will be deemed to be delivered on the earlier of the date of actual delivery or one of the following: (i) at the time delivered by hand, or (ii) one (1) business

day after being placed in the hands of a nationally recognized commercial courier service for next business day delivery. I may change my address only by delivering to Franchisor written notice of the change.

Intending to be legally bound, I have executed this Guaranty and Personal Undertaking on the date set forth below:

GUARANTOR	
	Date:
Guarantor's Name, an Individual	
Address:	_

RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT

ATTACHMENT D LEASE ADDENDUM

This Lease Addendum ("Addendum") is entered into by the undersigned Landlord and Tenant on the dates set forth below.

RECITALS

A.	The parties have entered into a Lease Agreement dated_	, 20, (" Lease ") for
the pr	remises located at	("Premises").
В.	Landlord acknowledges that Tenant has agreed to open	*
	nt's Franchise Agreement ("Franchise Agreement") with	•
limite	ed liability company ("Franchisor"), under the name "Rust	y Taco [®] " or other name Franchisor designates
("Res	staurant").	-

C. The parties desire to amend the Lease in accordance with the terms and conditions contained in this Addendum to provide Franchisor the opportunity to preserve the Premises as a Rusty Taco[®]-branded restaurant as provided herein.

AGREEMENT

Landlord and Tenant agree to amend the Lease as follows:

1. <u>Remodeling and Decor.</u> Landlord agrees that Tenant has the right to remodel, equip, paint, and decorate the interior of the Premises and to display such proprietary marks and signs on the interior and exterior of the Premises as Tenant is reasonably required to do pursuant to the Franchise Agreement and any successor Franchise Agreement under which Tenant may operate a Restaurant on the Premises. Any remodel of the building and/or its signs shall be subject to Landlord's prior and reasonable approval.

2. <u>Assignment by Tenant.</u>

- (a) Tenant does not have the right to sublease or assign the Lease to any third party without Franchisor's and Landlord's written approval.
- (b) So long as Tenant is in good standing under the Lease, Tenant has the right to assign all of its right, title, and interest in the Lease to Franchisor, its Affiliates, or its parent company during the term of the Lease, including any extensions or renewals, without first obtaining Landlord's consent. No assignment will be effective, however, until Franchisor or its designated Affiliate ("**Franchisor Entity**") gives Landlord written notice of its acceptance of the assignment. Franchisor will be responsible for the lease obligations incurred after the effective date of the assignment.
- (c) If a Franchisor Entity elects to assume the Lease, under this subparagraph or unilaterally assumes the lease as provided for in subparagraph 3(a) or 4, Landlord and Tenant agree that (i) Tenant will remain liable for the responsibilities and obligations, including amounts owed to Landlord, prior to the date of assignment and assumption, and (ii) the Franchisor Entity will have the right to sublease the Premises to another franchisee with Landlord's prior reasonable approval, provided the franchisee meets Franchisor's then-current standards and requirements for franchisees and agrees to operate the Restaurant as a Rusty Taco®-branded restaurant pursuant to a Franchise Agreement with Franchisor. Upon receipt by Landlord of an assumption agreement pursuant to which the assignee agrees to assume the Lease and to observe the terms, conditions, and agreements on the part of Tenant to be performed under the Lease, the Franchisor Entity shall thereupon be released from all liability as

tenant under the Lease from and after the date of assignment, without any need of a written acknowledgment of such release by Landlord.

3. Default and Notice.

- (a) Landlord shall send Franchisor copies of all notices of default it gives to Tenant concurrently with giving such notices to Tenant. If Tenant fails to cure any defaults within the period specified in the Lease, then Landlord shall promptly give Franchisor written notice thereof, specifying the defaults Tenant failed to cure. Franchisor has the right, but not the obligation, to unilaterally assume the Lease if Tenant fails to cure such defaults. Franchisor shall have 15 days from the date it receives such notice to exercise, by written notice to Landlord and Tenant, its right for Franchisor or a Franchisor Entity to assume the Lease. Franchisor shall have an additional 15 days from the expiration of Tenant's cure period in which to cure the default or violation.
- (b) All notices to Franchisor must be sent by registered or certified mail, postage prepaid, to the following address:

General Counsel Rusty Taco Franchising, LLC 5412 W. Plano Pkwy., Suite 100, Plano, Texas 75093

Franchisor may change its address for receiving notices by giving Landlord written notice of the new address. Landlord agrees that it will notify both Tenant and Franchisor of any change in Landlord's mailing address to which notices should be sent.

- 4. <u>Termination, Non-Renewal, Expiration.</u> If the Franchise Agreement is terminated for any reason during the term of the Lease or any extension thereof, a Franchisor Entity will have the right, but not the obligation, to unilaterally assume the Lease by giving Landlord written notice. Not more often than three times per year, Franchisor may request of Landlord, and Landlord shall, within 15 business days after being requested to do so, execute, acknowledge and deliver to Franchisor an instrument, certifying (a) that the Lease is unmodified and in full force and effect (or, if there has been any modification thereof, that it is in full force and effect as so modified, stating therein the nature of such modification); and (b) as to whether any notice of default has been given by either Landlord or Tenant, which default has not been cured; and acknowledging and agreeing that any statement contained in such certificate may be relied upon by Franchisor. Not more often than three times per year, Landlord may request of Franchisor, and Franchisor shall, within 15 business days after being requested to do so, execute, acknowledge, and deliver to Landlord an instrument, certifying (a) that the Franchise Agreement is in full force and effect, and (b) as to whether any notice of default has been given by either Franchisor or Tenant under the Franchise Agreement which default has not been cured; and acknowledging and agreeing that any statement contained in such certificate may be relied upon by Landlord.
- 5. Access to Premises Following Expiration or Termination of Lease. Upon the expiration or termination of the Lease, Landlord will cooperate with and assist Franchisor in gaining possession of the Premises, and if a Franchisor Entity does not elect to enter into a new lease for the Premises with Landlord on terms reasonably acceptable to the Franchisor Entity, Landlord will allow Franchisor to enter the Premises, without being guilty of trespass and without incurring any liability to Landlord, except for any damages caused by Franchisor's willful misconduct or gross negligence, to remove all signs, awnings, and all other items identifying the Premises as a Rusty Taco® Restaurant and to make such other modifications (such as repainting) as are reasonably necessary to protect the Rusty Taco® marks and system and repair any damage caused by such removal. In the event Franchisor exercises its option to purchase assets of Tenant, Landlord must permit Franchisor to remove all such assets being purchased by Franchisor.

6. Additional Provisions.

(a) Landlord hereby acknowledges that the provisions of this Addendum are required pursuant to the

Franchise Agreement under which Tenant plans to operate its business and the Tenant would not lease the Premises without this Addendum.

- (b) Landlord further acknowledges that Tenant is not an agent or employee of Franchisor and that Tenant has no authority or power to act for, or to create any liability on behalf of, or to in any way bind Franchisor or any Affiliate of Franchisor, and that Landlord has entered into this Addendum with full understanding that it creates no duties, obligations, or liabilities of or against Franchisor or any Affiliate of Franchisor, unless and until the Lease is assigned to, and accepted in writing by, Franchisor or a Franchisor Entity.
- (c) Franchisor Entity may elect not to assume or be bound by the terms of any amendment to the Lease executed by Tenant without obtaining Franchisor's prior written approval, which shall not be unreasonably withheld or delayed.
- 7. <u>Modification</u>. No amendment or variation of the terms of this Addendum is valid unless made in writing and signed by the parties and the parties have obtained the written consent of Franchisor.
- 8. <u>Reaffirmation of Lease</u>. Except as amended or modified in this Addendum, all of the terms, conditions and covenants of the Lease remain in full force and effect and are incorporated by reference and made a part of this Addendum as though copied herein in full. In the event of any conflict between the terms of this Addendum and those in the Lease, the terms of this Addendum shall control.
- 9. <u>Beneficiary</u>. Landlord and Tenant expressly agree that Franchisor is a third-party beneficiary of this Addendum.

By:	
By: Name:	
TENANT:	
TENANT: By: Name:	

LANDLORD:

EXHIBIT D

GENERAL RELEASE

RUSTY TACO FRANCHISING, LLC

GRANT OF FRANCHISOR CONSENT AND RELEASE

development agreement] dated, 20 (the " Agreement "). You have asked us to take the following action or to agree to the following request:
. We have the right under the Agreement to obtain a general release from you and your owners as a condition of taking this action or agreeing to this request. Therefore, we are willing to take the action or agree to the request specified above if you and your owners give us the release and covenant not to sue provided below in this document. You and your owners are willing to give us the release and covenant not to sue provided below as partial consideration for our willingness to take the action or agree to the request described above.

Consistent with the previous introduction, you, on your own behalf and on behalf of your successors, heirs, executors, administrators, personal representatives, agents, assigns, partners, owners, managers, directors, officers, principals, employees, and affiliated entities (collectively, the "Releasing Parties"), hereby forever release and discharge us and our current and former officers, directors, owners, managers, principals, employees, agents, representatives, affiliated entities, successors, and assigns (collectively, the "Franchisor Parties") from any and all claims, damages (known and unknown), demands, causes of action, suits, duties, liabilities, and agreements of any nature and kind (collectively, "Claims") that you and any of the other Releasing Parties now has, ever had, or, but for this document, hereafter would or could have against any of the Franchisor Parties, including without limitation, (1) arising out of or related to the Franchisor Parties' obligations under the Agreement, or (2) otherwise arising from or related to your and the other Releasing Parties' relationship, from the beginning of time to the date of your signature below, with any of the Franchisor Parties. You, on your own behalf and on behalf of the other Releasing Parties, further covenant not to sue any of the Franchisor Parties on any of the Claims released by this paragraph and represent that you have not assigned any of the Claims released by this paragraph to any individual or entity who is not bound by this paragraph.

IF THE FRANCHISE YOU OPERATE UNDER THE AGREEMENT IS LOCATED IN CALIFORNIA OR ANY OF THE RELEASING PARTIES IS A RESIDENT OF CALIFORNIA, THE FOLLOWING SHALL APPLY:

SECTION 1542 ACKNOWLEDGMENT. IT IS YOUR INTENTION, ON YOUR OWN BEHALF AND ON BEHALF OF THE RELEASING PARTIES, IN EXECUTING THIS RELEASE THAT THIS INSTRUMENT BE AND IS A GENERAL RELEASE WHICH SHALL BE EFFECTIVE AS A BAR TO EACH AND EVERY CLAIM, DEMAND, OR CAUSE OF ACTION RELEASED BY YOU OR THE RELEASING PARTIES. YOU RECOGNIZE THAT YOU OR THE RELEASING PARTIES MAY HAVE SOME CLAIM, DEMAND, OR CAUSE OF ACTION AGAINST ANY OF THE RELEASED PARTIES OF WHICH YOU, HE, SHE, OR IT IS TOTALLY UNAWARE AND UNSUSPECTING, WHICH YOU, HE, SHE, OR IT IS GIVING UP BY EXECUTING THIS RELEASE. IT IS YOUR INTENTION, ON YOUR OWN BEHALF AND ON BEHALF OF THE RELEASING PARTIES, IN EXECUTING THIS INSTRUMENT THAT IT WILL DEPRIVE YOU, HIM, HER, OR IT OF EACH SUCH CLAIM, DEMAND, OR CAUSE OF ACTION AND PREVENT YOU, HIM, HER, OR

1

IT FROM ASSERTING IT AGAINST ANY OF THE RELEASED PARTIES. IN FURTHERANCE OF THIS INTENTION, YOU, ON YOUR OWN BEHALF AND ON BEHALF OF THE RELEASING PARTIES, EXPRESSLY WAIVE ANY RIGHTS OR BENEFITS CONFERRED BY THE PROVISIONS OF SECTION 1542 OF THE CALIFORNIA CIVIL CODE, WHICH PROVIDES AS FOLLOWS:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY."

YOU ACKNOWLEDGE AND REPRESENT THAT YOU HAVE CONSULTED WITH LEGAL COUNSEL BEFORE EXECUTING THIS RELEASE AND THAT YOU UNDERSTAND ITS MEANING, INCLUDING THE EFFECT OF SECTION 1542 OF THE CALIFORNIA CIVIL CODE, AND EXPRESSLY CONSENT THAT THIS RELEASE SHALL BE GIVEN FULL FORCE AND EFFECT ACCORDING TO EACH AND ALL OF ITS EXPRESS TERMS AND PROVISIONS, INCLUDING, WITHOUT LIMITATION, THOSE RELATING TO THE RELEASE OF UNKNOWN AND UNSUSPECTED CLAIMS, DEMANDS, AND CAUSES OF ACTION.

If the franchise you operate under the Agreement is located in Maryland or if any of the Releasing Parties is a resident of Maryland, the following shall apply:

All representations requiring prospective franchisees to assent to a release, estoppel, or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland franchise registration and disclosure law.

We also are entitled to release and covenant not to sue from your owners. By his, her, or their separate signatures below, your owners likewise grant to us the release and covenant not to sue provided above.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the parties have executed and delivered this Agreement on the date stated below.

RUSTY TACO FRANCHISING, LLC a Delaware limited liability company	FRANCHISEE: (IF YOU ARE A CORPORATION, LIMITED LIABILITY COMPANY, OR PARTNERSHIP):		
By:			
Name:	By:		
Title:	Name:		
Dated:	Title:		
	(IF YOU ARE AN INDIVIDUAL AND NOT A LEGAL ENTITY; AND/OR ALL FRANCHISEE OWNERS):		
	Signature		
	Print Name		
	Signature		
	Print Name		

EXHIBIT E

LIST OF FRANCHISEES

List of Rusty Taco® Franchisees as of December 29, 2024

	State	City	Address	Zip	Phone #	Franchisee Entity
1.	AZ	Gilbert	1907 E. Williams Field Road, Suite 108	85295	480-272-8226	SONORAN RESTAURANT GROUP LLC
2.	AZ	Scottsdale	9290 E. Via De Ventura #105	85258	480-291-7020	SONORAN RESTAURANT GROUP #2, LLC
3.	AZ	Tempe	707 S Forest Avenue, Suite 112	85281	480-955-2955	SONORAN RESTAURANT GROUP #3, LLC
4.	СО	Brighton	2129 Barr Place	80603	720-603-9213	GOOD 2 GO EATS, LLC
5.	СО	Castle Rock	3991 Limelight Ave	80109	720-603-3492	GOOD 2 GO EATS, LLC
6.	СО	Dacono	107 Laura Ct	80514	720-603-9191	GOOD 2 GO EATS, LLC
7.	GA	Decatur	163 Clairmont Ave	30030	404-968-9236	MBC CONSESSIONS LLC
8.	GA	Summerhill	572 Hand Aaron Drive	30312	470-977-3206	MBC CONCESSIONS LLC
9.	ID	Ammon	250 S 45 th East	83401	208-535-5936	GOOD 2 GO EATS, LLC
10.	ID	Rexburg	1055 S 12 th West	83440	208-609-3492	GOOD 2 GO EATS, LLC
11.	IA	Dubuque	3333 Asbury Rd.	52002	563-287-2922	FIVE HORSEMEN, LLC
12.	MN	Inver Grove	522 East Hennepin Avenue	55414	612-315-5372	STEVE STENDER
13.	MN	Maple Grove	6346 Vinewood Lane North	55311	763-355-5064	RYAN JUNG
14.	MN	Minneapolis	522 East Hennepin Avenue	55414	612-315-5372	DAVE WAGMAN
15.	MN	Minneapolis	1005 4th Street South	55415	612-315-5372	ARAMARK SPORTS ENTERTAINMENT SERVICES
16.	NE	Lincoln	2600 Pine Lake Road, Suite 1	68516	402-975-2068	USTACO, INC.*
17.	NE	Omaha	14919 West Maple Road	68116	402-934-9990	NETACO, INC.*
18.	NE	Omaha	12995 West Center Road	68144	402-932-6936	NETACO, INC.*
19.	NE	Elkhorn	5331 204 th St., Suite 1	68022	531-710-4476	NETACO, INC.*
20.	NM	Albuquerque	6600 Menaul Blvd., Suite T-002	87110	505-313-8319	JJ&D ENTERPRISES, LLC*
21.	ОН	Beavercreek	2760 Towne Drive	45431	937-306-8080	NORTHWITT, LLC*
22.	ОН	Dayton	1822 Brown Street	45409	937-938-7384	NORTHWITT, LLC*
23.	ОН	Maumee	1470 Ford Street	43537	419-887-0831	NY RESTAURANTS, LLC
24.	TX	Beaumont	4215 Dowlen Road, Suite A	77706	409-223-1622	ODG, LLC

	State	City	Address	Zip	Phone #	Franchisee Entity
25.	TX	Dallas	12050 Inwood Road, Suite 110	75244	214-462-7974	GREENBRIAR PRESTON HOLLOW RESTAURANT, LLC
26.	TX	Fort Worth	3206 Winthrop	76116	817-349-0772	GREENBRIAR RIDGLEA RESTAURANT, LLC
27.	TX	Denton	210 East Hickory	76201	940-483-8226	GREENBRIAR DENTON RESTAURANT, LLC
28.	TX	Hurst	775 Grapevine Highway, Suite 400	76054	817-605-9075	GREENBRIAR HURST RESTAURANT, LLC
29.	TX	Midland	3303 N. Midkiff Rd	79705	432-218-6636	LIONS DEN, LLC*
30.	TX	Lubbock	6030 Marsha Sharpe Frwy	79407	806-503-2118	WRT2LUBBOCK, LLC
31.	TX	Plano	8000 Coit Road	75025	972-244-4047	DOS JOSES, LLC
32.	TX	San Antonio	17026 Bulverde Rd.	78247	210-265-5626	LONE STAR RESTAURANT GROUP, LLC*
33.	UT	Pleasant View	380 W 2650 North	84414	385-865-8137	HOME RUN RESTAURANT GROUP, INC.*
34.	UT	Herriman	5075 W Herriman Blvd	84096	801-992-1919	GOOD 2 GO EATS, LLC
35	VA	Midlothian	15816 West Center Main Street	23113	804-594-5345	RVA TACO, LLC*
36.	VA	Dulles	Dulles Int'l Airport, Gate D10	20166	801-992-1919	AFFINITY, LLC

^{*} These locations were developed pursuant to a Development Agreement.

Fran	Franchisees Who Have Signed a Franchise Agreement But Have Not Yet Opened as of December 29, 2024				
State	State City Telephone Franchisee				
IL	Rockford	815-990-1046	BUCKY'S TACOS, LLC*		
СО	Colorado Springs	208-535-5936	GOOD 2 GO EATS, LLC*		
СО	Colorado Springs	208-535-5936	GOOD 2 GO EATS, LLC*		
СО	Denver	208-535-5936	GOOD 2 GO EATS, LLC*		
СО	Fountain	208-535-5936	GOOD 2 GO EATS, LLC*		

^{*} These locations will be developed pursuant to a Development Agreement.

FRANCHISEES THAT LEFT THE SYSTEM DURING THE 2024 FISCAL YEAR

The following franchisees left the system in the calendar year 2024 or have not communicated with us within 10 weeks of this Disclosure Document's issuance date.

	Former Franchisees				
	State	City	Telephone	Franchisee Entity	Reason
1.	IA	Dubuque	N/A	Flatin RT Des Moines, LLC	Transfer
2.	KS	Lawrence	785-856-8226	Flatin RT Lawrence, LLC	Termination
3.	NM	Albuquerque	505-508-5989	Dia De Los Tacos LLC	Termination
4.	NV	Reno	775-376-1085	Ocathain Partners, LLC	Termination
5.	TX	Arlington	817-272-2011	Compass Group USA, Inc.	Termination
6.	VA	Short Pump	804-562-3153	RVA Taco, LLC	Termination

If you buy a Rusty Taco® franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

EXHIBIT F

TABLE OF CONTENTS OF OPERATIONS MANUAL

RESTAURANT OPERATIONS







INDEX

OVERVIEW 3

BRAND STRATEGY 4-6 RUSTY TACO STORY 7-10

VENDORS 11

VENDOR CONTACT LIST 12-16

JOB RESPONSIBILITIES 17

CASHIER 18-25 LINE COOK 26-29 PREP COOK 30-32 HOST 33-36 CHIPPER 37-39

OPENING AND CLOSING CHECKLISTS 40

MANAGER 41-44 CASHIER 45-47 LINE COOK 48-51

CLEANING SCHEDULES MAINTENANCE 52

DAILY CLEANING 53
WEEKLY CLEANING 54
30 DAY MAINTENANCE 55-56
90 DAY MAINTENANCE 57

FORMS AND GUIDES 58

DAILY PREP SHEET 59 SHIFT PLAN 60 **DAILY TEMP SHEET 61 QUALITY LINE CHECK FORM 62 COOLING LOG 63** DAILY WASTE LOG 64 DAILY OPS CHECK 65 GLOSSARY 66-70 **HOT HOLDING TIMES 71** FRYER COOKING TIMES 72 **HOT LINE DIAGRAM 73 COLD LINE DIAGRAM 74 TACOS AND GARNISHES 75 PORTION GUIDE 76-78** ATTENDANCE TRACKER 79 **ILLNESS LOG 80**

RECOMMENDED POLICIES 81

UNIFORM POLICY 82-84
CASH HANDLING POLICY 85-87
SHIFT MEAL POLICY 88
TIP POLICY 89-93
ALCOHOL POLICY 94-96
ATTENDANCE POLICY 97

POS 98

MENU MGMT REQUEST FORM 99-101 HARDWARE/SOFTWARE 102-104 OPERATIONS GUIDE 105-114

GIFT CARDS 115

QUICK REFERNCE GUIDE 116 GIFT CARD MANUAL 117-121

ALLERGY/NUTRITION 122

ALLERGY AWARENESS 123-125 NUTRITIONAL CHART 126-1128 ALLERGEN IDENTIFIER 129-130 POSTINGS 131-133

THIS IS HOW WE DO IT 134

INVENTORY 135-136 PERIOD END 137 REFUNDS 138

TELEPHONE ANSWERING 139

FOOD SAFETY 140

FOOD SAFETY MANUAL 141-164 IMPLEMENTATION GUIDE 165-172 TRAINING GUIDE 173-201 POSTERS 202 FORMS 203-210

EXHIBIT G

 $\frac{\textbf{LIST OF STATE ADMINISTRATORS/AGENTS FOR SERVICE OF}}{\textbf{PROCESS}}$

STATE ADMINISTRATORS/AGENTS FOR SERVICE OF PROCESS

Listed here are the names, addresses and telephone numbers of the state agencies having responsibility for the franchising disclosure/registration laws. We may not yet be registered to sell franchises in any or all of these states. There may be states in addition to those listed below in which we have appointed an agent for service of process. There may also be additional agents appointed in some of these states.

CALIFORNIA

Department of Financial Protection & Innovation:

Toll Free: 1 (866) 275-2677

Los Angeles

Commissioner of Financial Protection & Innovation 320 West 4th Street Suite 750 Los Angeles, California 90013 (213) 576-7500

Sacramento

Commissioner of Financial Protection & Innovation 2101 Arena Blvd. Sacramento, CA 95834 (916) 445-7205

San Diego

Commissioner of Financial Protection & Innovation 1455 Frazee Road, Suite 315 San Diego, California 92108 (619) 610-2093

San Francisco

Commissioner of Financial Protection & Innovation
One Sansome Street, Suite 600
San Francisco, California 94104-4428
(415) 972-8559

HAWAII

(state administrator)

Business Registration Division Securities Compliance Branch Department of Commerce and Consumer Affairs P.O. Box 40 Honolulu, Hawaii 96810 (808) 586-2727

ILLINOIS

Franchise Bureau Office of the Attorney General 500 South Second Street Springfield, Illinois 62706 (217) 782-4465

INDIANA

(state administrator)

Indiana Secretary of State Securities Division, E-111 302 West Washington Street Indianapolis, Indiana 46204 (317) 232-6681

(agent for service of process)

Indiana Secretary of State 200 West Washington Street, Room 201 Indianapolis, Indiana 46204 (317) 232-6531

MARYLAND

(state administrator)

Office of the Attorney General Securities Division 200 St. Paul Place Baltimore, Maryland 21202-2021 (410) 576-6300

(agent for service of process)

Maryland Securities Commissioner at the Office of the Attorney General Securities Division 200 St. Paul Place Baltimore, Maryland 21202-2021 (410) 576-6360

MICHIGAN

(state administrator)

Michigan Attorney General's Office Consumer Protection Division Attn: Franchise Section G. Mennen Williams Building 525 West Ottawa Street Lansing, Michigan 48909 (517) 373-7117

(agent for service of process)

Michigan Department of Commerce, Corporations, Securities, and Commercial Licensing Bureau P.O. Box 30018 Lansing, Michigan 48909

MINNESOTA

(state administrator)

Commissioner of Commerce Minnesota Department of Commerce 85 7th Place East, Suite 280 St. Paul, Minnesota 55101 (651) 539-1600 (agent for service of process)

Commissioner of Commerce Minnesota Department of Commerce 85 7th Place East, Suite 280 St. Paul, Minnesota 55101 (651) 539-1600

NEW YORK

(state administrator)

NYS Department of Law Investor Protection Bureau 28 Liberty Street, 21st Floor New York, New York 10005 (212) 416-8222

(agent for service of process)

Secretary of State 99 Washington Avenue Albany, New York 12231 (518) 473-2492

NORTH DAKOTA

North Dakota Securities Department 600 East Boulevard Avenue State Capitol, 14th Floor, Dept. 414 Bismarck, North Dakota 58505 (701) 328-4712

RHODE ISLAND

Department of Business Regulation Division of Securities John O. Pastore Complex Building 69-1 1511 Pontiac Avenue Cranston, Rhode Island 02920 (401) 462-9645

SOUTH DAKOTA

Division of Insurance Securities Regulation 124 S. Euclid, Second Floor Pierre, South Dakota 57501 (605) 773-3563

VIRGINIA

(state administrator)
State Corporation Commission
Division of Securities and Retail Franchising
1300 East Main Street, Ninth Floor
Richmond, Virginia 23219
(804) 371-9051

(agent for service of process)
Clerk, State Corporation Commission
1300 East Main Street, 1st Floor
Richmond, Virginia 23219
(804) 371-9733

WISCONSIN

(state administrator)

Securities and Franchise Registration Wisconsin Department of Financial Institutions 4822 Madison Yards Way, North Tower Madison, Wisconsin 53705 (608) 266-0448

(agent for service of process)

Office of the Secretary Wisconsin Department of Financial Institutions P.O. Box 8861 Madison, Wisconsin 53708-8861 (608) 261-9555

WASHINGTON

(state administrator)

Department of Financial Institutions Securities Division P.O. Box 41200 Olympia, Washington 98504-1200 (360) 902-8760

EXHIBIT H STATE ADDENDA AND RIDERS

ADDITIONAL DISCLOSURES FOR THE FRANCHISE DISCLOSURE DOCUMENT OF RUSTY TACO FRANCHISING, LLC

The following are additional disclosures for the Franchise Disclosure Document of Rusty Taco Franchising, LLC required by various state franchise laws. Each provision of these additional disclosures will only apply to you if the applicable state franchise law applies to you.

<u>FOR THE FOLLOWING STATES</u>: CALIFORNIA, ILLINOIS, INDIANA, MARYLAND, MICHIGAN, MINNESOTA, NEW YORK, NORTH DAKOTA, VIRGINIA, OR WISCONSIN.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

FOR THE STATE OF CALIFORNIA

The registration of this franchise offering by the California Department of Financial Protection and Innovation does not constitute approval, recommendation, or endorsement by the commissioner.

- 1. THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE DISCLOSURE DOCUMENT.
- 2. SECTION 31125 OF THE CALIFORNIA CORPORATIONS CODE REQUIRES US TO GIVE YOU A DISCLOSURE DOCUMENT, IN A FORM CONTAINING THE INFORMATION THAT THE COMMISSIONER MAY BY RULE OR ORDER REQUIRE, BEFORE A SOLICITATION OF A PROPOSED MATERIAL MODIFICATION OF AN EXISTING FRANCHISE.
- 3. NEITHER WE, OUR PARENT, PREDECESSOR OR AFFILIATE NOR ANY PERSON IN ITEM 2 OF THE FRANCHISE DISCLOSURE DOCUMENT IS SUBJECT TO ANY CURRENTLY EFFECTIVE ORDER OF ANY NATIONAL SECURITIES ASSOCIATION OR NATIONAL SECURITIES EXCHANGE, AS DEFINED IN THE SECURITIES EXCHANGE AC OF 1934, 15 U.S.C.A SECTIONS 78A ET SEQ., SUSPENDING OR EXPELLING SUCH PERSONS FROM MEMBERSHIP IN THAT ASSOCIATION OR EXCHANGE.
- 4. OUR WEBSITE, www.rustytaco.com, HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION & INNOVATION. ANY COMPLAINTS CONCERNING THE CONTENT OF THE WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION & INNOVATION AT www.dfpi.ca.gov.
- 5. Section 31512.1 of the California Corporations Code requires that any provision of the Franchise Agreement, Disclosure Document, acknowledgment, questionnaire, or other writing, including any exhibit thereto, disclaiming or denying any of the following shall be deemed contrary to public policy and shall be void and unenforceable: (a) representations made by the franchisor or its personnel or agents to a prospective franchisee; (b) reliance by a franchisee on any representations made by the franchisor or

1

its personnel or agents; (c) reliance by a franchisee on the franchise disclosure document, including any exhibit thereto; or (d) violations of any provision of this division.

6. The following is added to the end of Item 5 and Item 7:

The California Department of Financial Protection and Innovation has required us to defer the collection of initial franchise fee and other initial payments you owe us until we have completed all of our pre-opening obligations under the Franchise Agreement and you have begun operating your Restaurant. If you sign a Development Agreement, the payment of the development and initial fees attributable to a specific unit in your Development Schedule is deferred until that unit is open.

7. The following paragraphs are added at the end of Item 17:

You must sign a general release if you renew or transfer your franchise or development rights. California Corporations Code § 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code §§ 31000 through 31516). Business and Professions Code § 20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code §§ 20000 through 20043).

The Franchise Agreement contains a liquidated damages clause. Under California Civil Code Section 1671, certain liquidated damages clauses are unenforceable.

California Business and Professions Code §§ 20000 through 20043 (the Franchise Relations Act) provide rights to the franchisee concerning termination, transfer, or non-renewal of a franchise. If the franchise agreement contains a provision that is inconsistent with the law, the law will control. In particular, Business and Professions Code § 20010 voids a waiver of your rights under the Franchise Relations Act.

The Franchise Agreement and Development Agreement contain a covenant not to compete that extends beyond termination of the franchise. This provision might not be enforceable under California law.

The Franchise Agreement provides for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law (11 U.S.C.A Section 101 et seq.).

Under the Franchise Agreement, we reserve the right to require that franchisees comply with maximum and minimum prices it sets for goods and services. The Antitrust Law Section of the Office of the California Attorney General views maximum price agreements as per se violations of the California's Cartwright Act (Cal. Bus. and Prof. Code §§ 16700 to 16770).

8. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

FOR THE STATE OF ILLINOIS

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

1. The following paragraph is added to the end of Items 5 and 7:

Pursuant to an order of the Illinois Attorney General's Office, imposed based on our financial condition, we will defer collection of the initial franchise fee and other initial payments you owe us until we have completed all of our pre-opening obligations to you under the Franchise Agreement and you have begun operating your Restaurant. If you sign a Development Agreement, the payment of the development and initial fees attributable to a specific unit in your Development Schedule is deferred until that unit is open.

2. The following paragraphs are added to the end of Item 17:

Except for the U.S. Federal Arbitration Act and other federal laws in the U.S., the laws of the State of Illinois will govern the Franchise Agreement.

Section 4 of the Illinois Franchise Disclosure Act provides that any provision in a franchise agreement that designates jurisdiction or venue outside the State of Illinois is void. However, a franchise agreement may provide for arbitration outside of Illinois.

Section 41 of the Illinois Franchise Disclosure Act provides that any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.

Your rights upon termination and non-renewal of a franchise agreement are subject to sections 19 and 20 of the Illinois Franchise Disclosure Act.

3. The following is added to the "Summary" section of row (t) in both tables in Item 17:

Notwithstanding the foregoing, nothing in this or any related agreement is intended to disclaim the express representations made in the Franchise Disclosure Document, its exhibits and amendments.

4. For info about obtaining a liquor license in Illinois, see: https://www.illinois.gov/ilcc/Pages/Forms-and-Applications.aspx. For info about obtaining TIPS certification in Illinois, see: https://www.tipscertified.com/tips-state-pages/illinois/. For the Illinois Dram Shop laws, see: Liquor Control Act of 1934, 235 ILCS 5/ (West 2018).

FOR THE STATE OF MARYLAND

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

1. The following language is added to the end of Item 5:

Based upon the franchisor's financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by franchisees shall be deferred until the franchisor completes its pre-opening obligations under the franchise agreement. In addition, all development fees and initial payments by area developers shall be deferred until the first franchise under the development agreement opens.

2. The following language is added at the end of Item 17.

The Development Agreement and Franchise Agreement provide for termination upon bankruptcy. This provision might not be enforceable under federal bankruptcy law (111 U.S.C. Sections 101 et seq.) but we will enforce it to the extent enforceable.

3. The following is added to the end of Item 17(c) and Item 17(m):

However, any release required as a condition of renewal and/or assignment/transfer will not apply to claims arising under the Maryland Franchise Registration and Disclosure Law.

4. The following language is added at the end of Item 17(u):

The Development Agreement and Franchise Agreement provide that disputes are resolved through arbitration. Maryland franchise regulation states that it is an unfair or deceptive practice to require a franchisee to waive its right to file a lawsuit in Maryland claiming a violation of the Maryland Franchise Law. In light of the Federal Arbitration Act, there is some dispute as to whether this forum selection requirement is legally enforceable

5. The following language is added at the end of Item 17(v):

A franchisee may bring a lawsuit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law. Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

FOR THE STATE OF MINNESOTA

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

1. The following language is added to the end of Items 5 and 7:

The Minnesota Department of Commerce (Securities Section) has required us to defer payment of all initial franchise fees and development fees until we have completed all of our pre-opening obligations and your business opens.

2. The following is added to the "Remarks" column of the "Non-Sufficient Funds Fee" in Item 6:

Minnesota Statute 604.113 caps service charges at \$30 per incurrence.

3. The following is added at the end of the chart in Item 17:

With respect to franchises governed by Minnesota law, we will comply with Minn. Stat. Sec. 80C.14, Subds. 3, 4 and 5 which require, except in certain specified cases, that you be given 90 days' notice of termination (with 60 days to cure) of the Franchise Agreement and Development Agreement and 180 days' notice for non-renewal of the Franchise Agreement and Development Agreement.

Minn. Stat. Sec. 80C.21 and Minn. Rule 2860.4400J prohibits us from requiring litigation to be conducted outside Minnesota, requiring waiver of a jury trial or requiring the Developer or Franchisee to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the Disclosure Document, Development Agreement or Franchise Agreement can abrogate or reduce any of Developer's or Franchisee's rights as provided for in Minnesota Statutes, Chapter 80C, or Developer's or Franchisee's rights to any procedure, forum or remedies provided for by the laws of the jurisdiction.

Any release required as a condition of renewal or transfer/assignment will not apply to the extent prohibited by Governing Law with respect to claims arising under Minn. Rule 2860.4400D.

FOR THE STATE OF NEW YORK

1. The following information is added to the cover page of the Franchise Disclosure Document:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT G OR YOUR PUBLIC LIBRARY FOR RESOURCES OR INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN ANYTHING IN THIS FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE APPROPRIATE STATE OR PROVINCIAL AUTHORITY. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS THAT ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. The following is added at the end of Item 3:

Except as provided above, the following applies to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

- A. No such party has an administrative, criminal, or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices or comparable civil or misdemeanor allegations.
- B. No such party has civil actions pending against that party, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature, or financial condition of the franchise system or its business operations.
- C. No such party has been convicted of a felony or pleaded <u>nolo contendere</u> to a felony charge or, within the ten years immediately preceding the application for registration, has been convicted of or pleaded <u>nolo contendere</u> to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices; or comparable allegations.
- D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.
- 3. The following is added to the end of end of the "Summary" sections of Item 17(c), entitled Requirements for franchisee to renew or extend, and Item 17(m), entitled Conditions for franchisor approval of transfer:

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force. This proviso intends that the non-waiver provisions of GBL Sections 687(4) and 687(5) be satisfied.

4. The following is added to the end of the "Summary" section of Item 17(d), titled <u>Termination by franchisee</u>:

You may terminate the Franchise Agreement on any grounds available by law.

5. The following is added to the end of the "Summary" sections of Item 17(v), titled <u>Choice of forum</u>, and Item 17(w), titled <u>Choice of law</u>:

However, the governing choice of law and choice of forum shall not be considered a waiver of any right conferred upon you by the provisions of Article 33 of the General Business Law of the State of New York.

- 6. Franchise Questionnaires and Acknowledgements-- No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 7. Receipts--Any sale made must be in compliance with § 683(8) of the Franchise Sale Act (N.Y. Gen. Bus. L. § 680 et seq.), which describes the Rev. April 2, 2024 time period a Franchise Disclosure Document (offering prospectus) must be provided to a prospective franchisee before a sale may be made. New York law requires a franchisor to provide the Franchise Disclosure Document at the earliest of the first personal meeting, ten (10) business days before the execution of the franchise or other agreement, or the payment of any consideration that relates to the franchise relationship.

FOR THE STATE OF NORTH DAKOTA

- 1. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 2. The following is added to the end of Item 5:

Pursuant to an order of the North Dakota Securities Commissioner, we will defer collection of the initial franchise fee and other initial payments you owe us until we have completed all of our pre-opening obligations to you under the Franchise Agreement, and you begin operating your Restaurant. If you sign a Development Agreement, no development fee will be due under the Development Agreement until we have completed all of our pre-opening obligations to you and you begin operating your first Restaurant.

3. The following is added to the "Remarks" column in the Item 6 chart for the row entitled Lost Revenue Damages, and to the end of the "Summary" section of Item 17(i), entitled Franchisee's obligations on termination or non-renewal:

Under North Dakota law, a requirement that you consent to liquidated damages or termination penalties in the event of termination of the Franchise Agreement is considered unenforceable.

4. The following is added to the end of the "Summary" sections of Item 17(c), entitled Requirements for franchisee to renew or extend, and Item 17(m), entitled Conditions for franchisor approval of transfer:

However, any release required as a condition of renewal and/or assignment/transfer will not apply to the extent prohibited by the North Dakota Franchise Investment Law.

5. The following is added to the end of the "Summary" section of Item 17(r), entitled <u>Non-competition covenants after the franchise is terminated or expires:</u>

Covenants not to compete such as those mentioned above are generally considered unenforceable in the State of North Dakota; however, we and you will enforce the covenants to the maximum extent the law allows.

6. The following is added to the end of the "Summary" section of Item 17(u), entitled <u>Dispute resolution by arbitration or mediation</u>:

To the extent required by the North Dakota Franchise Investment Law (unless such requirement is preempted by the Federal Arbitration Act), arbitration will be at a site to which we and you mutually agree.

7. The following is added to the end of the "Summary" section of Item 17(v), entitled <u>Choice of forum:</u>

However, subject to your arbitration obligation, and to the extent required by North Dakota Franchise Investment Law, you may bring an action in North Dakota.

8. The "Summary" section of Item 17(w), entitled <u>Choice of law</u>, is deleted and replaced with the following:

Except as otherwise required by North Dakota law, the laws of the State of Delaware will apply.

FOR THE STATE OF VIRGINIA

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

1. The following is added to the end of Item 5:

The Virginia State Corporation Commission's Division of Securities and Retail Franchising requires us to defer payment of the initial franchise fee and other initial payments owed by franchisees to the franchisor until the franchisor has completed its pre-opening obligations under the franchise agreement.

2. The following language is added to the end of Item 17(h):

Pursuant to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any grounds for default or termination stated in the Development Agreement or Franchise Agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

THE FOLLOWING PAGES IN THIS EXHIBIT ARE STATE-SPECIFIC RIDERS TO THE DEVELOPMENT AGREEMENT

RIDER TO THE RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT FOR USE IN CALIFORNIA

THIS RIDER (this "Rider") is made and entered	d into by and between RUSTY TACO
FRANCHISING, LLC, a Delaware limited liability company	with its principal business address at 5412
W. Plano Pkwy., Suite 100, Plano, Texas 75093 ("us"), and	d
whose principal business address is	
1. BACKGROUND. We and you are parties to, 20 (the "Development Agreement") that had of this Rider. This Rider is annexed to and forms part of the Development is amended only as set forth below. In the even Development Agreement and this Rider, this Rider shall gove signed because (a) the offer for sale of the development riging California; or (b) your Development Area is or will be located domiciled in the State of California.	as been signed concurrently with the signing relopment Agreement, and the Development ent of conflict between the terms of the ern such inconsistency. This Rider is being this was made or accepted in the State of
2. FINANCIAL ASSURANCE . The following	is added to the end of Section 2.1 of the
Development Agreement:	
The California Department of Financial Protection and the collection of initial franchise fee and other initial prompleted all of our pre-opening obligations under the begun operating your Restaurant. If you sign a Development and initial fees attributable to a subschedule is deferred until that unit is open.	Franchise Agreement and you have opment Agreement, the payment of

[Signature Page Follows]

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Development Agreement.

RUSTY TACO FRANCHISING, LLC	[DEVELOPER]
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

RIDER TO THE RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT FOR USE IN ILLINOIS

THIS KIDER (this Kide	er) is made	and entered	i into by	and between	KUSIY	IACO
FRANCHISING, LLC, a Delaware	limited liabil	lity company	with its pr	rincipal busines	s address	at 5412
W. Plano Pkwy., Suite 100, Plano,	Texas 75093	3 (" us "), and				,
whose principal business address is				("you'	').	
				· -		

- 1. <u>BACKGROUND</u>. We and you are parties to that certain Development Agreement dated ______, 20___ (the "Development Agreement") that has been signed concurrently with the signing of this Rider. This Rider is annexed to and forms part of the Development Agreement, and the Development Agreement is amended only as set forth below. In the event of conflict between the terms of the Development Agreement and this Rider, this Rider shall govern such inconsistency. This Rider is being signed because (a) you are domiciled in the State of Illinois, or (b) the offer to sell the Development Rights was made or accepted in the State of Illinois and the Development Area is located in the State of Illinois.
- 2. <u>ACKNOWLEDGMENT</u>. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 3. **FINANCIAL ASSURANCE.** The following is added to the end of Section 2.1 of the Development Agreement:

Pursuant to an order of the Illinois Attorney General's Office, imposed based on our financial condition, we will defer collection of the initial franchise fee and other initial payments you owe us until we have completed all of our pre-opening obligations to you under the Franchise Agreement and you have begun operating your Restaurant. If you sign a Development Agreement, the payment of the development and initial fees attributable to a specific unit in your Development Schedule is deferred until that unit is open.

4. **<u>DISPUTES</u>**. The following is added to the end of Section 11.8 of the Development Agreement:

Nothing contained in this Section shall constitute a condition, stipulation, or provision purporting to bind any person to waive compliance with any provision of the Illinois Franchise Disclosure Act or any other law of the State of Illinois, to the extent applicable.

5. <u>ILLINOIS FRANCHISE DISCLOSURE ACT</u>. The following language is added to the end of the Development Agreement:

Except for the U.S. Federal Arbitration Act and other federal laws in the U.S., the laws of the State of Illinois will govern this Agreement.

Section 4 of the Illinois Franchise Disclosure Act provides that any provision in a Development Agreement that designates jurisdiction or venue outside the State of Illinois is void. However, a Development Agreement may provide for arbitration outside of Illinois.

Section 41 of the Illinois Franchise Disclosure Act provides that any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.

Your rights upon termination and non-renewal of a Development Agreement are subject to sections 19 and 20 of the Illinois Franchise Disclosure Act.

6. <u>LIQUOR LICENSING</u>. For info about obtaining a liquor license in Illinois, see: https://www.illinois.gov/ilcc/Pages/Forms-and-Applications.aspx. For info about obtaining TIPS certification in Illinois, see: https://www.tipscertified.com/tips-state-pages/illinois/. For the Illinois Dram Shop laws, see: Liquor Control Act of 1934, 235 ILCS 5/ (West 2018).

[Signature Page to Follow]

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Development Agreement.

RUSTY TACO FRANCHISING, LLC	[DEVELOPER]
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

RIDER TO THE RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT FOR USE IN MARYLAND

THIS RIDER (this " Rider ") is made and entere	ed into by and between RUSTY TACO
FRANCHISING, LLC, a Delaware limited liability company	y with its principal business address at 5412
W. Plano Pkwy., Suite 100, Plano, Texas 75093 ("us"), an	nd,
whose principal business address is	("you").
1 RACKCROUND We and you are parties to t	that certain Develonment Agreement dated

- 1. <u>BACKGROUND</u>. We and you are parties to that certain Development Agreement dated ______, 20____ (the "Development Agreement") that has been signed concurrently with the signing of this Rider. This Rider is annexed to and forms part of the Development Agreement, and the Development Agreement is amended only as set forth below. In the event of conflict between the terms of the Development Agreement and this Rider, this Rider shall govern such inconsistency. This Rider is being signed because (a) you are a resident of the State of Maryland; <u>or</u> (b) the Development Area is located in the State of Maryland; <u>or</u> (c) the offer to sell the Development Rights was made in the State of Maryland; <u>or</u> (d) the offer to buy the Development Rights was accepted in the State of Maryland.
- 2. <u>ACKNOWLEDGMENT</u>. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 3. **FINANCIAL ASSURANCE**. The following is added to the end of Section 2.1 of the Development Agreement:

In addition, all development fees and initial payments by area developers shall be deferred until the first franchise under the development agreement opens.

4. **INSOLVENCY**. The following is added to the end of Section 7.1 of the Development Agreement:

The provision which provides for termination upon your bankruptcy might not be enforceable under federal bankruptcy law (11 U.S.C. Sections 101 et seq.).

5. <u>VENUE</u>. Sections 11.2 and 11.9 of the Development Agreement is supplemented by adding the following to the end of the Section:

A franchisee may bring a lawsuit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.

A Maryland franchise regulation states that it is an unfair or deceptive practice to require a franchisee to waive its right to file a lawsuit in Maryland claiming a violation of the Maryland Franchise Registration and Disclosure Law. In light of the Federal Arbitration Act, there is some dispute as to whether this forum selection requirement is legally enforceable.

6. <u>LIMITATIONS OF CLAIMS</u>. The following is added to the end of Section 11.8 of the Development Agreement:

Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

7. <u>RELEASES</u>. The Development Agreement is further amended to state that "All representations requiring prospective franchisees to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law."

[Signature Page to Follow]

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Development Agreement.

RUSTY TACO FRANCHISING, LLC	[DEVELOPER]
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

RIDER TO THE RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT FOR USE IN MINNESOTA

THIS RIDER (this "Rider") is made and entered into by and between RUSTY TACO

FRANCHISING, LLC, a Delaware limited liability com	pany with its principal business address at 5412
W. Plano Pkwy., Suite 100, Plano, Texas 75093 ("us")	, and,
whose principal business address is	("you").
1. BACKGROUND . We and you are parties	to that certain Development Agreement dated
, 20 (the "Development Agreement") t	hat has been signed concurrently with the signing
of this Rider. This Rider is annexed to and forms part of the	e Development Agreement, and the Development
Agreement is amended only as set forth below. In the	e event of conflict between the terms of the
Development Agreement and this Rider, this Rider shall	govern such inconsistency. This Rider is being
signed because (a) the Development Area is in the State of	Minnesota; (b) you are a resident of the State of
Minnesota; and/or (c) any of the offering or sales activity	relating to the Development Rights occurred in

- 2. <u>ACKNOWLEDGMENT</u>. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 3. **FINANCIAL ASSURANCE**. The following is added to the end of Section 2.1 ("Development Fee") of the Development Agreement:

Notwithstanding anything to the contrary, all development fees payable by us to you shall be deferred until the first franchise under this Agreement opens.

4. **INJUNCTIVE RELIEF**. The following language is added to the end of Section 11.6 of the Development Agreement:

Notwithstanding the foregoing, a court will determine if a bond is required.

- 5. **LIMITATIONS OF CLAIMS**. The following is added to the end of Section 11.8 of the Development Agreement:
 - ; provided, however, that Minnesota law provides that no action may be commenced under Minn. Stat. Sec. 80C.17 more than 3 years after the cause of action accrues.
- 6. MINNESOTA LAW. Notwithstanding anything to the contrary contained in the Development Agreement, Minn. Stat. Sec. 80C.21 and Minn. Rule 2860.4400J prohibit us from requiring you to waive your rights to a jury trial or to waive your rights to any procedure, forum or remedies provided for by the laws of the jurisdiction, or to consent to liquidated damages, termination penalties or judgment notes.

[Signature Page to Follow]

the State of Minnesota.

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Development Agreement.

RUSTY TACO FRANCHISING, LLC	[DEVELOPER]				
By:	By:				
Name:	Name:				
Title:	Title:				
Date:	Date:				

RIDER TO THE RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT FOR USE IN NEW YORK

THIS RIDER (this Rider) is made and entered into	o by and between RUSIY IACO
FRANCHISING, LLC, a Delaware limited liability company with	its principal business address at 5412
W. Plano Pkwy., Suite 100, Plano, Texas 75093 ("us"), and	
whose principal business address is	("you").
1. <u>BACKGROUND</u> . We and you are parties to that of this Rider. This Rider is annexed to and forms part of the Developer Agreement is amended only as set forth below. In the event of Development Agreement and this Rider, this Rider shall govern sufficient because (a) an offer to sell the Development Rights is made to buy the Development Rights is accepted in New York, (c) you are and the Development Area is or will be located in the State of New York.	certain Development Agreement dated en signed concurrently with the signing ment Agreement, and the Development of conflict between the terms of the ach inconsistency. This Rider is being in the State of New York; (b) an offer re domiciled in the State of New York
2. TRANSFER BY US . The following language is a Development Agreement:	added to the end of Section 6.1 of the
However, to the extent required by applicable law, no transassignee who, in our good faith judgment, is willing and a under this Development Agreement.	•
3. DEFAULT AND TERMINATION . The following Section 7 of the Development Agreement:	ing language is added to the end of
You also may terminate this Development Agreement on a under the provisions of Article 33 of the General Business L	• •
4. CHOICE OF LAW . The following language is ac	dded to the end of Section 11.1 of the

5. **CONSENT TO JURISDICTION**. The following language is added to the end of Section 11.9 of the Development Agreement:

This Section shall not be considered a waiver of any right conferred upon you by the provisions of Article 33 of the New York State General Business Law, as amended, and

This Section shall not be considered a waiver of any right conferred upon you by the provisions of Article 33 of the New York State General Business Law, as amended, and the regulations issued thereunder.

[Signature Page to Follow]

Development Agreement:

the regulations issued thereunder.

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Development Agreement.

RUSTY TACO FRANCHISING, LLC	[DEVELOPER]			
By:	Ву:			
Name:	Name:			
Title:	Title:			
Date:	Date:			

RIDER TO THE RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT FOR USE IN NORTH DAKOTA

THIS (this "Rider") is made and entered into by and between RUSTY TACO FRANCHISING,
LLC, a Delaware limited liability company with its principal business address at 5412 W. Plano Pkwy.,
Suite 100, Plano, Texas 75093 ("us"), and, whose principal business address is ("you").
business address is ("you").
1. BACKGROUND . We and you are parties to that certain Development Agreement dated
, 20 (the "Development Agreement") that has been signed concurrently with the signing
of this Rider. This Rider is annexed to and forms part of the Development Agreement, and the Development
Agreement is amended only as set forth below. In the event of conflict between the terms of the
Development Agreement and this Rider, this Rider shall govern such inconsistency. This Rider is being
signed because (a) an offer to sell the Development Rights is made or accepted in the State of North Dakota;
or (b) you are domiciled in the State of North Dakota, the Development Area is or will be developed in the
State of North Dakota.
2. FINANCIAL ASSURANCE . The following is added to the end of Section 3 of the
Development Agreement:
Pursuant to an order of the North Dakota Securities Commissioner, we will defer collection
of the development fee and other initial payments you owe us until we have completed all
of our pre-opening obligations to you, and you begin operating your first Restaurant.
3. CHOICE OF LAW. The following is added to the end of Section 11.1 (Choice of Law)
of the Development Agreement:
Except as otherwise required by North Dakota law, this Agreement will be governed by
and construed in accordance with the laws of the State of Delaware.
and construct in accordance with the laws of the state of Delaware.
4. ARBITRATION . The following is added to the end of the fourth paragraph of
Section 11.2 (Requirement to Submit Disputes to Arbitration) and Section 11.9 (Consent to Jurisdiction) of
the Development Agreement:
Notwithstanding the foregoing, to the extent required by the North Dakota Franchise
Investment Law, and subject to your arbitration obligations, you may (i) bring an action in
North Dakota for claims arising under the North Dakota Franchise Investment Law, and
(ii) arbitration will be at a site to which we and you mutually agree.

The statutes of limitations under North Dakota law applies with respect to claims arising under the North Dakota Franchise Investment Law.

North Dakota law, Section 11.4 and Section 11.5 of the Development Agreement are hereby deleted.

WAIVER OF JURY TRIAL; PUNITIVE DAMAGES. Only to the extent required by

LIMITATION OF CLAIMS & CLASS ACTION BAR. The following is added to the

[Signature Page to Follow]

end of Section 11.8 of the Development Agreement:

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Development Agreement.

RUSTY TACO FRANCHISING, LLC	[FRANCHISEE]				
By:	By:				
Name:	Name:				
Title:	Title:				
Date:	Date:				

RIDER TO THE RUSTY TACO FRANCHISING, LLC DEVELOPMENT AGREEMENT FOR USE IN VIRGINIA

LLC, a Delaware limited liability company with	into by and between RUSTY TACO FRANCHISING , its principal business address at 5412 W. Plano Pkwy., whose principal, whose principal
, 20 (the " Development Agreen of this Rider. This Rider is annexed to and forms paragreement is amended only as set forth below	are parties to that certain Development Agreement dated nent") that has been signed concurrently with the signing art of the Development Agreement, and the Development 7. In the event of conflict between the terms of the er shall govern such inconsistency. This Rider is being the of Virginia.
2. FINANCIAL ASSURANCE . To Development Agreement:	he following is added to the end of Section 3 of the
Franchising requires us to defer paymen	nission's Division of Securities and Retail at of the initial franchise fee and other initial chisor until the franchisor has completed its pre-greement.
IN WITNESS WHEREOF, the parties had of the effective date of the Development Agreement	ave executed and delivered this Rider to be effective as nt.
RUSTY TACO FRANCHISING, LLC	[FRANCHISEE]
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

THE FOLLOWING PAGES IN THIS EXHIBIT ARE STATE-SPECIFIC RIDERS TO THE FRANCHISE AGREEMENT

RIDER TO THE RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT FOR USE IN CALIFORNIA

THIS RIDER (this "Rider") is made and entered into by and between RUSTY	TACO
FRANCHISING, LLC, a Delaware limited liability company with its principal business address	at 5412
W. Plano Pkwy., Suite 100, Plano, Texas 75093 ("us"), and	,
whose principal business address is("you").	
1. BACKGROUND . We and you are parties to that certain Development Agreemer, 20 (the " Franchise Agreement ") that has been signed concurrently with the signal concurrently with the signal concurrence.	
this Rider. This Rider is annexed to and forms part of the Franchise Agreement, and the Fr Agreement is amended only as set forth below. In the event of conflict between the terms of the Fr Agreement and this Rider, this Rider shall govern such inconsistency. This Rider is being signed to the offer for sale of the franchise was made or accepted in the State of California; or (b) the Rest that you will develop or operate under your Franchise Agreement is or will be located in the State of California and you are domiciled in the State of California.	anchise because staurant
2. FINANCIAL ASSURANCE . The following is added to the end of Section 4.1 Franchise Agreement:	of the
The California Department of Financial Protection and Innovation has required us to defer the collection of initial franchise fee and other initial payments you owe us until we have completed all of our pre-opening obligations under the Franchise Agreement and you have begun operating your Restaurant. If you sign a Development Agreement, the payment of the development and initial fees attributable to a specific unit in your Development Schedule is deferred until that unit is open.	

[Signature Page Follows]

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Franchise Agreement.

RUSTY TACO FRANCHISING, LLC	[FRANCHISEE]			
By:	By:			
Name:	Name:			
Title:	Title:			
Date:	Date:			

RIDER TO THE RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT FOR USE IN ILLINOIS

THIS (this "Rider") is made and entered into by and between RUSTY TACO FRANCHISING,
LLC, a Delaware limited liability company with its principal business address at 5412 W. Plano Pkwy.,
Suite 100, Plano, Texas 75093 ("us"), and, whose principal
business address is ("you").
1. BACKGROUND. We and you are parties to that certain Development Agreement dated, 20 (the "Franchise Agreement") that has been signed concurrently with the signing of this Rider. This Rider is annexed to and forms part of the Franchise Agreement, and the Franchise Agreement is amended only as set forth below. In the event of conflict between the terms of the Franchise Agreement and this Rider, this Rider shall govern such inconsistency. This Rider is being signed because (a) you are domiciled in the State of Illinois, or (b) the offer to sell the franchise was made or accepted in the State of Illinois and the Restaurant that you will develop or operate under your Franchise Agreement is or will be operated in the State of Illinois.
2. <u>ACKNOWLEDGMENT</u> . No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
3. FINANCIAL ASSURANCE. The following is added to the end of Section 4.1 of the Franchise Agreement:
Pursuant to an order of the Illinois Attorney General's Office, imposed based on our financial

condition, we will defer collection of the initial franchise fee and other initial payments you owe us until we have completed all of our pre-opening obligations to you under the Franchise Agreement and you have begun operating your Restaurant. If you sign a Development Agreement, the payment of the development and initial fees attributable to a specific unit in your Development Schedule is deferred until that unit is open.

4. <u>LIMITATIONS OF CLAIMS</u>. Section 19.8 of the Franchise Agreement are amended by adding the following:

However, nothing contained in this section shall constitute a condition, stipulation, or provision purporting to bind any person to waive compliance with any provision of the Illinois Franchise Disclosure Act or any other law of the State of Illinois, to the extent applicable.

5. <u>ILLINOIS FRANCHISE DISCLOSURE ACT</u>. The following language is added to the end of the Franchise Agreement:

Except for the U.S. Federal Arbitration Act and other federal laws in the U.S., the laws of the State of Illinois will govern this Agreement.

Section 4 of the Illinois Franchise Disclosure Act provides that any provision in a franchise agreement that designates jurisdiction or venue outside the State of Illinois is void.

However, a franchise agreement may provide for arbitration outside of Illinois.

Section 41 of the Illinois Franchise Disclosure Act provides that any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.

Your rights upon termination and non-renewal of a franchise agreement are subject to sections 19 and 20 of the Illinois Franchise Disclosure Act.

6. <u>LIQUOR LICENSING</u>. For info about obtaining a liquor license in Illinois, see: https://www.illinois.gov/ilcc/Pages/Forms-and-Applications.aspx. For info about obtaining TIPS certification in Illinois, see: https://www.tipscertified.com/tips-state-pages/illinois/. For the Illinois Dram Shop laws, see: Liquor Control Act of 1934, 235 ILCS 5/ (West 2018).

[Signature Page to Follow]

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Franchise Agreement.

RUSTY TACO FRANCHISING, LLC	[FRANCHISEE]				
By:	Ву:				
Name:	Name:				
Title:	Title:				
Date:	Date:				

RIDER TO THE RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT FOR USE IN MARYLAND

THIS RIDER (this "Rid	e r ") is mad	le and en	tered into	by and	between	RUSTY	TACO
FRANCHISING, LLC, a Delawar	e limited lial	oility comp	oany with	its princip	al busines	s address	at 5412
W. Plano Pkwy., Suite 100, Plano	, Texas 750	93 ("us"),	and				
whose principal business address is					("you"	').	

- 1. <u>BACKGROUND</u>. We and you are parties to that certain Development Agreement dated ______, 20____ (the "Franchise Agreement") that has been signed concurrently with the signing of this Rider. This Rider is annexed to and forms part of the Franchise Agreement, and the Franchise Agreement is amended only as set forth below. In the event of conflict between the terms of the Franchise Agreement and this Rider, this Rider shall govern such inconsistency. This Rider is being signed because (a) you are a resident of the State of Maryland; <u>or</u> (b) the Restaurant that you will develop or operate under your Franchise Agreement is or will be operated in the State of Maryland; <u>or</u> (c) the offer to sell the franchise is made in the State of Maryland; <u>or</u> (d) the offer to buy the franchise is accepted in the State of Maryland.
- 2. <u>ACKNOWLEDGMENT</u>. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 3. **FINANCIAL ASSURANCE**. The following is added to the end of Section 4.1 of the Franchise Agreement:

Based upon the franchisor's financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by franchisees shall be deferred until the franchisor completes its pre-opening obligations under the franchise agreement.

4. **RELEASES**. The following is added to the end of Sections 2.1, 3.7, and 12.2.5 of the Franchise Agreement:

Pursuant to COMAR 02.02.08.16L, any release required as a condition of renewal and/or assignment/transfer will not apply to claims arising under the Maryland Franchise Registration and Disclosure Law.

5. **INSOLVENCY**. The following is added to the end of Section 13.1 of the Franchise Agreement:

This Section might not be enforceable under federal bankruptcy law (11 U.S.C. Sections 101 et seq.).

6. <u>VENUE</u>. Sections 19.2 and 19.9 of the Franchise Agreement is supplemented by adding the following to the end of the Section:

A franchisee may bring a lawsuit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.

1

A Maryland franchise regulation states that it is an unfair or deceptive practice to require a franchisee to waive its right to file a lawsuit in Maryland claiming a violation of the Maryland Franchise Registration and Disclosure Law. In light of the Federal Arbitration Act, there is some dispute as to whether this forum selection requirement is legally enforceable.

7. **LIMITATIONS OF CLAIMS**. The following is added to the end of Section 19.8 of the Franchise Agreement:

Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

8. <u>MARYLAND LAW</u>. The Franchise Agreement is further amended to state that "All representations requiring prospective franchisees to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.

[Signature Page to Follow]

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Franchise Agreement.

RUSTY TACO FRANCHISING, LLC	[FRANCHISEE]	
By:	By:	
Name:	Name:	
Title:	Title:	
Date:	Date:	

RIDER TO THE RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT FOR USE IN MINNESOTA

THIS RIDER (this "Rider") is made and entered	into by and between RUSTY TACO
FRANCHISING, LLC, a Delaware limited liability company	with its principal business address at 5412
W. Plano Pkwy., Suite 100, Plano, Texas 75093 ("us"), and	,
whose principal business address is	("you").

- 1. <u>BACKGROUND</u>. We and you are parties to that certain Development Agreement dated ______, 20____ (the "Franchise Agreement") that has been signed concurrently with the signing of this Rider. This Rider is annexed to and forms part of the Franchise Agreement, and the Franchise Agreement is amended only as set forth below. In the event of conflict between the terms of the Franchise Agreement and this Rider, this Rider shall govern such inconsistency. This Rider is being signed because (a) you are a resident of the State of Minnesota; (b) the Restaurant that you will develop is or will be located in the State of Minnesota; or (c) any of the offering or sales activity relating to the Franchise Agreement occurred in the State of Minnesota.
- 2. **FINANCIAL ASSURANCE**. The following is added to the end of Section 4.1 ("Initial Franchise Fee") of the Franchise Agreement:

Notwithstanding anything to the contrary, all initial fees and other payments owed by you to us or our affiliates will be deferred until your Restaurant opens and we have completed our pre-opening obligations under this Agreement.

3. **INTEREST ON LATE PAYMENTS**. The following language is added to the end of the first paragraph of Section 4.10 of the Franchise Agreement:

Notwithstanding the foregoing, you and we acknowledge that under Minnesota Statute 604.113 your penalty for an insufficient funds check will be limited to \$30 per occurrence.

4. **RELEASES.** The following is added to the end of Sections 2.1, 3.7, and 12.2.5 of the Franchise Agreement:

Any release required as a condition of renewal and/or assignment/transfer will not apply to the extent prohibited by the Minnesota Franchises Law.

5. **RENEWAL AND TERMINATION**. The following is added to the end of Sections 2 and 13 of the Franchise Agreement:

However, with respect to franchises governed by Minnesota law, we will comply with Minn. Stat. Sec. 80C.14, Subds. 3, 4 and 5 which require, except in certain specified cases, that you be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice of non-renewal of this Agreement.

6. **LOST REVENUE DAMAGES**. The following language is added to the end of Section 14.7 of the Franchise Agreement

We and you acknowledge that certain parts of this provision might not be enforceable under Minn. Rule Part 2860.4400J. However, we and you agree to enforce the provision to the

extent the law allows.

7. **INJUNCTIVE RELIEF**. Section 19.7 of the Franchise Agreement is deleted and replaced with the following:

Notwithstanding the foregoing, a court will determine if a bond is required.

8. **LIMITATIONS OF CLAIMS**. The following is added to the end of Section 19.8 of the Franchise Agreement:

; provided, however, that Minnesota law provides that no action may be commenced under Minn. Stat. Sec. 80C.17 more than 3 years after the cause of action accrues.

9. MINNESOTA LAW. Notwithstanding anything to the contrary contained in the Franchise Agreement, Minn. Stat. Sec. 80C.21 and Minn. Rule 2860.4400J prohibit us from requiring you to waive your rights to a jury trial or to waive your rights to any procedure, forum or remedies provided for by the laws of the jurisdiction, or to consent to liquidated damages, termination penalties or judgment notes.

[Signature Page to Follow]

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Franchise Agreement.

RUSTY TACO FRANCHISING, LLC	[FRANCHISEE]	
By:	By:	
Name:	Name:	
Title:	Title:	
Date:	Date:	

RIDER TO THE RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT FOR USE IN NEW YORK

THIS RIDER (this "Rider") is made and entered into by and between RUSTY TACC
FRANCHISING, LLC, a Delaware limited liability company with its principal business address at 5412
W. Plano Pkwy., Suite 100, Plano, Texas 75093 ("us"), and
whose principal business address is ("you").
1. BACKGROUND. We and you are parties to that certain Franchise Agreement dated , 20 (the " Franchise Agreement ") that has been signed concurrently with the signing of this Rider. This Rider is annexed to and forms part of the Franchise Agreement, and the Franchise Agreement is amended only as set forth below. In the event of conflict between the terms of the Franchise
Agreement and this Rider, this Rider shall govern such inconsistency. This Rider is being signed because
(a) an offer to sell the franchise is made in the State of New York; (b) an offer to buy the franchise is accepted in New York, or (c) you are domiciled in the State of New York and the Restaurant that you develop under your Franchise Agreement is or will be located in the State of New York.
2. TRANSFER BY US. The following language is added to the end of Section 12.1 of the Franchise Agreement:
However, to the extent required by applicable law, no transfer will be made except to an assignee who, in our good faith judgment, is willing and able to assume our obligations under this Franchise Agreement.
3. RELEASES . The following language is added to the end of Section 2.1, Section 3.7, and Section 12.2.5 of the Franchise Agreement:
Notwithstanding the foregoing all rights enjoyed by you and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force to the extent required by the non-waiver provisions of GBL Sections 687.4 and 687.5, as amended.
4. TERMINATION OF AGREEMENT – BY YOU . The following language is added to

the end of Section 13 of the Franchise Agreement:

You also may terminate this Development Agreement on any grounds available by law

under the provisions of Article 33 of the General Business Law of the State of New York.

5. **CHOICE OF LAW**. The following language is added to the end of Section 19.1 of the Franchise Agreement:

This Section shall not be considered a waiver of any right conferred upon you by the provisions of Article 33 of the New York State General Business Law, as amended, and the regulations issued thereunder.

6. <u>CONSENT TO JURISDICTION</u>. The following language is added to the end of Section 19.9 of the Franchise Agreement:

This Section shall not be considered a waiver of any right conferred upon you by the provisions of Article 33 of the New York State General Business Law, as amended, and the regulations issued thereunder.

[Signature Page to Follow]

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Franchise Agreement.

RUSTY TACO FRANCHISING, LLC	[FRANCHISEE]	
By:	By:	
Name:	Name:	
Title:	Title:	
Date:	Date:	

RIDER TO THE RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT FOR USE IN NORTH DAKOTA

THIS RIDER (this "Rider") is made and entered into by and between RUSTY TACO

FRANCHISING, LLC , a Delaware limited liability company with its principal business address at 54 W. Plano Pkwy., Suite 100, Plano, Texas 75093 ("us"), and	
whose principal business address is	("you").
	are parties to that certain Franchise Agreement dated
, 20 (the "Franchise Agreement	") that has been signed concurrently with the signing of
this Rider. This Rider is annexed to and forms	part of the Franchise Agreement, and the Franchise
Agreement is amended only as set forth below. In the	he event of conflict between the terms of the Franchise
Agreement and this Rider, this Rider shall govern s	such inconsistency. This Rider is being signed because
(a) an offer to sell the franchise was made or acce	pted in North Dakota; or (b) you are domiciled in the
	ill develop or operate is or will be in the State of North
Dakota.	

2. **FINANCIAL ASSURANCE**. The following is added to the end of Section 4.1 (Initial Franchise Fee) of the Franchise Agreement:

Pursuant to an order of the North Dakota Securities Commissioner, we will defer collection of the initial franchise fee and other initial payments you owe us until we have completed all of our preopening obligations to you under the Franchise Agreement, and you begin operating your Restaurant.

3. **RELEASES**. The following is added to the end of Sections 2.1 (Renewal), 3.7 (Relocation), and 12.2.5 (Transfer) of the Franchise Agreement:

Any release required as a condition of renewal and/or assignment/transfer and/or sale will not apply to the extent prohibited by the North Dakota Franchise Investment Law.

4. <u>LIQUIDATED DAMAGES</u>. The following is added to the end of Section 14.7 (Liquidated Damages) and 18.10 (Remedies Cumulative) of the Franchise Agreement:

However, any payment of liquidated damages will not be enforced to the extent prohibited by the North Dakota Franchise Investment Law

5. <u>COVENANT NOT TO COMPETE</u>. The following is added to the end of Section 15.2 (Noncompetition After Expiration or Termination of Agreement) of the Franchise Agreement:

Covenants not to compete such as those mentioned above are generally considered unenforceable in the State of North Dakota; however, we will enforce the covenants to the maximum extent the law allows.

6. <u>CHOICE OF LAW.</u> The following is added to Section 19.1 (Choice of Law) of the Franchise Agreement:

Except as otherwise required by North Dakota law, this Agreement will be governed by and construed in accordance with the laws of the State of Delaware.

7. **ARBITRATION**. The following is added to Section 19.2 (Requirement to Submit Disputes to Arbitration) and Section 19.9 (Consent to Jurisdiction) of the Franchise Agreement:

Notwithstanding the foregoing, to the extent required by the North Dakota Franchise Investment Law, and subject to your arbitration obligations, you may (i) bring an action in North Dakota for claims arising under the North Dakota Franchise Investment Law, and (ii) arbitration will be at a site to which we and you mutually agree.

- 8. <u>WAIVER OF JURY TRIAL; PUNITIVE DAMAGES</u>. Only to the extent required under North Dakota law, Section 19.4 (Waiver of Trial by Jury) and Section 19.5 (Waiver of Punitive Damages) of the Franchise Agreement are hereby deleted.
- 9. <u>LIMITATION OF CLAIMS & CLASS ACTION BAR</u>. The following is added to the end of Section 19.8 (Limitation of Claims and Class Action Bar) of the Franchise Agreement:

The statutes of limitations under North Dakota law applies with respect to claims arising under the North Dakota Franchise Investment Law.

[Signature Page to Follow]

IN WITNESS WHEREOF, the parties have executed and delivered this Rider to be effective as of the effective date of the Franchise Agreement.

RUSTY TACO FRANCHISING, LLC	[FRANCHISEE]	
By:	By:	
Name:	Name:	
Title:	Title:	
Date:	Date:	

RIDER TO THE RUSTY TACO FRANCHISING, LLC FRANCHISE AGREEMENT FOR USE IN VIRGINIA

	into by and between RUSTY TACO FRANCHISING ,
	its principal business address at 5412 W. Plano Pkwy.,
Suite 100, Plano, Texas 75093 ("us"), and	, whose principal,
business address is	("you").
, 20 (the "Franchise Agreemen this Rider. This Rider is annexed to and forms Agreement is amended only as set forth below. In	are parties to that certain Franchise Agreement dated at") that has been signed concurrently with the signing of a part of the Franchise Agreement, and the Franchise the event of conflict between the terms of the Franchise such inconsistency. This Rider is being signed because ia.
2. FINANCIAL ASSURANCE . The Franchise Agreement:	he following is added to the end of Section 4.1 of the
Franchising requires us to defer paymen	mission's Division of Securities and Retail nt of the initial franchise fee and other initial chisor until the franchisor has completed its pregreement.
IN WITNESS WHEREOF, the parties he of the effective date of the Franchise Agreement.	have executed and delivered this Rider to be effective as
RUSTY TACO FRANCHISING, LLC	[FRANCHISEE]
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

EXHIBIT I

REPRESENTATION & ACKNOWLEDGMENT STATEMENT

REPRESENTATION & ACKNOWLEDGMENT STATEMENT

DO NOT SIGN THIS QUESTIONNAIRE IF YOU ARE LOCATED IN, OR YOUR FRANCHISED BUSINESS WILL BE LOCATED IN, OR THE FRANCHISE GRANTED IS SUBJECT TO THE FRANCHISE REGISTRATION OR DISCLOSURE LAWS OF: CALIFORNIA, HAWAII, ILLINOIS, INDIANA, MARYLAND, MICHIGAN, MINNESOTA, NEW YORK, NORTH DAKOTA, RHODE ISLAND, SOUTH DAKOTA, VIRGINIA, WASHINGTON, OR WISCONSIN.

The purpose of this Statement is to demonstrate to **RUSTY TACO FRANCHISING, LLC**, a Delaware limited liability company ("Franchisor") that the person(s) signing below ("I," "me" or "my"), whether acting individually or on behalf of any legal entity established to acquire the development and/or franchise rights ("Franchisee"), (a) fully understands that the purchase of a Rusty Taco Restaurant franchise is a significant long-term commitment, complete with its associated risks, and (b) is not relying on any statements, representations, promises or assurances that are not specifically set forth in Franchisor's Franchise Disclosure Document and Exhibits (collectively, the "FDD") with which you were provided prior to signing the Agreement in deciding to purchase the franchise.

In that regard, I represent to Franchisor and acknowledge that:

EXHIBIT B	INITIAL:
DEVELOPMENT AGREEMENT I understand that buying a franchise is not a guarantee of success. Purchasing or establishing any business is risky, and the success or failure of the franchise is subject to many variables such as my skills and abilities (and those of my partners, officers, employees), the time my associates and I devote to the business, competition, interest rates, the economy, inflation, operation costs, location, lease terms, the market place generally and other economic and business factors. I am aware of and am willing to undertake these business risks. I understand that the success or failure of my business will depend primarily upon my efforts and not those of Franchisor.	
I acknowledge that I have had the opportunity to personally and carefully review the FDD and have, in fact, done so. I have been advised to have professionals (such as lawyers and accountants) review the documents for me and to have them help me understand these documents. I have also been advised to consult with other franchisees regarding the risks associated with the purchase of the franchise.	INITIAL:
Neither the Franchisor nor any of its officers, employees or agents (including any franchise broker) has made a statement, promise or assurance to me concerning any matter related to the franchise (including those regarding advertising, marketing, training, support service or assistance provided by Franchisor) that is contrary to, or different from, the information contained in the FDD.	INITIAL:
My decision to purchase the Franchise has not been influenced by any oral representations, assurances, warranties, guarantees or promises whatsoever made by the Franchisor or any of its officers, employees or agents (including any franchise broker), including as to the likelihood of success of the franchise.	INITIAL:
I have made my own independent determination as to whether I have the capital necessary to fund the business and my living expenses, particularly during the start-up phase.	INITIAL:

PLEASE READ THE FOLLOWING QUESTION CAREFULLY. THEN SELECT YES OR NO AND PLACE YOUR INITIALS WHERE INDICATED.	INITIAL:
Have you received any information from the Franchisor or any of its officers, employees or agents (including any franchise broker) concerning actual, average, projected or forecasted sales, revenues, income, profits or earnings of the franchise business (including any statement, promise or assurance concerning the likelihood of success) other than information contained in the FDD?	
☐ Yes ☐ No (Initial Here:)	
If you selected "Yes," please describe the information you received on the lines below:	

Prohibited Parties Clause. I acknowledge that Franchisor, its employees and its agents are subject to U.S. laws that prohibit or restrict (a) transactions with certain parties, and (b) the conduct of transactions involving certain foreign parties. These laws include, without limitation, U.S. Executive Order 13224, the U.S. Foreign Corrupt Practices Act, the Bank Secrecy Act, the International Money Laundering Abatement and Anti-terrorism Financing Act, the Export Administration Act, the Arms Export Control Act, the U.S. Patriot Act, and the International Economic Emergency Powers Act, and the regulations issued pursuant to these and other U.S. laws. As part of the express consideration for the purchase of the franchise, I represent that neither I nor any of my employees, agents, or representatives, nor any other person or entity associated with me, is now, or has been listed on:

- 1. the U.S. Treasury Department's List of Specially Designated Nationals;
- 2. the U.S. Commerce Department's Denied Persons List, Unverified List, Entity List, or General Orders:
- 3. the U.S. State Department's Debarred List or Nonproliferation Sanctions; or
- 4. the Annex to U.S. Executive Order 13224.

I warrant that neither I nor any of my employees, agents, or representatives, nor any other person or entity associated with me, is now, or has been: (i) a person or entity who assists, sponsors, or supports terrorists or acts of terrorism; or (ii) is owned or controlled by terrorists or sponsors of terrorism. I warrant that I am now, and have been, in compliance with U.S. anti-money laundering and counter-terrorism financing laws and regulations, and that any funds provided by me to Franchisor were legally obtained in compliance with these laws.

I further covenant that neither I nor any of my employees, agents, or representatives, nor any other person or entity associated with me, will, during the term of the Franchise Agreement, become a person or entity described above or otherwise become a target of any anti-terrorism law.

[Signature page follows]

Sign: Name:	
Capacity:	Individually, and for and on behalf of [Franchisee Name]
Sign:	
Name:	
	Individually, and for and on behalf of [Franchisee Name]
Sign:	
Name:	
Capacity:	Individually, and for and on behalf of [Franchisee Name]

NEW YORK REPRESENTATIONS PAGE

FRANCHISOR REPRESENTS THAT THIS FRANCHISE DISCLOSURE DOCUMENT DOES NOT KNOWINGLY OMIT ANY MATERIAL FACT OR CONTAIN ANY UNTRUE STATEMENT OF A MATERIAL FACT.

EXHIBIT J

RECEIPTS

STATE EFFECTIVE DATES

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the states, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered, or exempt from registration, as of the Effective Date stated below:

State	Effective Date
California	Pending
Hawaii	Not Registered
Illinois	April 24, 2025
Indiana	April 24, 2025
Maryland	Pending
Michigan	April 24, 2025
Minnesota	Pending
New York	Pending
North Dakota	Pending
Rhode Island	Not Registered
South Dakota	Not Registered
Virginia	Pending
Washington	Not Registered
Wisconsin	April 24, 2025

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

ITEM 23 RECEIPT

This Disclosure Document summarizes certain provisions of the Development Agreement and Franchise Agreement and other information in plain language. Read this Disclosure Document and all agreements carefully.

If Rusty Taco Franchising, LLC offers you a franchise, it must provide this Disclosure Document to you 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale. Under Iowa law, Rusty Taco Franchising, LLC must give you this Disclosure Document at the earlier of the 1st personal meeting or 14 calendar days before you sign an agreement with, or make a payment to, franchisor or an affiliate in connection with the proposed franchise sale. Under Michigan law, Rusty Taco Franchising, LLC must provide this Disclosure Document at least 10 business days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale. Under New York law, we must provide this disclosure document at the earlier of the 1st personal meeting or 10 business days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale.

If Rusty Taco Franchising, LLC does not deliver this Disclosure Document on time or if it contains a false or misleading statement, or a material omission, a violation of federal and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state agency identified on Exhibit G.

Issuance date: April 24, 2025

(972) 467-8095

The franchise seller who offered you a Rusty Taco franchise is:

The Franchisor is Rusty Taco Franchising, LLC, located at 5412 W. Plano Pkwy., Suite 100, Plano, Texas 75093. Its telephone number is (972) 467-8095.

□ Daniel Smith
Rusty Taco Franchising, LLC,
5412 W. Plano Pkwy., Suite 100,
Plano, Texas 75093
□ Denise Fenton
Rusty Taco Franchising, LLC,
5412 W. Plano Pkwy., Suite 100,
Plano, Texas 75093
□ Rusty Taco Franchising, LLC,
5412 W. Plano Pkwy., Suite 100,
Plano, Texas 75093
Plano, Texas 75093

Rusty Taco Franchising, LLC authorizes the respective state agencies identified on Exhibit G to receive service of process for it in the particular state.

(972) 467-8095

I received a disclosure document dated **April 24, 2025** that included the following Exhibits:

(972) 467-8095

Exhibit A – Financial Statements

Exhibit B – Development Agreement

Exhibit C – Franchise Agreement

Exhibit D – General Release

Exhibit E – List of Franchisees

PROSPECTIVE FRANCHISEE:

Exhibit F – Table of Contents of Operations Manual

Exhibit F – Table of Contents of Operations Manual

Exhibit F – Table of Contents of Operations Manual

Exhibit G – State Administrators/Agents for Service of Process

Exhibit H – State Addenda and Agreement Riders

Exhibit I – Representation & Acknowledgment Statement

Exhibit J – Receipts

If an individual:

If a business entity:

Name of Business Entity:

By:

Its:

(Print Name):

Dated:

Please sign this copy of the receipt, print the date on which you received this Disclosure Document, and return it by mail to General Counsel, Rusty Taco Franchising, LLC, located at 35412 W. Plano Pkwy., Suite 100, Plano, Texas 75093; or by email to franchise@rustytaco.com.

RECEIPT

This Disclosure Document summarizes certain provisions of the Development Agreement and Franchise Agreement and other information in plain language. Read this Disclosure Document and all agreements carefully.

If Rusty Taco Franchising, LLC offers you a franchise, it must provide this Disclosure Document to you 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale. Under Iowa law, Rusty Taco Franchising, LLC must give you this Disclosure Document at the earlier of the 1st personal meeting or 14 calendar days before you sign an agreement with, or make a payment to, franchisor or an affiliate in connection with the proposed franchise sale. Under Michigan law, Rusty Taco Franchising, LLC must provide this Disclosure Document at least 10 business days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale. Under New York law, we must provide this disclosure document at the earlier of the 1st personal meeting or 10 business days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale.

If Rusty Taco Franchising, LLC does not deliver this Disclosure Document on time or if it contains a false or misleading statement, or a material omission, a violation of federal and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state agency identified on Exhibit G.

Issuance date: April 24, 2025

The Franchisor is Rusty Taco Franchising, LLC, located at 5412 W. Plano Pkwy., Suite 100, Plano, Texas 75093. Its telephone number is (972) 467-8095.

The franchise seller who offered you a Rusty Taco franchise is:

☐ Daniel Smith	☐ Denise Fenton	U
Rusty Taco Franchising, LLC,	Rusty Taco Franchising, LLC,	Rusty Taco Franchising, LLC,
5412 W. Plano Pkwy., Suite 100,	5412 W. Plano Pkwy., Suite 100,	5412 W. Plano Pkwy., Suite 100
Plano, Texas 75093	Plano, Texas 75093	Plano, Texas 75093
(972) 467-8095	(972) 467-8095	(972) 467-8095

Rusty Taco Franchising, LLC authorizes the respective state agencies identified on Exhibit G to receive service of process for it in the particular state.

S

I received a disclosure document dated **April 24, 2025** that included the following Exhibits:

Exhibit A – Financial Statements Exhibit B – Development Agreement Exhibit C – Franchise Agreement Exhibit D – General Release Exhibit E – List of Franchisees	Exhibit F – Table of Contents of Operations Manual Exhibit G – State Administrators/Agents for Service of Proces Exhibit H – State Addenda and Agreement Riders Exhibit I – Representation & Acknowledgment Statement Exhibit J – Receipts	
PROSPECTIVE FRANCHISEE:		
If a business entity:	If an individual:	
Name of Business Entity: By:	(Print Name):	
Its:(Print Name):	Dated:	

Please sign this copy of the receipt, print the date on which you received this Disclosure Document, and keep it for your records.